# Stability Analysis and Controller Synthesis for a Class of Piecewise Smooth Systems

Behzad Samadi

A Thesis

in

The Department

of

Mechanical and Industrial Engineering

Presented in Partial Fulfillment of the Requirements for the Degree of Doctor of Philosophy at Concordia University Montréal, Québec, Canada

April 2008

© Behzad Samadi, 2008



Library and Archives Canada

Published Heritage Branch

395 Wellington Street Ottawa ON K1A 0N4 Canada Bibliothèque et Archives Canada

Direction du Patrimoine de l'édition

395, rue Wellington Ottawa ON K1A 0N4 Canada

> Your file Votre référence ISBN: 978-0-494-37764-2 Our file Notre référence ISBN: 978-0-494-37764-2

## NOTICE:

The author has granted a nonexclusive license allowing Library and Archives Canada to reproduce, publish, archive, preserve, conserve, communicate to the public by telecommunication or on the Internet, loan, distribute and sell theses worldwide, for commercial or noncommercial purposes, in microform, paper, electronic and/or any other formats.

The author retains copyright ownership and moral rights in this thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without the author's permission.

## AVIS:

L'auteur a accordé une licence non exclusive permettant à la Bibliothèque et Archives Canada de reproduire, publier, archiver, sauvegarder, conserver, transmettre au public par télécommunication ou par l'Internet, prêter, distribuer et vendre des thèses partout dans le monde, à des fins commerciales ou autres, sur support microforme, papier, électronique et/ou autres formats.

L'auteur conserve la propriété du droit d'auteur et des droits moraux qui protège cette thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

In compliance with the Canadian Privacy Act some supporting forms may have been removed from this thesis.

While these forms may be included in the document page count, their removal does not represent any loss of content from the thesis.

Canada

Conformément à la loi canadienne sur la protection de la vie privée, quelques formulaires secondaires ont été enlevés de cette thèse.

Bien que ces formulaires aient inclus dans la pagination, il n'y aura aucun contenu manquant.

#### CONCORDIA UNIVERSITY

#### School of Graduate Studies

This is to certify that the thesis proposal prepared

#### By: Behzad Samadi

## Entitled: Stability Analysis and Controller Synthesis for a Class of Piecewise Smooth Systems

and submitted in partial fulfilment of the requirements for the degree of

#### **Doctor of Philosophy**

complies with the regulations of this University and meets the accepted standards with respect to originality and quality.

Signed by the final examining committee:



Approved by \_\_\_\_\_

Chair of the ME Department

Dean of Engineering

## ABSTRACT

Stability Analysis and Controller Synthesis for a Class of Piecewise Smooth Systems

Behzad Samadi, Ph.D.

Concordia Unviersity, 2008

This thesis deals with the analysis and synthesis of piecewise smooth (PWS) systems. In general, PWS systems are nonsmooth systems, which means their vector fields are discontinuous functions of the state vector. Dynamic behavior of nonsmooth systems is richer than smooth systems. For example, there are phenomena such as sliding modes that occur only in nonsmooth systems. In this thesis, a Lyapunov stability theorem is proved to provide the theoretical framework for the stability analysis of PWS systems. Piecewise affine (PWA) and piecewise polynomial (PWP) systems are then introduced as important subclasses of PWS systems.

The objective of this thesis is to propose efficient *computational* controller synthesis methods for PWA and PWP systems. Three synthesis methods are presented in this thesis. The first method extends linear controllers for uncertain nonlinear systems to PWA controllers. The result is a PWA controller that maintains the performance of the linear controller while extending its region of convergence. However, the synthesis problem for the first method is formulated as a set of bilinear matrix inequalities (BMIs), which are not easy to solve. Two controller synthesis methods are then presented to formulate PWA and PWP controller synthesis as convex problems, which are numerically tractable. Finally, to address practical implementation issues, a time-delay approach to stability analysis of sampled-data PWA systems is presented. The proposed method calculates the maximum sampling time for a sampled-data PWA system consisting of a continuous-time plant and a discrete-time emulation of a continuous-time PWA state feedback controller.

## Dedicated to Hermin,

my mother,

whose unconditional love is absolutely incredible

## ACKNOWLEDGEMENTS

The road toward a PhD is long and full of turns, twists and unexpected obstacles. It is crucially important to have a dedicated and insightful leader. I was very fortunate during my PhD to have Dr. Luis Rodrigues as my supervisor. He gave me very interesting questions at first and then let me free to find my way to the target. Although I was free in choosing the path, I always had the luxury of his comments and hints. Writing this thesis was definitely not possible without his technical, moral and generous financial support.

I am very honored to have very distinguished researchers in my PhD defence committee. I am very grateful to Dr. Pablo Parrilo for accepting to travel from Boston to Montreal to be a member of my PhD defence committee and for his valuable comments. I would like to thank Dr. Khashayar Khorasani as a member of my PhD defence committee for his thought-provoking questions. I am also very thankful to Dr. Subhash Rakheja for being a member of my PhD defence committee. I also thank Dr. Rakheja who, as the Graduate Program Director at the time, encouraged me to apply to Concordia University when I contacted him from Iran. I thank Dr. Waiz Ahmed for being a member of my PhD defence and for his enjoyable and useful course on vehicle dynamics. I am also grateful to Dr. Brandon Gordon for giving me the opportunity to come to Concordia and work in his research group.

My sincere acknowledgments go to all professors and teachers that I had the opportunity and honor to be their student. I am specially grateful to Dr. Mohammad Ali Massoumnia who is a role model for me as an expert in the field of control theory and engineering. It was always intellectually stimulating to be in his classes at Sharif University of Technology, from the very first course on fundamentals of control theory to advanced courses on multivariable control and estimation theory. I

v

am also very grateful to Dr. Kamaleddin Nikravesh, my supervisor during my Master of Science program at Amirkabir University of Technology (Tehran Polytechnic), for teaching me how to perform high quality research.

I would like to thank Concordia University and le Fonds québécois de la recherche sur la nature et les technologies (FQRNT) for the awards, scholarships and financial support. I would also like to thank the staff of the Department of Mechanical and Industrial Engineering for their support, specially Andie Zeliger who was always welcoming and had the answers to all my questions in the first days I arrived at Concordia; Leslie Hosein for her thoughtful care and support; Arlene Zimmerman whose kind attention in forwarding my email to Dr. Subhash Rakheja led to my application and acceptance to Concordia University; Gilles Huard for his dedication to supporting students in their practical projects and for his very stimulating and fundamental discussions and finally Joseph Hulet for his great work in keeping the computers at Concordia University functional.

Special heartfelt thanks to all my friends in Montreal including my officemates, Iranian friends and my fellow toastmasters at Speak with Style Toastmasters Club. They made living in Montreal a life-changing experience for me. I would like to thank Farzaneh Abdollahi, Nader Meskin and Hani Khoshdel Nikkhoo for their technical help on my thesis and presentation. I am also grateful to all my friends who supported me through the years in Montreal including but not limited to Adrian, Ahmad Reza, Ali, Allan, Amaan, Amin, Amir, Atefeh, Azam, Behnood, Behnoush, Carole, Diana, Elham, Farhad, François, Hagire, Hamid, Heather, Helen, Homa, Hoora, Joy, Khosro, Mahsa, Maryam, Mohsen, Mojdeh, Narendra, Nargess, Nazanin, Negin, Payam, Peter, Ralf, Ramin, Rayhaneh, Roksana, Sahar, Samer, Sanaz, Sara, Scott, Shadi, Shirin, Siamak, Stefan, Sven, Vivian, Xiaoxi and Yashar. I am sure that this list is incomplete and I do apologize for the missing names.

vi

My deepest gratitude and love go to my family in Iran and Canada. My aunt Mrs. Mehran Rahnama, her husband Dr. Iraj Mansouri, my cousin Mehrnaz (Paris), her fiance Alexandre and my other cousin Mehrtash (Mack) made me feel at home. I will never forget their love and support.

Words can not describe my gratitude to my mother, Hermin, for her sacrifices for our family. Although we have lived on different sides of the planet for more than 4 years, her incredible and unconditional love has always been a great source of support and motivation for me. I am also deeply grateful to my sister, Samareh, my brother-in-law, Farhad, my brother, Amir (Mehrzad) and my sister-in-law, Parisa. I would also like to thank my father, Behrouz for his support all through my life. To them I owe all that I have ever accomplished. A worthier notion shall not arise The God of fame in whom powers reside Provider, Sustainer, the Ultimate Guide. Creator of the world & the orderly universal run The light giver to the Moon, Mercury and Sun. Transcends all name, label and notion

Praise Lord of Life, God the Wise

Is the author of form and motion

Capable is he who is wise

Happiness from wisdom will arise.

The Epic of the Kings, by Ferdowsi, Persian poet (935-1020), translated by Sh. Shahriari

به نام خداوند جان و خرد کزین برتر اندیشه برنکذرد

خداوندنام وخداوندجاي

خداوندروزي ده رسماي

خداوند کیوان و کردان سیر

فروزنده ماه وناہيدو مهر

ز نام و نثان و کان برتر ست

تخارنده ی برشده پیکر ست

توانابودهرکه دانابود زدانش دل پیربرنابود

شاینامه فردوسی

# TABLE OF CONTENTS

| List of Figures                                            |              |                                                                    |  |  |
|------------------------------------------------------------|--------------|--------------------------------------------------------------------|--|--|
| List of Tables                                             |              |                                                                    |  |  |
| List of Symbols and Abbreviations                          |              |                                                                    |  |  |
| 1                                                          | Introduction |                                                                    |  |  |
|                                                            | 1.1          | Motivation                                                         |  |  |
|                                                            | 1.2          | Objective                                                          |  |  |
| 1.3 Literature Review                                      |              | Literature Review                                                  |  |  |
|                                                            |              | 1.3.1 Previous work on hybrid dynamical systems                    |  |  |
|                                                            |              | 1.3.2 Previous work on PWP systems                                 |  |  |
|                                                            |              | 1.3.3 Previous work on PWA systems                                 |  |  |
| 1.4 Contributions                                          |              | Contributions                                                      |  |  |
|                                                            |              | 1.4.1 Publications                                                 |  |  |
|                                                            | 1.5          | Structure of the Thesis                                            |  |  |
| 2                                                          | Lyaj         | punov Stability for Piecewise Smooth Systems 16                    |  |  |
|                                                            | 2.1          | Introduction                                                       |  |  |
|                                                            | 2.2          | PWS Systems                                                        |  |  |
|                                                            | 2.3          | Lyapunov stability                                                 |  |  |
| <ul> <li>2.4 Monotonicity of nonsmooth functions</li></ul> |              | Monotonicity of nonsmooth functions                                |  |  |
|                                                            |              | Dissipativity                                                      |  |  |
|                                                            |              | 2.5.1 Piecewise smooth systems with discontinuous vector fields 26 |  |  |
|                                                            |              | 2.5.2 Piecewise smooth systems with continuous vector fields 27    |  |  |
|                                                            | 2.6          | PWA Systems                                                        |  |  |
|                                                            |              | 2.6.1 Quadratic Lyapunov Function                                  |  |  |
|                                                            |              | 2.6.2 Piecewise Quadratic Lyapunov Function                        |  |  |

|   |                                                                                           | 2.6.3 Sum of Squares Lyapunov Function                                                                                                                                                                                                                                                                                               | 41                                                                                                                                             |
|---|-------------------------------------------------------------------------------------------|--------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------------------------------------------------------------------------------------------------------------------------------------------|
|   | 2.7                                                                                       | PWA Slab Systems                                                                                                                                                                                                                                                                                                                     | 44                                                                                                                                             |
|   | 2.8                                                                                       | PWP Systems                                                                                                                                                                                                                                                                                                                          | 46                                                                                                                                             |
|   | 2.9                                                                                       | Conclusions                                                                                                                                                                                                                                                                                                                          | 47                                                                                                                                             |
| 3 | $\mathbf{Ext}$                                                                            | ension of local linear controllers to global piecewise affine con-                                                                                                                                                                                                                                                                   |                                                                                                                                                |
|   | trol                                                                                      | llers for uncertain nonlinear systems                                                                                                                                                                                                                                                                                                | 48                                                                                                                                             |
|   | 3.1                                                                                       | Introduction                                                                                                                                                                                                                                                                                                                         | 49                                                                                                                                             |
|   | 3.2                                                                                       | Illustrative Example                                                                                                                                                                                                                                                                                                                 | 51                                                                                                                                             |
|   | 3.3                                                                                       | Extension of a Linear Controller to a PWA Controller                                                                                                                                                                                                                                                                                 | 53                                                                                                                                             |
|   |                                                                                           | 3.3.1 Step 1: Robust linear controller design                                                                                                                                                                                                                                                                                        | 55                                                                                                                                             |
|   |                                                                                           | 3.3.2 Step 2: PWA state feedback design                                                                                                                                                                                                                                                                                              | 57                                                                                                                                             |
|   | 3.4                                                                                       | Numerical Examples                                                                                                                                                                                                                                                                                                                   | 65                                                                                                                                             |
|   | 3.5                                                                                       | Conclusions                                                                                                                                                                                                                                                                                                                          | 68                                                                                                                                             |
| 4 | Cor                                                                                       | ntroller synthesis for piecewise affine slab differential inclusions:                                                                                                                                                                                                                                                                |                                                                                                                                                |
|   |                                                                                           |                                                                                                                                                                                                                                                                                                                                      |                                                                                                                                                |
|   | a du                                                                                      | uality-based convex optimization approach                                                                                                                                                                                                                                                                                            | 73                                                                                                                                             |
|   | <b>a d</b><br>4.1                                                                         | uality-based convex optimization approachIntroduction                                                                                                                                                                                                                                                                                | <b>73</b><br>74                                                                                                                                |
|   | a du<br>4.1<br>4.2                                                                        | uality-based convex optimization approach         Introduction         Mathematical Preliminaries                                                                                                                                                                                                                                    | <b>73</b><br>74<br>76                                                                                                                          |
|   | <ul> <li>a du</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> </ul>                           | uality-based convex optimization approach         Introduction         Mathematical Preliminaries         Polytopic PWA Slab Differential Inclusions                                                                                                                                                                                 | <b>73</b><br>74<br>76<br>76                                                                                                                    |
|   | <ul><li>a du</li><li>4.1</li><li>4.2</li><li>4.3</li></ul>                                | uality-based convex optimization approach         Introduction                                                                                                                                                                                                                                                                       | <b>73</b><br>74<br>76<br>76<br>78                                                                                                              |
|   | <ul> <li>a du</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> </ul>              | uality-based convex optimization approach         Introduction         Mathematical Preliminaries         Polytopic PWA Slab Differential Inclusions         4.3.1         PWA envelope for nonlinear systems         Analysis                                                                                                       | <ul> <li><b>73</b></li> <li>74</li> <li>76</li> <li>76</li> <li>78</li> <li>79</li> </ul>                                                      |
|   | <ul> <li>a du</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> </ul>              | uality-based convex optimization approach         Introduction                                                                                                                                                                                                                                                                       | <ul> <li>73</li> <li>74</li> <li>76</li> <li>76</li> <li>78</li> <li>79</li> <li>80</li> </ul>                                                 |
|   | <ul> <li>a du</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> </ul>              | uality-based convex optimization approach         Introduction                                                                                                                                                                                                                                                                       | <ul> <li><b>73</b></li> <li>74</li> <li>76</li> <li>76</li> <li>78</li> <li>79</li> <li>80</li> <li>83</li> </ul>                              |
|   | <ul> <li>a du</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> <li>4.5</li> </ul> | uality-based convex optimization approach         Introduction                                                                                                                                                                                                                                                                       | <ul> <li>73</li> <li>74</li> <li>76</li> <li>76</li> <li>78</li> <li>79</li> <li>80</li> <li>83</li> <li>89</li> </ul>                         |
|   | <ul> <li>a du</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> <li>4.5</li> </ul> | uality-based convex optimization approach         Introduction         Mathematical Preliminaries         Polytopic PWA Slab Differential Inclusions         4.3.1         PWA envelope for nonlinear systems         Analysis         4.4.1         Stability analysis         Controller Synthesis         4.5.1         Stability | <ul> <li>73</li> <li>74</li> <li>76</li> <li>76</li> <li>78</li> <li>79</li> <li>80</li> <li>83</li> <li>89</li> <li>90</li> </ul>             |
|   | <ul> <li>a du</li> <li>4.1</li> <li>4.2</li> <li>4.3</li> <li>4.4</li> <li>4.5</li> </ul> | uality-based convex optimization approach         Introduction                                                                                                                                                                                                                                                                       | <ul> <li>73</li> <li>74</li> <li>76</li> <li>76</li> <li>78</li> <li>79</li> <li>80</li> <li>83</li> <li>89</li> <li>90</li> <li>93</li> </ul> |

|            | 4.7                                                               | Conclusions                                        |  |
|------------|-------------------------------------------------------------------|----------------------------------------------------|--|
| 5          | 5 Backstepping Controller Synthesis for Piecewise Polynomial Sys- |                                                    |  |
|            | tem                                                               | s: A Sum of Squares Approach 101                   |  |
|            | 5.1                                                               | Introduction                                       |  |
|            | 5.2                                                               | Integrator Backstepping                            |  |
|            |                                                                   | 5.2.1 PWP systems with discontinuous vector fields |  |
|            |                                                                   | 5.2.2 PWP systems with continuous vector fields    |  |
|            | 5.3                                                               | Recursive Backstepping Controller Design           |  |
|            |                                                                   | 5.3.1 PWP systems with discontinuous vector fields |  |
|            |                                                                   | 5.3.2 PWP systems with continuous vector fields    |  |
|            | 5.4                                                               | Numerical Examples                                 |  |
|            | 5.5                                                               | Conclusions                                        |  |
| 6          | 6 Sampled-Data Piecewise Affine Systems: A Time-Delay Approach13  |                                                    |  |
|            | 6.1                                                               | Introduction                                       |  |
|            | 6.2                                                               | Stability of Sampled-Data PWA Systems              |  |
|            | 6.3                                                               | Numerical Example                                  |  |
|            | 6.4                                                               | Conclusions                                        |  |
| 7          | Cor                                                               | aclusions 152                                      |  |
| References |                                                                   |                                                    |  |

# List of Figures

| 1.1  | Box and beam model                                                   | <b>2</b> |
|------|----------------------------------------------------------------------|----------|
| 1.2  | The class of PWS systems and its subclasses                          | 4        |
| 1.3  | Structure of the thesis                                              | 15       |
| 2.1  | Geometric illustration of sets in the proof of Theorem 2.1           | 23       |
| 2.2  | Trajectories of a stable PWA system with a sliding mode              | 35       |
| 2.3  | Trajectories of an unstable PWA system with a sliding mode           | 35       |
| 2.4  | The SOS Lyapunov function                                            | 43       |
| 2.5  | Trajectories of a stable PWA system (black) and contours of the Lya- |          |
|      | punov function (gray)                                                | 44       |
| 3.1  | Structure of the proposed PWA controller design method               | 50       |
| 3.2  | The phase plane of the open loop system                              | 53       |
| 3.3  | PWA differential inclusion                                           | 54       |
| 3.4  | The phase plane of the system with the linear controller             | 55       |
| 3.5  | The computed Lyapunov function - Example 3.1                         | 66       |
| 3.6  | The designed PWA controller - Example 3.1                            | 67       |
| 3.7  | The phase plane of the closed-loop system with the PWA controller    |          |
|      | (solid) and the linear controller (dashed) - Example 3.1 $\ldots$    | 68       |
| 3.8  | PWA function - Example 3.2                                           | 69       |
| 3.9  | Trajectories of the closed-loop system for the PWA controller pro-   |          |
|      | posed in [96] - Example 3.2                                          | 69       |
| 3.10 | Trajectories of the closed-loop system for the proposed PWA con-     |          |
|      | troller - Example 3.2                                                | 70       |
| 3.11 | Trajectories of the open loop system - Example 3.3                   | 70       |

| 3.12 Trajectories of the closed-loop system for the linear controller - Ex-                                                                                                                   |            |
|-----------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------------|------------|
| ample 3.3                                                                                                                                                                                     | '1         |
| 3.13 Trajectories of the closed-loop system for the PWA controller - Ex-                                                                                                                      |            |
| ample 3.3                                                                                                                                                                                     | '1         |
| 3.14 Piecewise quadratic Lyapunov function - Example 3.3                                                                                                                                      | '2         |
| 4.1 The computed bounding envelope for a scalar function 9                                                                                                                                    | )5         |
| 4.2 Trajectories of the closed loop nonlinear system                                                                                                                                          | )7         |
| 4.3 The computed bounding envelope for a scalar function 9                                                                                                                                    | )8         |
| 4.4 Trajectories of the closed loop nonlinear system 9                                                                                                                                        | )9         |
| 4.5 Trajectories of the closed loop nonlinear system                                                                                                                                          | )0         |
| 5.1 The trajectory of the closed-loop PWA system for $x(0) = [0.1 \ 0.5]^{\text{T}}$ . 11                                                                                                     | 8          |
| 5.2 Single-link flexible-joint robot                                                                                                                                                          | 9          |
| 5.3 State variables of the nonlinear model - open loop $\ldots \ldots \ldots$ | 20         |
| 5.4 PWP approximation of $f_1(x_1)$                                                                                                                                                           | 21         |
| 5.5 PWP approximation of $f_2(x_4)$                                                                                                                                                           | 21         |
| 5.6 The structure of the regions of the PWP system (5.85) $\ldots \ldots \ldots 12$                                                                                                           | 25         |
| 5.7 State variables of the nonlinear model - PWA controller                                                                                                                                   | 27         |
| 5.8 Control input - PWA controller                                                                                                                                                            | 27         |
| 5.9 State variables of the nonlinear model - PWP controller                                                                                                                                   | 30         |
| 5.10 Control input - PWA controller                                                                                                                                                           | 30         |
| 6.1 Pitch model for the experimental helicopter [38]                                                                                                                                          | <b>l</b> 6 |
| 6.2 PWA function - Helicopter model                                                                                                                                                           | 19         |
| 6.3 Trajectories of the nonlinear Helicopter model - continuous time PWA                                                                                                                      |            |
| controller $\ldots \ldots 15$                                                        | 60         |
| 6.4 Trajectories of the nonlinear Helicopter model - sampled data PWA                                                                                                                         |            |
| controller                                                                                                                                                                                    | 50         |

## List of Tables

# List of Symbols and Abbreviations

| LMI                                             | Linear Matrix Inequality                                                          |
|-------------------------------------------------|-----------------------------------------------------------------------------------|
| BMI                                             | Bilinear Matrix Inequality                                                        |
| PWA                                             | Piecewise Affine                                                                  |
| PWL                                             | Piecewise Linear                                                                  |
| PWP                                             | Piecewise Polynomial                                                              |
| PWQ                                             | Piecewise Quadratic                                                               |
| PWS                                             | Piecewise Smooth                                                                  |
| SOS                                             | Sum of Squares                                                                    |
| X                                               | State space                                                                       |
| ${\mathcal W}$                                  | Disturbance space                                                                 |
| $\mathcal{R}_i$                                 | Polytopic region in $\mathcal{X}$                                                 |
| $\mathcal{R}^{\mathcal{X} 	imes \mathcal{W}}_i$ | Polytopic region in $\mathcal{X} \times \mathcal{W}$                              |
| $\overline{\mathcal{R}}_i$                      | Closure of $\mathcal{R}_i$                                                        |
| $\mathcal{I}(x)$                                | $=\{i x\in\overline{\mathcal{R}}_i\}$                                             |
| $\mathcal{I}(x,w)$                              | $= \{i  \ (x,w) \in \overline{\mathcal{R}}_i^{\mathcal{X} \times \mathcal{W}} \}$ |
|                                                 |                                                                                   |

# Chapter 1

# Introduction

## 1.1 Motivation

Piecewise smooth (PWS) systems are multi-model systems that offer a good modeling framework for complex dynamical systems. For example, many engineering systems of practical interest have nonlinear components that can naturally be modeled by PWS characteristics. Some examples are:

- Saturation [56], [32]
- Dead-zone [32]
- Backlash [73]
- Electrical circuits with diodes [103], [22], [47]
- Mechanical oscillators with clearance [77]
- Moving parts with Coulomb friction [3]

To illustrate PWS systems, consider a nonlinear mechanical system consisting of a box and a beam (Fig. 1.1) described by the following model

$$\dot{x}_{1} = x_{2}$$

$$\dot{x}_{2} = \begin{cases} -\mu g \cos(x_{3}) - g \sin(x_{3}) & \text{if } x_{2} > 0 \\ \mu g \cos(x_{3}) - g \sin(x_{3}) & \text{if } x_{2} < 0 \end{cases}$$

$$\dot{x}_{3} = \frac{1}{\tau} (-x_{3} + u)$$
(1.1)

where  $x_1 = x_b$ ,  $x_2 = \dot{x}_b$ ,  $x_3 = \theta$ ,  $g = 9.8 \text{m/s}^2$  and  $\mu = 0.1$  is the static friction coefficient. u is the reference signal for  $\theta$  and  $\tau$  is the time-constant of the actuator. The set of state space equations (1.1) is a special case of PWS systems in the following form

$$\dot{x}(t) = f_i(x(t)) + g_i(x(t))u(t), \ x(t) \in \mathcal{R}_i.$$
 (1.2)

where a subset of the state space  $\mathcal{X} \subset \mathbb{R}^n$  is partitioned into regions  $\mathcal{R}_i$  for  $i=1,\ldots,M$ such that  $\bigcup_{i=1}^M \overline{\mathcal{R}}_i = \mathcal{X}, \ \mathcal{R}_i \cap \mathcal{R}_j = \emptyset, \ i \neq j$ , where  $\overline{\mathcal{R}}_i$  denotes the closure of  $\mathcal{R}_i$ . In the case of the box and beam system, the regions are defined as

$$\mathcal{R}_{1} = \{ x \in \mathbb{R}^{3} | x_{2} > 0 \},$$
  
$$\mathcal{R}_{2} = \{ x \in \mathbb{R}^{3} | x_{2} < 0 \}$$
(1.3)

In general, PWS systems are nonsmooth systems, which means their vector fields are discontinuous functions of the state vector. The PWS system (1.1) is itself an example of a nonsmooth system. However, most of the classical methods for analysis and synthesis of dynamical systems concentrate on smooth systems. Dynamic behavior of nonsmooth systems is richer than smooth systems. For example, there



Figure 1.1: Box and beam model

are phenomena such as sliding modes that occur only in nonsmooth systems. In fact, even the definition of trajectories of a system must be generalized to describe sliding modes of nonsmooth systems. Existing literature on nonsmooth systems mainly concentrates on the analysis of the dynamics of nonsmooth systems, while controller synthesis has not received much attention.

## 1.2 Objective

The objective of this thesis is to propose efficient *computational* controller synthesis methods for the following subclasses of PWS systems:

- Piecewise polynomial (PWP) systems
- Piecewise affine (PWA) systems
- Piecewise linear (PWL) systems

The main focus will be on PWA and PWP systems. Figure 1.2 shows the relative hierarchy of these classes of PWS systems.

It is desired to address the controller synthesis problems with convex optimization techniques such as linear matrix inequalities (LMIs) and sum of squares (SOS) programs. The reason is that there exist numerically efficient tools to solve these convex optimization problems.

In the following section, the relevant literature is reviewed to show what has been done in the field and how the proposed methods relate to previous research.

## **1.3** Literature Review

A hybrid system is defined in [65] as a dynamical system with interacting continuoustime-driven and discrete-event-driven components. The continuous part of a hybrid

3



Figure 1.2: The class of PWS systems and its subclasses

system is usually described by a differential or difference equation and the discrete part is described by a finite-state machine or a set of logic-based rules. Therefore, a hybrid system has two distinct types of state variables: real-valued and discretevalued state variables. Modern computer-based control systems that act on physical systems can be modeled by hybrid dynamical systems. As a result, the analysis and design of such systems has recently received great attention. In the following subsection, previous research on hybrid systems is briefly reviewed. The literature on PWP and PWA systems will be reviewed in separate subsections to describe in more detail the background of the proposed research.

#### **1.3.1** Previous work on hybrid dynamical systems

Hybrid systems have attracted significant attention in recent years. Special Issues on Hybrid Control Systems of the *IEEE Transactions on Automatic Control* (April 1998), *Automatica* (March 1999), *Systems and Control Letters* (October 1999), the *Proceedings of the IEEE* (July 2000), *International Journal of Robust and Nonlinear Control* (April 2001) and a new journal, *Nonlinear Analysis: Hybrid Systems* (March 2007) illustrate the fast pace of advances in this field. The development of a unified and systematic hybrid systems theory is still a growing and vibrant research area. There have been some important research efforts toward an overall unified model [13, 20, 58], a unified analysis methodology for a class of hybrid systems [20, 35], as well as a unifying view for the subclass of PWA systems [66]. Reference [105] is one of very few contributions toward a unified controller synthesis method that would provide a systematic control design tool for a large class of hybrid systems and enable designers to use the same methodology for a broad set of models and applications. A great deal of attention and efforts in hybrid systems have been focused on the modeling [11,46,111,126] and stability [15,19,35,61,79,91,93,128]. However, there are also many results on control design methods for hybrid systems. Most of the proposed controller synthesis methods fall into the following approaches:

- Supervisory Control: In this approach, continuous controllers are combined with discrete logic. A *supervisor* is used to effectively switch between several continuous control laws [60, 74, 127].
- Hierarchical Control: The controller is decomposed into hierarchical levels and it can guarantee a certain performance [21, 48, 86].
- Optimal Control: The optimal control problem is to find an input that drives the system to a desired state while minimizing a cost function that depends on the trajectory followed and the control input itself. Optimal control has recently been extended to discrete-event systems [112] and hybrid systems [20, 23, 28, 50, 113, 114].
- Distributed Control: The control task is divided among a collection of agents to increase reliability. These agents may communicate with each other to transfer information related to their sensing and decision making [2,84].
- Game theoretic approach: Control problems in this category usually require that all trajectories of the system satisfy certain properties. Properties include safety properties (for example, requiring that the state of the system

remains in a certain safety set) and liveness properties (requiring that the state eventually enters a certain target set or visits a set infinitely often) [10, 125].

For a more detailed review on hybrid systems, the reader is referred to [5] and [33]. In the following, we review the special case of PWP and PWA systems and focus on the methods that have used the structure of these systems.

#### **1.3.2** Previous work on PWP systems

PWP or spline approximation of curves and surfaces has been widely used in many different scientific contexts and engineering applications [1, 34]. However, the lack of proper methods to check the sign of polynomials has prevented PWP systems to be commonly used in the field of control systems. Recently, Ebenbauer proposed analysis and design methods for polynomial systems using sum of squares techniques in [37]. For PWP systems, one of the first attempts to design controllers was made in [87]. Paul proposed in [87] to partition the state space of an affine in the input nonlinear system into cells and to approximate the dynamics of the system in each cell by a model that is polynomial in the state. A controller is then designed for each cell using feedback linearization. A global controller is then formed by joining the individual cell controllers. The proposed method was employed in [87] to design controllers for a few examples of nonlinear systems. However, there is no guarantee for the stability of the closed loop system because a switched system consisting of stable subsystems can be unstable in general.

Recently, the class of discrete-time PWP systems was defined in [45] and a new method based on Cylindrical Algebraic Decomposition (CAD) was proposed to address the constrained finite-time optimal control problem for this class of systems. This seems to be the first systematic approach to controller synthesis for discretetime PWP systems. However, according to the authors of [45], the method suffers from excessive computational burden. For continuous-time PWP systems, a stability analysis was proposed in [93] and [85] using PWP Lyapunov functions. The advantage of the proposed method is that the analysis problem is formulated as an SOS programming which is a convex optimization problem. There exist numerical tools such as SOSTOOLS [95] and Yalmip [76] to solve SOS programming problems efficiently. However, systems with infinitely fast switching or sliding modes are excluded from the discussion in [93] and [85]. This will be one of the main topics of the thesis.

#### **1.3.3** Previous work on PWA systems

The roots of PWA systems date back to the pioneering work of Andronov (1901-1952). Andronov's first major piece of research into nonlinear dynamics concerned what is known in Russian as the *metod pripasovyvaniya*, the technique in which separate solutions for the various linear regimes of a PWL problem are joined to form a complete solution - they are "stitched together" as the graphic alternative Russian term *metod sshivaniya* puts it. More details of Andronov's research can be found in [14].

The theory of PWA systems was also used in the analysis and synthesis of nonlinear electrical circuits with most pioneering works done up until the 1970's [4,29,30,120]. In the early 1980's, Sontag [117] developed a Piecewise Linear Algebra mainly for discrete-time PWA systems. For continuous-time dynamics, a technique based on vector field considerations was developed by Pettit [92] to provide a qualitative analysis of PWL systems. In the following, the analysis and synthesis methods for discrete-time and continuous-time PWA systems are briefly reviewed.

7

#### Discrete-time PWA systems

Algorithms for computing feedback controllers for constrained discrete-time PWA systems were presented for quadratic and linear objectives in [16] and [9], respectively. Instead of computing the feedback controllers that minimize a finite time cost objective, it is also possible to obtain the infinite time optimal solution for discrete-time PWA systems [8]. These problems are formulated as Mixed Integer Quadratic or Linear Programming. Even though these approaches rely on off-line computation of a feedback law, the computation can quickly become prohibitive for larger problems. This is not only due to the high complexity of the multi-parametric programs involved, but mainly because of the exponential number of transitions between regions which can occur when a controller is computed in a dynamic programming fashion [16, 69]. As a result, some methods were proposed to obtain controllers of low complexity for linear and PWA systems as presented in [52–54]. In general, synthesis methods for discrete-time PWA systems can be classified into the following groups:

- Infinite Time Optimal Control [8,51]
- Finite Time Optimal Control [9, 12, 16, 78]
- Minimum Time Control [51, 53, 54]
- Bilinear matrix inequality (BMI) based methods for stabilization [116]
- PWA control with performance [40, 42, 43]

The controller designed using any of these methods is typically much more complex than the PWA system to be controlled. As an example, for a PWA system with 4 regions, the number of regions for the controller can range from 138 to 3904 [52]. In addition, one of the main drawbacks of the methods in [16] is the lack of an *a-priori* stability guarantee for the closed-loop system. This problem has recently been addressed in [52].

#### Continuous-time PWA systems

Sufficient conditions for analysis of continuous-time PWA systems by searching for a Lyapunov function to prove stability, can be formulated as convex optimization programs involving LMIs [17]. These mathematical programs can then be solved efficiently using polynomial-time algorithms [83]. The analysis methods are only approximate in the sense that there are no guarantees that a Lyapunov function can be found. However, if one is found, the result is unambiguous. This has been the trend of research in the linear parameter varying approach to gain scheduling (see [108] and references therein) and in the more recent work on the analysis of PWA systems based on Lyapunov functions and LMIs [18,35,49,56,67,90,91,103].

The work on switched linear systems initiated in [90] is one of the first attempts to apply Lyapunov-based methods to PWA systems. Following this work, and its extensions to nonlinear dynamics [18], a unified approach to the analysis of PWA systems and a class of hybrid systems was formulated in [35]. Several promising Lyapunov-based methods have recently been developed to analyze PWL and PWA systems [49, 56, 67, 91, 98]. Some synthesis methods [55, 56, 67, 101] have also been developed. A specific technique for state feedback control of PWA systems on simplices and rectangles was proposed in [55]. Synthesis methods using convex optimization programs based on the analysis methods in [56, 67, 91] were developed in [56, 96]. The resulting controllers designed by these methods are either patched LQRs [96] or cannot guarantee that sliding modes are avoided [56, 96] and, therefore, are not provably stabilizing. In [97, 103], a synthesis method based on BMIs has been proposed for state and output feedback stabilization of PWA systems. The method has the advantage of guaranteeing that sliding modes are not generated at the switching and the controllers are therefore provably stabilizing. Another important feature of this method for practical implementation of the controllers is that continuity of the control input can also be guaranteed at the switching. However, BMI problems are not generally convex problems and thus, are not easy to be solved efficiently.

A very important subclass of PWA systems is the class of PWA *slab sys*tems [101], for which the partition of the state space is a function of a scalar variable. Hassibi and Boyd [56] proposed methods for quadratic stabilizability and  $\mathcal{L}_2$  gain synthesis for PWA systems using PWL controllers. Three different algorithms for PWA controller synthesis for slab PWA systems have also been proposed in [101]. It has been shown that by considering an affine term in the controller, the synthesis problem can be formulated as a set of Linear Matrix Inequalities (LMIs) parametrized by a vector. Furthermore, it has also been shown that by relaxing the problem to a finite set of LMIs, it can be solved efficiently to a point near the global optimum. In addition, the global solution can be exactly found under some conditions.

In general, design methods for continuous-time PWA systems can be classified into the following groups:

- Methods based on Hamilton Jacobi Bellman Inequality [66,96]
- PWL and PWA stabilization of slab PWA systems [56, 66, 101]
- BMI-based PWA controller design methods for stabilization of generic PWA systems [97, 102, 103]
- PWL control with L<sub>2</sub> gain performance [26, 56, 66]

Considering the existing approaches for PWA and PWP controller synthesis, the contributions of this thesis are stated in the next section.

10

## **1.4 Contributions**

This thesis addresses the following questions

- How can we design a PWA controller that keeps the performance of a linear controller in a neighborhood of the equilibrium point and guarantees a larger region of attraction?
- Is it possible to formulate the PWA/PWP controller synthesis as a convex optimization problem?
- For a sampled-data implementation of a continuous-time PWA controller, how large can the sampling time be?

Therefore, the main contributions of this thesis are

1. To present a unified approach for stability analysis of PWA systems with continuous and discontinuous vector fields. The Filippov definition is considered for the solution of PWA systems and then a Lyapunov stability theorem is proved. The importance of this theorem is to show that sufficient conditions for the stability of a PWA system can be formed using a *differentiable Lyapunov function* without any need for a priori information about *attractive sliding modes* on switching surfaces. This is a great advantage over existing stability results for PWA systems in the literature because obtaining this a priori information is difficult in general. Sufficient conditions for quadratic and sum of squares (SOS) polynomial Lyapunov stability are then formulated as convex problems. The SOS conditions are less conservative than the quadratic conditions. It is shown in an example that the proposed SOS program can prove stability where quadratic and differentiable piecewise quadratic (PWQ) functions fail.

- 2. To propose a two-step controller synthesis method for a class of uncertain nonlinear systems described by PWA differential inclusions. In the first step, a robust linear controller is designed for the linear differential inclusion that describes the dynamics of the nonlinear system close to the equilibrium point. In the second step, a stabilizing PWA controller is designed that coincides with the linear controller in a region around the equilibrium point. The proposed method has two objectives: global stability and local performance. It thus enables to use well known techniques in linear control design for local stability and performance while delivering a global PWA controller that is guaranteed to stabilize the nonlinear system.
- 3. To introduce for the first time a duality-based interpretation of PWA systems. This enables controller synthesis for PWA slab systems to be formulated as a convex optimization problem. PWA L2-gain analysis and synthesis is also extended to PWA systems whose output is a PWA function of the state (as opposed to a PWL function). In addition, a convex optimization program is proposed to compute a PWA differential inclusion for nonlinear systems for which the nonlinearity is a function of one variable.
- 4. To propose a nonsmooth backstepping controller synthesis for PWP systems. The main contribution of the proposed method is to formulate controller design for a large class of PWP and PWA systems as a convex problem. The controller synthesis problem is divided in two cases. The first case consists of the construction of a sum of squares (SOS) Lyapunov function for PWP systems with discontinuous vector fields. The second case addresses the construction of a PWP Lyapunov function for PWP systems with continuous vector fields. After constructing a Lyapunov function, controller synthesis for a PWP system can be formulated as an SOS program, which is a convex optimization problem and can be efficiently solved.

5. To propose a time-delay approach to stability analysis of sampled data PWA systems consisting of a continuous-time plant and a discrete-time emulation of a continuous-time PWA state feedback controller. The sampled-data system is considered as a delayed system with a variable delay. Conditions under which the trajectories of the sampled data closed-loop system will converge to an attractive invariant set are then presented. It is also shown that when the sampling period converges to zero, the conditions of the proposed theorem coincide with sufficient conditions for the non-fragility of the stabilizing continuous-time PWA state feedback controller.

The results of the current research were submitted to and published in a few conferences and journals. The details of the publications is listed in the following subsection.

#### **1.4.1** Publications

The following publications contain the main contributions of the thesis

- 1. B. Samadi and L. Rodrigues, "Controller synthesis for piecewise affine differential inclusions: a duality-based convex optimization approach," under second revision for publication in *Automatica*.
- 2. B. Samadi and L. Rodrigues, "Extension of local linear controllers to global piecewise affine controllers for uncertain nonlinear systems," accepted for publication in the *International Journal of Systems Science*.
- B. Samadi and L. Rodrigues, "Sampled-Data Piecewise Affine Slab Systems: A Time-Delay Approach," in *Proc. of the American Control Conference*, Seattle, WA, Jun. 2008.
- 4. B. Samadi and L. Rodrigues, "Controller synthesis for piecewise affine slab

differential inclusions: a duality-based convex optimization approach," in *Proc.* of the 46th Conference on Decision and Control, New Orleans, LA, Dec. 2007.

- B. Samadi and L. Rodrigues, "Backstepping Controller Synthesis for Piecewise Affine Systems: A Sum of Squares Approach," in *Proc. of the IEEE International Conference on Systems, Man, and Cybernetics (SMC 2007)*, Montreal, Oct. 2007.
- B. Samadi and L. Rodrigues, "Extension of a local linear controller to a stabilizing semi-global piecewise-affine controller," 7th Portuguese Conference on Automatic Control, Lisbon, Portugal, Sep. 2006.

## **1.5** Structure of the Thesis

The thesis is structured as shown in Figure 1.3. Chapter 2 defines PWS, PWP and PWA systems and presents a unified approach for stability analysis of PWS systems with continuous and discontinuous vector fields. In Chapter 3, continuous PWA differential inclusions are defined and sufficient conditions for monotonicity of PWQ Lyapunov functions for these inclusions are proved. A two-step controller synthesis method is then presented for a class of uncertain nonlinear systems described by PWA differential inclusions. Chapter 4 introduces the parameter set and the dual parameter set for PWA slab systems. It then provides stability and performance analysis and synthesis tools for PWA slab systems. In Chapter 5, a backstepping approach to controller synthesis for PWP systems in strict feedback form is presented. Chapter 6 addresses stability analysis of sampled-data PWA systems consisting of a continuous-time plant in feedback connection with a discrete-time emulation of a continuous-time PWA state feedback controller.



Figure 1.3: Structure of the thesis

# Chapter 2

# Lyapunov Stability for Piecewise Smooth Systems

The main objective of this chapter is to present a unified approach to stability analysis of PWS systems with continuous and discontinuous vector fields. The Filippov definition is considered for the solution of these systems and then a Lyapunov stability theorem is proved. The importance of this theorem is to show that sufficient conditions for the stability of a PWS system can be formed without any need for a-priori information about *attractive sliding modes* on switching surfaces. This is a significant advantage over existing stability results for switched systems in the literature because obtaining this a-priori information is difficult in general.

## 2.1 Introduction

There have been different approaches to construct a Lyapunov function to provide sufficient conditions for the stability of PWS systems. For a survey of existing approaches for stability analysis of hybrid systems and switched linear systems the readers are referred to [35] and [75]. A general framework for analyzing stability of nonlinear switched systems using multiple Lyapunov functions is presented in [25]. It is shown in [56] that by searching for quadratic Lyapunov functions, sufficient conditions for stability of PWA systems can be formulated as convex optimization problems with LMI constraints. Finding a common quadratic Lyapunov function for all linear modes is used to analyze the stability of switched linear systems under arbitrary switching in [74].

However, there are stable PWA systems for which a quadratic Lyapunov function does not exist. Examples of such systems are shown in [66, p. 47]. Conservativeness of a quadratic form is the motivation for studying nonquadratic Lyapunov functions. As an example, continuous PWQ Lyapunov functions were extensively investigated in recent years (see [19, 66, 91, 102]). However, it is a common misunderstanding in the literature to believe that if there is a continuous PWQ or PWP function that is positive definite and decreasing with time along each vector field of a switched affine system then the system is stable. A counter-example will be provided in section 2.6.2.

A recent result in [88] shows that the existence of a common quadratic Lyapunov function for the linear parts of a PWA system in every mode is sufficient for exponential convergence of the system if the vector field of the PWA system is continuous. Exponential convergence is defined in the same reference. The case of discontinuous vector fields is studied in [89] and it is shown that the existence of a common quadratic Lyapunov function for linear parts of the system is not a sufficient condition for convergence. Necessary and sufficient conditions for quadratic convergence of the special case of bimodal PWA systems are then derived.

SOS polynomials were also proposed as candidate Lyapunov functions. In fact, quadratic Lyapunov functions are a special class of SOS Lyapunov functions [94]. In addition, by using the SOS approach, it is possible to analyze the stability of systems with nonlinear polynomial vector fields. Stability analysis tools based on the SOS decomposition for classes of nonlinear systems, hybrid systems, switched systems,

17

and time-delay systems are presented in [85]. In the same reference it is proposed to use PWP Lyapunov functions for hybrid systems, which is a generalization of PWQ Lyapunov functions. However, systems with infinitely fast switching or sliding modes are excluded from the discussion in [85].

Although there is a vast amount of work on stability of switched linear and PWA systems, sliding modes or infinitely fast switching are not usually considered in the literature. Important exceptions are the references [19, 66, 89, 103]. In [19], it is proposed to add the sliding modes and their associated sliding dynamics to the modes of the system before doing the stability analysis. However, this needs a-priori information about the sliding modes of the system, which is typically hard to get. In another approach, an extra condition is introduced in [66, p.64] to extend the analysis to systems with attractive sliding modes. However, one needs to identify potential sets in which sliding modes can occur and then the corresponding condition can be formed and added to the analysis problem. This might again be hard and make the problem complex if there is no previous information about sliding modes. In [103], a synthesis method based on BMIs was proposed for state and output feedback stabilization of PWA systems. The synthesis method includes linear constraints on controller gains to guarantee that sliding modes are not generated at the switching. Finally, reference [89] has addressed sliding modes but has concentrated on the specific case of common quadratic Lyapunov functions for bimodal PWA systems. A question that still remains to be answered is when the necessity to check the existence of unstable sliding modes of a general PWA system can be removed.

Based on the aforementioned limitations, this chapter will present a nonsmooth Lyapunov stability theorem. This theorem applies to PWS systems with continuous or discontinuous vector fields. The theorem states that a sufficient condition for the stability of a PWS system is the existence of a  $C^1$  positive definite function that decreases with time inside each region. The importance of the proposed theorem is to show that by using a  $C^1$  Lyapunov function, no additional condition is needed to address potential sliding modes of PWS systems and, therefore, no a-priori knowledge about sliding modes is necessary.

This chapter is structured as follows. PWS systems are defined in section 2.2. In section 2.3, a stability theorem is stated and proved. Sufficient conditons for monotonicity of nonsmooth functions and dissipativity of PWS systems are described in sections 2.4 and 2.5, respectively. PWA systems and PWA slab systems are then introduced in sections 2.6 and 2.7, respectively. For PWA systems, as the first choice for a  $C^1$  Lyapunov function, a quadratic form is considered in subsection 2.6.1. Then, PWQ Lyapunov functions are discussed in subsection 2.6.2. As a counter example, it is shown that it is possible to find a PWQ positive definite function which is decreasing with time in each region for an unstable PWA system. A proposition for SOS Lyapunov functions is then proved in subsection 2.6.3. It is also shown by an example that it is possible to find an SOS Lyapunov functions. PWP systems are then introduced in section 2.8 and a stability proposition based on polynomial Lyapunov functions is presented.

## 2.2 PWS Systems

The dynamics of a PWS system can be written as

$$\dot{x} = f_i(x), \ x \in \mathcal{R}_i \tag{2.1}$$

where  $x(t) \in \mathcal{X} \subset \mathbb{R}^n$  is the state vector and the initial state is  $x(0) = x_0$ . A subset of the state space  $\mathcal{X}$  is partitioned into M regions,  $\mathcal{R}_i$ ,  $i = 1, \ldots, M$ , such that

$$\cup_{i=1}^{M} \overline{\mathcal{R}}_{i} = \mathcal{X} \tag{2.2}$$

$$\mathcal{R}_i \cap \mathcal{R}_j = \emptyset, \, i \neq j \tag{2.3}$$

where  $\overline{\mathcal{R}}_i$  denotes the closure of  $\mathcal{R}_i$ . The function  $f_i(x) : \overline{\mathcal{R}}_i \to \mathbb{R}^n$  is continuous and locally bounded. The Filippov definition of trajectories is considered for the solution of (2.1) (see [44] and [66]).

**Definition 2.1.** (Filippov solution) A continuous function x(t) is regarded to be a Filippov solution to (2.1) if it is a solution of the differential inclusion

$$\dot{x}(t) \in \mathcal{F}(x) \tag{2.4}$$

where

$$\mathcal{F}(x) \triangleq \mathbf{co}\{f_i(x)|i \in \mathcal{I}(x)\}$$
(2.5)

co stands for the convex hull of a set and

$$\mathcal{I}(x) = \{i | x \in \overline{\mathcal{R}}_i\}.$$
(2.6)

Note that if  $x \in \mathcal{R}_i$ , then

$$\mathcal{F}(x) = \{f_i(x)\}.\tag{2.7}$$

**Example 2.1.** Consider the following simple scalar differential equation

$$\dot{x} = -\operatorname{sgn}(x), x(0) = 1$$
 (2.8)

It can easily be seen that the differential inclusion (2.4) becomes

$$\dot{x} \in -SGN(x) \tag{2.9}$$

where SGN is the set-valued sign function decribed below.

$$SGN(x) = \begin{cases} \{-1\} &, \text{ for } x < 0 \\ \{+1\} &, \text{ for } x > 0 \\ [-1,1] &, \text{ for } x = 0 \end{cases}$$
(2.10)
# 2.3 Lyapunov stability

In this section, a Lyapunov stability theorem is proved for nonsmooth Lyapunov functions. This theorem forms the theoretical framework for using piecewise smooth Lyapunov functions in stability analysis of nonlinear systems. There are other nonsmooth versions of Lyapunov theorems in the literature e.g. [24, 31, 59, 106, 118]. However, certain conditions in these theorems (such as, for example, the conditions on the Dini derivative, the proximal subdifferential or the upper bound of the Lyapunov function) are difficult to check or not needed in the cases described in this thesis. The objective of Theorem 2.1 will thus be to extend the standard Lyapunov stability theorem in [70] to nonsmooth Lyapunov functions and to fit the framework needed in this thesis. To the best of the author's knowledge, this theorem in this exact form does not appear in the literature.

Consider the following autonomous nonlinear system

$$\dot{x}(t) = f(x(t))$$
 (2.11)

where  $x(t) \in \mathbb{R}^n$  is the state vector, the initial state,  $x(0) = x_0$ , is bounded and  $f : \mathcal{X} \to \mathbb{R}^n$  is piecewise continuous and bounded in  $\mathcal{X} \subset \mathbb{R}^n$ . The following theorem describes sufficient conditions for stability of system (2.11) in the sense of Lyapunov based on a continuous Lyapunov function that is not necessarily differentiable everywhere. Because of its importance, the theorem is proved here. The proof combines the proof of the standard Lyapunov theorem in [70] for stability and the proof of the nonsmooth Lyapunov theorem in [31] for asymptotic stability.

**Theorem 2.1.** For nonlinear system (2.11), if there exists a continuous function V(x) such that

$$V(x^{\star}) = 0 \tag{2.12}$$

$$V(x) > 0 \text{ for all } x \neq x^* \text{ in } \mathcal{X}$$

$$(2.13)$$

$$t_1 \le t_2 \Rightarrow V(x(t_1)) \ge V(x(t_2)) \tag{2.14}$$

then  $x = x^*$  is a stable equilibrium point. Moreover if there exists a continuous function W(x) such that

$$W(x^{\star}) = 0 \tag{2.15}$$

$$W(x) > 0 \text{ for all } x \neq x^* \text{ in } \mathcal{X}$$

$$(2.16)$$

$$t_1 \le t_2 \Rightarrow V(x(t_1)) \ge V(x(t_2)) + \int_{t_1}^{t_2} W(x(\tau)) d\tau$$
 (2.17)

and

$$||x|| \to \infty \Rightarrow V(x) \to \infty \tag{2.18}$$

then all trajectories in  $\mathcal{X}$  asymptotically converge to  $x = x^*$ .

*Proof.* For stability, we want to prove

$$\forall \varepsilon > 0, \exists \delta(\varepsilon) > 0 \text{ s.t. } \|x_0 - x^\star\| < \delta \Rightarrow \|x(t) - x^\star\| < \varepsilon, \forall t \ge 0$$
 (2.19)

Following [70], for a given  $\varepsilon > 0$  we choose  $r \in (0, \varepsilon]$  such that

$$\mathcal{B}_r = \{x | \|x - x^\star\| \le r\} \subset \mathcal{X}$$
(2.20)

Let  $\alpha = \min_{\|x-x^{\star}\|=r} V(x)$ . Then  $\alpha > 0$  by (2.13). Take  $\beta \in (0, \alpha)$  and let  $\Omega_{\beta} = \{x \in \mathcal{B}_r | V(x) \leq \beta\}$ . Then  $\Omega_{\beta}$  is in the interior of  $\mathcal{B}_r$  (Figure 2.1). If  $x_0 \in \Omega_{\beta}$  then (2.14) implies that  $x(t) \in \Omega_{\beta}$  for all  $t \geq 0$ . As V(x) is continuous and  $V(x^{\star}) = 0$ , there is a  $\delta > 0$  such that  $\|x - x^{\star}\| \leq \delta \Rightarrow V(x) < \beta$ . Then

$$\mathcal{B}_{\delta} = \{ x | \| x - x^{\star} \| \le \delta \} \subset \Omega_{\beta} \subset \mathcal{B}_{r}$$
(2.21)

and  $x_0 \in \mathcal{B}_{\delta} \Rightarrow x_0 \in \Omega_{\beta} \Rightarrow x(t) \in \Omega_{\beta} \Rightarrow x(t) \in \mathcal{B}_r \Rightarrow x(t) \in \mathcal{B}_{\varepsilon}$ . Therefore

$$\|x_0 - x^\star\| < \delta \Rightarrow \|x(t) - x^\star\| < r \le \varepsilon, \forall t \ge 0$$
(2.22)

This implies that  $x = x^*$  is a stable equilibrium point.

To prove asymptotic stability, following [31], we show that x(t) converges to  $x^*$  as  $t \to \infty$ . It follows from  $x(0) = x_0$  and (2.17) that

$$V(x(t)) + \int_0^t W(x(t)) d\tau \le V(x_0)$$
(2.23)



Figure 2.1: Geometric illustration of sets in the proof of Theorem 2.1.

Then (2.13), (2.16), (2.23) and the fact  $x_0$  is bounded imply that V(x(t)) and  $\int_0^t W(x(t))d\tau$  are bounded. Because V(x(t)) is bounded, it follows from (2.18) that ||x(t)|| is bounded. Since f(x) is bounded in  $\mathcal{X}$ ,  $\dot{x}(t)$  is bounded and x(t) satisfies a global Lipschitz condition on  $t \in [0, +\infty)$  with constant L.

Assume that x(t) fails to converge to  $x^*$ . Then for some  $\varepsilon > 0$  there exists a sequence of points  $t_i$  tending to infinity such that

$$||x(t_i) - x^*|| \ge \varepsilon, \ i = 1, 2, \dots$$
 (2.24)

Without loss of generality, possibly by selecting a subsequence, the sequence  $t_i$  can always be chosen such that

$$|t_{i+1} - t_i| > \frac{\varepsilon}{2L} \tag{2.25}$$

Since ||x(t)|| is bounded, there exists a  $\lambda > 0$  such that  $||x(t) - x^*|| \le \lambda$ . Consider

$$\mathcal{A}_{\left[\frac{\varepsilon}{2},\lambda\right]} = \left\{ x \mid \frac{\varepsilon}{2} \le \|x - x^{\star}\| \le \lambda \right\}$$
(2.26)

 $\mathcal{A}_{[\frac{\varepsilon}{2},\lambda]}$  is not empty since  $\lambda \geq \varepsilon > \frac{\varepsilon}{2}$ . Let  $\eta > 0$  be such that

$$x \in \mathcal{A}_{\left[\frac{\varepsilon}{2},\lambda\right]} \Rightarrow W(x) \ge \eta \tag{2.27}$$

Such  $\eta$  exists because of (2.16) and the fact that  $\mathcal{A}_{[\frac{\varepsilon}{2},\lambda]}$  is not empty. Consider t such that  $|t - t_i| < \frac{\varepsilon}{2L}$ . Since x(t) is globally Lipschitz continuous with constant L,

we have

$$\|x(t) - x(t_i)\| < \frac{\varepsilon}{2} \tag{2.28}$$

Inequalities (2.24), (2.28) and the following triangle inequality

$$\|x(t) - x^{\star}\| \ge \|x(t_i) - x^{\star}\| - \|x(t) - x(t_i)\|$$
(2.29)

imply  $||x(t) - x^*|| > \frac{\varepsilon}{2}$  and consequently  $x(t) \in \mathcal{A}_{[\frac{\varepsilon}{2},\lambda]}$ . Therefore, from (2.27)

$$\int_{t_i - \frac{\varepsilon}{2L}}^{t_i + \frac{\varepsilon}{2L}} W(x(\tau)) d\tau \ge \frac{\eta \varepsilon}{L}$$
(2.30)

and then using (2.25) and (2.16)

$$\int_{t_{i-1}}^{t_{i+1}} W(x(\tau)) d\tau > \frac{\eta \varepsilon}{L}$$
(2.31)

This would imply that  $\int_0^t W(x(t)) d\tau$  diverges as  $t \to \infty$ , which is a contradiction with (2.23) and the conclusion that  $\int_0^t W(x(t)) d\tau$  is bounded. This proves that x(t) converges to  $x^*$  as  $t \to \infty$ .

**Remark 2.1.** In this work, the equilibrium point  $x = x^*$  is said to be globally stable if all trajectories in  $\mathcal{X}$ , the domain of nonlinear system (2.11), asymptotically converge to  $x = x^*$ .

The next section presents conditions to verify the monotonicity of nonsmooth functions. Using these conditions, one can verify (2.14) and (2.17) for nonlinear system (2.11) without needing to know the trajectories of the system.

### 2.4 Monotonicity of nonsmooth functions

In this section, sufficient conditions for monotonicity of nonsmooth functions of state variables along the trajectories of a piecewise smooth system will be provided. In the following, the concept of generalized gradient is introduced. Note that the monotonicity conditions can also be described by the Dini derivative (such as it is done in [106]). However, in this chapter, the theorem of Rademacher [31, p. 93] is used to define the generalized gradient. This definition will be employed to formulate the monotonicity condition for continuous piecewise quadratic and piecewise polynomial functions.

**Definition 2.2.** [31] For a locally Lipschitz continuous function  $V : \mathbb{R}^n \to \mathbb{R}$ , the generalized gradient is defined as

$$\partial_C V(x) = \operatorname{conv}\{\lim_{i \to +\infty} \nabla V(x_i) | x_i \to x, x_i \notin N\}$$
(2.32)

where N is the set of measure zero where the gradient of V does not exist.

**Proposition 2.1.** (Section 2.4 of [24]) Let  $\mathcal{F} : \mathbb{R}^n \to 2^{\mathbb{R}^n} \setminus \emptyset$  be continuous and let  $V : \mathbb{R}^n \to \mathbb{R}$  be Lipschitz continuous. V is nonincreasing along all solutions of

$$\dot{x} \in \mathcal{F}(x) \tag{2.33}$$

if and only if

$$\forall x \in \mathbb{R}^n, \forall f \in \mathcal{F}(x), \max\{p^T f | p \in \partial_C V(x)\} \le 0$$
(2.34)

#### 2.5 Dissipativity

Consider the following piecewise smooth system

$$\dot{x} = f_i(x) + g_i(x)w, \ x \in \mathcal{R}_i$$

$$y = h(x, w)$$
(2.35)

where  $x(t) \in \mathbb{R}^n$  denotes the state,  $w(t) \in \mathbb{R}^{n_w}$  is the exogenous input and  $y(t) \in \mathbb{R}^{n_y}$ is the output. The functions  $f_i(x) : \overline{\mathcal{R}}_i \to \mathbb{R}^n$ ,  $g_i(x) : \overline{\mathcal{R}}_i \to \mathbb{R}^{n \times n_w}$  and h(x, w) : $\mathcal{X} \times \mathbb{R}^{n_w} \to \mathbb{R}^{n_y}$  are continuous in x and locally bounded.

In this section, the notion of *dissipativity* is defined. Roughly speaking, a system is considered dissipative if the amount of energy that the system can provide to its environment is less than what it receives from external sources [110].

**Definition 2.3.** [110] The system (2.35) is said to be dissipative with supply rate W(y, w) and storage function V(x), if V(x) is a nonnegative function such that

$$t_1 \le t_2 \Rightarrow V(x(t_1)) + \int_{t_1}^{t_2} W(y(\tau), w(\tau)) d\tau \ge V(x(t_2))$$
 (2.36)

Two propositions are provided in the following to describe the sufficient conditions for the system (2.35) to be dissipative in two cases of discontinuous and continuous vector fields. The importance of these propositions lies in the fact that to check the dissipativity of the system, it suffices to verify a condition on the storage function, the supply rate and the vector field of the subsystem in each region separately. There is therefore no need to examine the storage function in one region with the vector field of another region, which would make the problem much more complicated.

# 2.5.1 Piecewise smooth systems with discontinuous vector fields

**Proposition 2.2.** (Smooth storage functions) The piecewise smooth system (2.35) is dissipative with a storage function V(x) and a supply rate W(y, w) if V(x) is a nonnegative  $C^1$  function, W(y, w) is a continuous function in y and for all  $x \in \overline{\mathcal{R}}_i$ ,  $i = 1, \ldots, M$  and any  $w \in \mathbb{R}^{n_w}$ 

$$\nabla V(x)^T (f_i(x) + g_i(x)w) \le W(y,w)$$
(2.37)

*Proof.* The inequality (2.37) can be rewritten as

$$\begin{bmatrix} \nabla V(x) \\ -W(y,w) \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} f_i(x) + g_i(x)w \\ 1 \end{bmatrix} \le 0$$
(2.38)

By appending time (t) to the state vector of the system (2.35), we have the following differential equation

$$\begin{bmatrix} \dot{x} \\ \dot{t} \end{bmatrix} = \begin{bmatrix} f_i(x) + g_i(x)w \\ 1 \end{bmatrix}, \ x \in \mathcal{R}_i$$
(2.39)

In the following, using Proposition 2.1 the following function

$$S(x,t) = V(x) - \int_0^t W(y,w) d\tau$$
 (2.40)

is shown to be non-increasing along the trajectories of (2.39). The fact that V(x) is a  $\mathcal{C}^1$  function implies that

$$\partial_c S(x,t) = \mathbf{conv} \left\{ \left[ \begin{array}{c} \nabla V(x(t)) \\ -W(y(t),w(\tau)) \end{array} \right] \middle| \tau \to t \right\}$$
(2.41)

Let x(t) be a Filippov solution of (2.35). Therefore, x(t) is a solution of the following differential inclusion

$$\begin{bmatrix} \dot{x} \\ \dot{t} \end{bmatrix} \in \mathbf{conv} \left\{ \begin{bmatrix} f_i(x(t)) + g_i(x(t))w(\tau) \\ 1 \end{bmatrix} \middle| i \in \mathcal{I}(x), \ \tau \to t \right\}$$
(2.42)

Note that if  $x(t) \in \mathcal{R}_i$  and w(t) is continuous at t, the vector field of (2.35) is continuous and we have

$$\begin{bmatrix} \dot{x} \\ \dot{t} \end{bmatrix} = \begin{bmatrix} f_i(x(t)) + g_i(x(t))w(\tau) \\ 1 \end{bmatrix}$$
(2.43)

Since (2.38) is satisfied for any w, it follows from (2.41) that (2.34) is satisfied for the differential inclusion (2.42). Therefore by Proposition 2.1, S(x,t) is nonincreasing along the trajectories of (2.35) in  $\mathcal{X}$  i.e.

$$t_1 \le t_2 \Rightarrow V(x(t_1)) - \int_0^{t_1} W(y, w) d\tau \ge V(x(t_2)) - \int_0^{t_2} W(y, w) d\tau$$
(2.44)

Therefore (2.36) is satisfied and the system (2.35) is dissipative with storage function V(x) and supply rate W(y, w).

#### 2.5.2 Piecewise smooth systems with continuous vector fields

**Proposition 2.3.** (Piecewise smooth storage functions) The piecewise smooth system (2.35) is dissipative with a storage function V(x) and a supply rate W(y, w) if

• V(x) is a nonnegative continuous function where

$$V(x) = V_i(x), x \in \overline{\mathcal{R}}_i \tag{2.45}$$

where  $V_i : \overline{\mathcal{R}}_i \to \mathbb{R}$  is a  $\mathcal{C}^1$  function,

- W(y, w) is a continuous function in y,
- the vector field of the system (2.35) is continuous in x, i.e. for any  $i, j \in \{1, \ldots, M\}$  such that  $\overline{\mathcal{R}}_i \cap \overline{\mathcal{R}}_j \neq \emptyset$ ,

$$\begin{cases} f_i(x) = f_j(x) \\ g_i(x) = g_j(x) \end{cases}, \ \forall x \in \overline{\mathcal{R}}_i \bigcap \overline{\mathcal{R}}_j \end{cases}$$
(2.46)

• for all  $x \in \overline{\mathcal{R}}_i$ , i = 1, ..., M and any  $w \in \mathbb{R}^{n_w}$ 

$$\nabla V_i(x)^T (f_i(x) + g_i(x)w) \le W(y,w)$$
(2.47)

*Proof.* The inequality (2.47) can be rewritten as

$$\begin{bmatrix} \nabla V_i(x) \\ -W(y,w) \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} f_i(x) + g_i(x)w \\ 1 \end{bmatrix} \le 0$$
 (2.48)

By appending time (t) to the state vector of the system (2.35), we have the following differential equation

$$\begin{bmatrix} \dot{x} \\ \dot{t} \end{bmatrix} = \begin{bmatrix} f_i(x) + g_i(x)w \\ 1 \end{bmatrix}, \ x \in \mathcal{R}_i$$
(2.49)

In the following, using Proposition 2.1 the following function

$$S(x,t) = V_i(x) - \int_0^t W(y,w) d\tau, x \in \overline{\mathcal{R}}_i$$
(2.50)

is shown to be non-increasing along the trajectories of (2.49). The fact that V(x) is a piecewise  $\mathcal{C}^1$  function implies that

$$\partial_c S(x,t) = \mathbf{conv} \left\{ \left[ \begin{array}{c} \nabla V_i(x(t)) \\ -W(y(t),w(\tau)) \end{array} \right] \middle| i \in \mathcal{I}(x), \tau \to t \right\}$$
(2.51)

Let x(t) be a Filippov solution of (2.35). Therefore, x(t) is a solution of the following differential inclusion

$$\begin{bmatrix} \dot{x} \\ \dot{t} \end{bmatrix} \in \mathbf{conv} \left\{ \begin{bmatrix} f_i(x(t)) + g_i(x(t))w(\tau) \\ 1 \end{bmatrix} \middle| i \in \mathcal{I}(x), \ \tau \to t \right\}$$
(2.52)

Consider the following two cases

• If  $x(t) \in \mathcal{R}_i$ , we have

$$\partial_c S(x,t) = \mathbf{conv} \left\{ \left[ \begin{array}{c} \nabla V_i(x(t)) \\ -W(y(t),w(\tau)) \end{array} \right] \middle| \tau \to t \right\}$$
(2.53)

and

$$\begin{bmatrix} \dot{x} \\ \dot{t} \end{bmatrix} \in \mathbf{conv} \left\{ \begin{bmatrix} f_i(x(t)) + g_i(x(t))w(\tau) \\ 1 \end{bmatrix} \middle| \tau \to t \right\}$$
(2.54)

Since (2.47) is satisfied for any w, it follows from (2.53) that (2.34) is satisfied for the differential inclusion (2.54).

• If x(t) is on the boundary of two or more regions i.e.  $x(t) \in \bigcap_{i \in \mathcal{I}(x)} \overline{\mathcal{R}}_i$ , it follows from (2.47) that for any j, k in  $\mathcal{I}(x)$ ,

$$\begin{bmatrix} \nabla V_{j}(x) \\ -W(y,w) \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} f_{j}(x) + g_{j}(x)w \\ 1 \end{bmatrix} \leq 0 \qquad (2.55)$$
$$\begin{bmatrix} \nabla V_{k}(x) \\ -W(y,w) \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} f_{k}(x) + g_{k}(x)w \\ 1 \end{bmatrix} \leq 0 \qquad (2.56)$$

In addition, the continuity condition (2.46) implies that

$$\begin{bmatrix} \nabla V_j(x) \\ -W(y,w) \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} f_k(x) + g_k(x)w \\ 1 \end{bmatrix} = \begin{bmatrix} \nabla V_j(x) \\ -W(y,w) \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} f_j(x) + g_j(x)w \\ 1 \end{bmatrix} \le 0$$
(2.57)

and

$$\begin{bmatrix} \nabla V_k(x) \\ -W(y,w) \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} f_j(x) + g_j(x)w \\ 1 \end{bmatrix} = \begin{bmatrix} \nabla V_k(x) \\ -W(y,w) \end{bmatrix}^{\mathrm{T}} \begin{bmatrix} f_k(x) + g_k(x)w \\ 1 \end{bmatrix} \le 0$$
(2.58)

From (2.55-2.58), it follows that (2.34) is satisfied for the differential inclusion (2.52).

In conclusion, by Proposition 2.1, S(x, t) is nonincreasing along the trajectories of (2.35) in  $\mathcal{X}$  i.e.

$$t_1 \le t_2 \Rightarrow V(x(t_1)) - \int_0^{t_1} W(y, w) d\tau \ge V(x(t_2)) - \int_0^{t_2} W(y, w) d\tau$$
(2.59)

Therefore (2.36) is satisfied and the system (2.35) is dissipative with storage function V(x) and supply rate W(y, w).

### 2.6 PWA Systems

A PWA system is described by

$$\dot{x} = A_i x + a_i, \text{ for } x \in \mathcal{R}_i$$
 (2.60)

where  $A_i \in \mathbb{R}^{n \times n}$ ,  $a_i \in \mathbb{R}^n$  for i = 1, ..., M. It is assumed that  $a_i = 0$  for  $i \in \mathcal{I}(0)$ . Therefore, the origin is an equilibrium point of the system. A subset of the state space  $\mathcal{X} \subset \mathbb{R}^n$  is partitioned into M polytopic regions  $\mathcal{R}_i$ . Each region is constructed as the intersection of a finite number of half spaces

$$\mathcal{R}_i = \{ x | E_i x + e_i \succ 0 \}, \tag{2.61}$$

where  $E_i \in \mathbb{R}^{p_i \times n}$ ,  $e_i \in \mathbb{R}^{p_i}$  and  $\succ$  represents an elementwise inequality. Each polytopic region  $\mathcal{R}_i$  can be outer approximated by a (possibly degenerate) quadratic curve  $\varepsilon_i$ 

$$\mathcal{R}_i \subseteq \varepsilon_i = \{ x | \bar{x}^T \bar{E}_i^T \bar{\Lambda}_i \bar{E}_i \bar{x} > 0 \}$$
(2.62)

where  $\bar{\Lambda}_i \in \mathbb{R}^{(p_i+1)\times(p_i+1)}$  is a matrix with nonnegative entries and

$$\bar{E}_i = \begin{bmatrix} E_i & e_i \\ 0 & 1 \end{bmatrix}$$
(2.63)

A parametric description of the boundaries between two regions  $\mathcal{R}_i$  and  $\mathcal{R}_j$  where  $\overline{\mathcal{R}_i} \cap \overline{\mathcal{R}}_j \neq \emptyset$  can also be obtained as (see [56] and [103] for more details)

$$\overline{\mathcal{R}_i} \cap \overline{\mathcal{R}}_j \subseteq \{x | x = F_{ij}s + f_{ij}, s \in \mathbb{R}^{n-1}\}$$
(2.64)

In this chapter, following [66], the following notation is considered

$$\forall x \in \mathbb{R}^n, \bar{x} = \begin{bmatrix} x\\1 \end{bmatrix}$$
(2.65)

Equation (2.60) can then be written as

$$\dot{\bar{x}}(t) = \bar{A}_i \bar{x}(t), \ x(t) \in \mathcal{R}_i$$
(2.66)

where

$$\bar{A}_i = \begin{bmatrix} A_i & a_i \\ 0 & 0 \end{bmatrix}$$
(2.67)

In the following, three types of candidate Lyapunov functions are examined for stability analysis of PWA systems.

#### 2.6.1 Quadratic Lyapunov Function

Perhaps the simplest candidate for a  $\mathcal{C}^1$  Lyapunov function is the quadratic form

$$V(x) = x^{\mathrm{T}} P x \tag{2.68}$$

where  $P = P^{T} > 0$ . The following proposition describes the sufficient conditions for the stability of the PWA system (2.60) using a quadratic Lyapunov function.

**Proposition 2.4.** (Sufficient conditions for quadratic Lyapunov stability) If for a given decay rate  $\alpha > 0$ , there exists  $P = P^T > 0$  satisfying

$$\begin{cases} PA_i + A_i^T P \leq -\alpha P, & \text{if } a_i = 0 \text{ and } e_i \neq 0 \\\\ PA_i + A_i^T P + E_i^T \Lambda_i E_i \leq -\alpha P, & \text{if } a_i = 0 \text{ and } e_i = 0 \\\\ \bar{P}\bar{A}_i + \bar{A}_i^T \bar{P} + \bar{E}_i^T \bar{\Lambda}_i \bar{E}_i \leq -\alpha \bar{P}, & \text{otherwise} \end{cases}$$
(2.69)

for  $i = 1, \ldots, M$  where

$$\bar{P} = \begin{bmatrix} P & 0_{n \times 1} \\ 0_{1 \times n} & 0 \end{bmatrix}, \qquad (2.70)$$

 $\Lambda_i \in \mathbb{R}^{p_i \times p_i}$  and  $\bar{\Lambda}_i \in \mathbb{R}^{(p_i+1) \times (p_i+1)}$  have nonnegative entries, x = 0 is asymptotically stable for the PWA system (2.60).

*Proof.* Consider  $V(x) = x^{T}Px$  as the candidate Lyapunov function. For this function,  $\nabla V(x) = 2Px$ . In the following, the regions  $\mathcal{R}_{i}$  will be divided into three groups:

1. If  $a_i = 0$  and  $e_i \neq 0$ , we conclude from (2.69) that for all  $x \in \mathbb{R}^n$ 

$$\nabla V(x)^{\mathrm{T}} A_{i} x = 2x^{\mathrm{T}} P A_{i} x$$
$$= x^{\mathrm{T}} (P A_{i} + A_{i}^{\mathrm{T}} P) x$$
$$\leq -\alpha x^{\mathrm{T}} P x = -\alpha V(x) \qquad (2.71)$$

2. If  $a_i = 0$  and  $e_i = 0$ ,  $\overline{\mathcal{R}}_i = \{x | E_i x \ge 0\}$  and for any  $\Lambda_i \in \mathbb{R}^{p_i \times p_i}$  with nonnegative entries and for all  $x \in \overline{\mathcal{R}}_i$ 

$$x^{\mathrm{T}} E_i^{\mathrm{T}} \Lambda_i E_i x \ge 0 \tag{2.72}$$

In this case, (2.69) leads to the following inequality for all  $x \in \overline{\mathcal{R}}_i$ .

$$\nabla V(x)^{\mathrm{T}} A_{i} x = 2x^{\mathrm{T}} P A_{i} x$$

$$= x^{\mathrm{T}} (P A_{i} + A_{i}^{\mathrm{T}} P) x$$

$$\leq -\alpha x^{\mathrm{T}} P x - x^{\mathrm{T}} E_{i}^{\mathrm{T}} \Lambda_{i} E_{i} x$$

$$\leq -\alpha x^{\mathrm{T}} P x = -\alpha V(x) \qquad (2.73)$$

3. If  $a_i \neq 0$ , we have  $\overline{\mathcal{R}}_i = \{x | \overline{E}_i \overline{x} \geq 0\}$  and similarly to the previous case, condition (2.69) implies that for all  $x \in \overline{\mathcal{R}}_i$ 

$$\nabla V(x)^{\mathrm{T}}(A_{i}x + a_{i}) = 2\bar{x}^{\mathrm{T}}\bar{P}\bar{A}_{i}\bar{x}$$

$$= \bar{x}^{\mathrm{T}}(\bar{P}\bar{A}_{i} + \bar{A}_{i}^{\mathrm{T}}\bar{P})\bar{x}$$

$$\leq -\alpha\bar{x}^{\mathrm{T}}\bar{P}\bar{x} - \bar{x}^{\mathrm{T}}\bar{E}_{i}^{\mathrm{T}}\bar{\Lambda}_{i}\bar{E}_{i}\bar{x}$$

$$\leq -\alpha\bar{x}^{\mathrm{T}}\bar{P}\bar{x} = -\alpha x^{\mathrm{T}}Px = -\alpha V(x) \qquad (2.74)$$

In summary, for all  $x \in \overline{\mathcal{R}}_i, i = 1, \ldots, M$ ,

$$\nabla V(x)^{\mathrm{T}}(A_i x + a_i) \le -\alpha V(x) \tag{2.75}$$

Therefore using Proposition 2.2, the system (2.60) is dissipative with the storage function V(x) and the supply rate  $-\alpha V(x)$ . Therefore, considering (2.40),  $S(x,t) = V(x) + \int_0^t \alpha V(x(\tau)) d\tau$  is nonincreasing along the trajectories of (2.60). This implies that the conditions of Theorem 2.1 are satisfied and all trajectories of the PWA system (2.60) in  $\mathcal{X}$  asymptotically converge to x = 0.

**Remark 2.2.** In Proposition 2.4, the origin is not required to be the equilibrium point of all the subsystems the PWA system (2.60). This makes Proposition 2.4 different from the common Lyapunov function approach in [75] which requires the origin to be the equilibrium point for all vector fields of the system.

Proposition 2.4 provides sufficient conditions for quadratic stability of PWA systems as a set of linear matrix inequalities (LMIs). LMIs can be solved efficiently using interior point algorithms implemented in software packages such as Yalmip [76] and SeDuMi [121]. The following examples illustrate Proposition 2.4.

**Example 2.2.** Consider the following Piecewise Linear (PWL) system:

$$\dot{x} = \begin{cases} A_1 x, & x_2 > 0 \\ \\ A_2 x, & x_2 < 0 \end{cases}$$
(2.76)

where

$$A_{1} = \begin{bmatrix} -1 & -2 \\ 2 & -2 \end{bmatrix}, A_{2} = \begin{bmatrix} -1 & 2 \\ -2 & -2 \end{bmatrix}$$
(2.77)

For this system, we have:

$$E_1 = \begin{bmatrix} 0 & 1 \end{bmatrix}, E_2 = \begin{bmatrix} 0 & -1 \end{bmatrix}$$
(2.78)

Solving the following LMIs based on Proposition 2.4

$$\begin{cases}
A_{1}^{T}P + PA_{1} + \lambda_{1}E_{1}^{T}E_{1} < -\alpha P \\
A_{2}^{T}P + PA_{2} + \lambda_{2}E_{2}^{T}E_{2} < -\alpha P \\
P > 0, \ \lambda_{1} > 0, \ \lambda_{2} > 0
\end{cases}$$
(2.79)

where  $\alpha = 0.1$ , yields

$$P = \begin{bmatrix} 0.6002 & 0\\ 0 & 0.5817 \end{bmatrix}, \ \lambda_1 = 1.1329, \ \lambda_2 = 1.1329.$$
 (2.80)

Therefore x = 0 is asymptotically stable. It is interesting to note that this system has an attractive sliding mode on the negative side of the  $x_1$  axis (Fig. 2.2). However, no separate condition was considered to check the existence or stability of the sliding mode.

**Example 2.3.** Consider the following system:

$$\dot{x} = \begin{cases} A_1 x, & x_2 > 0 \\ \\ A_2 x, & x_2 < 0 \end{cases}$$
(2.81)

where

$$A_{1} = \begin{bmatrix} 1 & -2 \\ 2 & -2 \end{bmatrix}, \quad A_{2} = \begin{bmatrix} 1 & 2 \\ -2 & -2 \end{bmatrix}$$
(2.82)



Figure 2.2: Trajectories of a stable PWA system with a sliding mode



Figure 2.3: Trajectories of an unstable PWA system with a sliding mode

For this system,  $E_1$  and  $E_2$  are defined as in Example 2.2. The LMI set (2.79) is infeasible in this case. In fact, although  $A_1$  and  $A_2$  are Hurwitz, there exists an unstable sliding mode and system (2.81) is unstable (Fig. 2.3).

#### 2.6.2 Piecewise Quadratic Lyapunov Function

For stability analysis of PWA systems, PWQ functions are less conservative than quadratic Lyapunov functions [66]. However, PWA systems with sliding modes are not usually considered in the literature of multiple Lyapunov functions. The reason is that the existence of a continuous positive definite PWQ function that decreases with time inside the regions is not a sufficient condition for stability of a PWA system. This is shown by the following counter-example.

**Example 2.4.** Consider the PWA system (2.81) and the following PWQ Lyapunov function candidate.

$$V(x) = \begin{cases} x^T P_1 x, & x_2 \ge 0 \\ \\ x^T P_2 x, & x_2 \le 0 \end{cases}$$
(2.83)

The following set of constraints is a sufficient condition for (2.83) to be continuous, positive definite and decreasing with time inside the regions.

• Continuous at  $x_2 = 0$ :

$$\begin{bmatrix} 1 & 0 \end{bmatrix} (P_1 - P_2) \begin{bmatrix} 1 \\ 0 \end{bmatrix} = 0$$
 (2.84)

• Positive definite:

$$P_1 > 0, \ P_2 > 0 \tag{2.85}$$

• Decreasing with time inside the regions:

$$\begin{cases}
A_1^T P_1 + P_1 A_1 + \lambda_1 E_1^T E_1 < -\alpha P_1 \\
A_2^T P_2 + P_2 A_2 + \lambda_2 E_2^T E_2 < -\alpha P_2 \\
\lambda_1 > 0, \ \lambda_2 > 0, \ \alpha = 0.1
\end{cases}$$
(2.86)

One solution of the above problem is

$$P_{1} = \begin{bmatrix} 1.8073 & -1.0745 \\ -1.0745 & 1.4261 \end{bmatrix},$$

$$P_{2} = \begin{bmatrix} 1.8073 & 1.0745 \\ 1.0745 & 1.4261 \end{bmatrix},$$

$$\lambda_{1} = 0.5755, \lambda_{2} = 0.5755.$$
(2.87)

V(x) in (2.83) is a continuous PWQ positive definite function that decreases with time inside the regions of system (2.81). However, system (2.81) is unstable. Therefore, the existence of such a function is not a sufficient condition for stability.

In [66, p.64], an extra condition is introduced for PWA systems which have sliding modes. However, the limitation of this method is that it requires previous knowledge of geometrical properties of the sliding modes. One way to solve this problem is to use  $C^1$  PWQ Lyapunov functions, which is also proposed in [66, p. 84].

Consider the piecewise quadratic candidate Lyapunov function continuous at the boundaries and defined in  $\mathcal{X}$  by the expression

$$V(x) = \bar{x}^T \bar{P}_i \bar{x}, \text{ for } x \in \overline{\mathcal{R}}_i$$
(2.88)

where  $\bar{P}_i = \bar{P}_i^T \in \mathbb{R}^{(n+1) \times (n+1)}$  and

$$\bar{P}_i = \begin{bmatrix} P_i & q_i \\ q_i^T & r_i \end{bmatrix}$$
(2.89)

with  $P_i \in \mathbb{R}^{n \times n}$ ,  $q_i \in \mathbb{R}^n$  and  $r_i \in \mathbb{R}$ . To simplify the notation, define

$$\bar{F}_{ij} = \begin{bmatrix} F_{ij} & f_{ij} \\ 0 & 1 \end{bmatrix}$$
(2.90)

The following proposition describes the sufficient conditions for the stability of the PWA system (2.60) based on a PWQ Lyapunov function.

**Proposition 2.5.** (Sufficient conditions for PWQ Lyapunov stability) Let there exist matrices  $\bar{P}_i = \bar{P}_i^T$  defined in (2.89),  $Z_i$ ,  $\bar{Z}_i$ ,  $\Lambda_i$  and  $\bar{\Lambda}_i$  that verify the following conditions for all i = 1, ..., M and a given decay rate  $\alpha > 0$ 

• Conditions on the vector field:

$$a_i = 0, \ if \ 0 \in \overline{\mathcal{R}}_i$$
 (2.91)

$$(\bar{A}_i - \bar{A}_j)\bar{F}_{ij} = 0, \ if \ \overline{\mathcal{R}}_i \bigcap \overline{\mathcal{R}}_j \neq \emptyset$$
 (2.92)

• Continuity of the Lyapunov function:

$$\bar{F}_{ij}^{T}(\bar{P}_{i}-\bar{P}_{j})\bar{F}_{ij}=0, \quad if \quad \overline{\mathcal{R}}_{i}\bigcap\overline{\mathcal{R}}_{j}\neq\emptyset$$
(2.93)

• Positive definiteness of the Lyapunov function:

1

$$q_i = 0, \ r_i = 0, \ if \ 0 \in \overline{\mathcal{R}}_i \tag{2.94}$$

$$P_i \ge \epsilon I, \ if \ 0 \in \overline{\mathcal{R}}_i \ and \ e_i \ne 0$$
 (2.95)

$$\begin{cases} Z_i \succeq 0\\ P_i - E_i^T Z_i E_i \ge \epsilon I \end{cases}, \text{ if } 0 \in \overline{\mathcal{R}}_i \text{ and } e_i = 0 \qquad (2.96) \end{cases}$$

$$\begin{cases} \bar{Z}_i \succeq 0\\ \bar{P}_i - \bar{E}_i^T Z_i \bar{E}_i \ge \epsilon \begin{bmatrix} I & 0\\ 0 & 0 \end{bmatrix}, & \text{if } 0 \notin \overline{\mathcal{R}}_i \end{cases}$$
(2.97)

• Monotonicity of the Lyapunov function:

$$P_i A_i + A_i^T P_i \le -\alpha P_i, \text{ if } 0 \in \overline{\mathcal{R}}_i \text{ and } e_i \ne 0$$
 (2.98)

$$\begin{cases} \Lambda_i \succeq 0\\ P_i A_i + A_i^T P_i + E_i^T \Lambda_i E_i \leq -\alpha P_i \end{cases}, & if \ 0 \in \overline{\mathcal{R}}_i \ and \ e_i = 0 \quad (2.99) \end{cases}$$

$$\begin{cases} \bar{\Lambda}_i \succeq 0 \\ \bar{P}_i \bar{A}_i + \bar{A}_i^T \bar{P}_i + \bar{E}_i^T \bar{\Lambda}_i \bar{E}_i \leq -\alpha \bar{P}_i \end{cases}, & \text{if } 0 \notin \overline{\mathcal{R}}_i \end{cases}$$
(2.100)

Then all the trajectories of (2.60) in  $\mathcal{X}$  asymptotically converge to x = 0.

Proof. Consider  $V(x) = \overline{x}^T \overline{P}_i \overline{x}$  for  $x \in \overline{\mathcal{R}}_i$  as the candidate Lyapunov function. It follows from (2.64) and (2.93) that for any  $x \in \overline{\mathcal{R}}_i \bigcap \overline{\mathcal{R}}_j$ ,  $V_i(x) = V_j(x)$ . Therefore V(x) is continuous over  $\mathcal{X}$ . In addition, constraint (2.94) implies that V(0) = 0. The rest of the proof is divided into three parts:

1. If  $a_i = 0$  and  $e_i \neq 0$ , we conclude from (2.95) that for all  $x \neq 0$  in  $\mathcal{R}_i$ 

$$V(x) = x^{\mathrm{T}} P_i x \ge \epsilon ||x||^2 > 0$$
(2.101)

and from (2.98) that for all  $x \neq 0$ 

١

$$\nabla V_i(x)^{\mathrm{T}} A_i x = 2x^{\mathrm{T}} P_i A_i x$$
  
=  $x^{\mathrm{T}} (P_i A_i + A_i^{\mathrm{T}} P_i) x$   
 $\leq -\alpha x^{\mathrm{T}} P_i x = -\alpha V(x)$  (2.102)

2. If  $a_i = 0$  and  $e_i = 0$ , we have  $\overline{\mathcal{R}}_i = \{x | E_i x \ge 0\}$  and for any  $Z_i \in \mathbb{R}^{p_i \times p_i}$  and  $\Lambda_i \in \mathbb{R}^{p_i \times p_i}$  with nonnegative entries and for all  $x \in \overline{\mathcal{R}}_i$ 

$$x^{\mathrm{T}} E_i^{\mathrm{T}} Z_i E_i x \geq 0 \tag{2.103}$$

$$x^{\mathrm{T}} E_i^{\mathrm{T}} \Lambda_i E_i x \geq 0 \tag{2.104}$$

In this case, (2.96) leads to the following inequality for all  $x \neq 0$  in  $\overline{\mathcal{R}}_i$ 

$$V_{i}(x) = x^{\mathrm{T}} P_{i} x$$
  

$$> x^{\mathrm{T}} E_{i}^{\mathrm{T}} Z_{i} E_{i} x + \epsilon ||x||^{2}$$
  

$$\geq \epsilon ||x||^{2} > 0 \qquad (2.105)$$

and (2.99) leads to the following inequality for all  $x \neq 0$  in  $\overline{\mathcal{R}}_i$ 

$$\nabla V_{i}(x)^{\mathrm{T}}A_{i}x = 2x^{\mathrm{T}}P_{i}A_{i}x$$

$$= x^{\mathrm{T}}(P_{i}A_{i} + A_{i}^{\mathrm{T}}P_{i})x$$

$$\leq -\alpha x^{\mathrm{T}}P_{i}x - x^{\mathrm{T}}E_{i}^{\mathrm{T}}\Lambda_{i}E_{i}x$$

$$\leq -\alpha x^{\mathrm{T}}P_{i}x = -\alpha V(x) \qquad (2.106)$$

3. If  $a_i \neq 0$ , we have  $\overline{\mathcal{R}}_i = \{x | \overline{E}_i \overline{x} \geq 0\}$  and similarly to the previous case, condition (2.97) implies that for all  $x \neq 0$  in  $\overline{\mathcal{R}}_i$ 

$$V_{i}(x) = \bar{x}^{\mathrm{T}} \bar{P}_{i} \bar{x}$$

$$> \bar{x}^{\mathrm{T}} \bar{E}_{i}^{\mathrm{T}} \bar{Z}_{i} \bar{E}_{i} \bar{x} + \epsilon ||x||^{2}$$

$$\geq \epsilon ||x||^{2} > 0 \qquad (2.107)$$

and condition (2.100) implies that for all  $x \neq 0$  in  $\overline{\mathcal{R}}_i$ 

$$\nabla V_i(x)^{\mathrm{T}}(A_i x + a_i) = 2\bar{x}^{\mathrm{T}} \bar{P}_i \bar{A}_i \bar{x}$$

$$= \bar{x}^{\mathrm{T}} (\bar{P}_i \bar{A}_i + \bar{A}_i^{\mathrm{T}} \bar{P}_i) \bar{x}$$

$$\leq -\alpha \bar{x}^{\mathrm{T}} \bar{P}_i \bar{x} - \bar{x}^{\mathrm{T}} \bar{E}_i^{\mathrm{T}} \bar{\Lambda}_i \bar{E}_i \bar{x}$$

$$\leq -\alpha \bar{x}^{\mathrm{T}} \bar{P}_i \bar{x} = -\alpha V(x) \qquad (2.108)$$

In summary, for all  $x \in \overline{\mathcal{R}}_i$  and for  $i = 1, \ldots, M$ ,

$$V_i(x) \ge \epsilon \|x\|^2 \tag{2.109}$$

$$\nabla V_i(x)^{\mathrm{T}}(A_i x + a_i) \le -\alpha V_i(x) \tag{2.110}$$

Therefore using Proposition 2.3, the system (2.60) is dissipative with the storage function V(x) and the supply rate  $-\alpha V(x)$ . Therefore  $S(x,t) = V(x) + \int_0^t \alpha V(x(\tau)) d\tau$  is nonincreasing along the trajectories of (2.60). This implies that the conditions of Theorem 2.1 are satisfied and all trajectories of the PWA system (2.60) in  $\mathcal{X}$  asymptotically converge to x = 0.

PWQ Lyapunov functions are less conservative than quadratic Lyapunov functions. However, it is stated in [66] that for the following PWA system, it is not possible to find a  $C^1$  PWQ Lyapunov function although the system is stable.

$$\dot{x} = \begin{cases} A_1 x = \begin{bmatrix} -2 & -2 \\ 4 & 1 \end{bmatrix} x, \quad x_2 > 0 \\ A_2 x = \begin{bmatrix} -2 & 2 \\ -4 & 1 \end{bmatrix} x, \quad x_2 < 0 \end{cases}$$
(2.111)

In the next section, it is shown that a polynomial Lyapunov function exists for this system and can be found using the proposed method in section 2.6.3.

#### 2.6.3 Sum of Squares Lyapunov Function

In this section, it is proposed to consider a sum of squares polynomial as a Lyapunov function candidate. For a tutorial about recent system analysis techniques based on the sum of squares decomposition see [85]. A sum of squares polynomial is defined in the following.

**Definition 2.4.** [94] A multivariate polynomial  $p(x_1, \ldots, x_n) \triangleq p(x)$  is a sum of squares, if there exist polynomials  $p_1(x), \ldots, p_m(x)$  such that

$$p(x) = \sum_{i=1}^{m} p_i^2(x).$$
(2.112)

SOS polynomials p(x) are globally nonnegative. Although verifying nonnegativity of a polynomial is an NP-hard problem [81], the SOS condition can be formulated as a convex problem in polynomial coefficients [94]. However, note that not all nonnegative polynomials are SOS. In the following proposition, nonnegativity of SOS polynomials is used to construct a Lyapunov function for PWA systems.

**Proposition 2.6.** If for the PWA system (2.60), there exists a polynomial V(x) satisfying

$$V(x) - \lambda(||x||^2)$$
 is SOS. (2.113)

$$-\nabla V(x)^{T}(A_{i}x+a_{i})-\Gamma_{i}(x)^{T}(E_{i}x+e_{i})-\alpha V(x) \text{ is SOS for all } i. \qquad (2.114)$$

where  $\lambda : \mathbb{R}^+ \to \mathbb{R}^+$  is a strictly increasing polynomial function,  $\lambda(0) = 0$ ,  $\alpha > 0$ and  $\Gamma_i : \mathbb{R}^n \to \mathbb{R}^{p_i \times 1}$  is a vector of SOS polynomials, x = 0 is asymptotically stable. Proof. Conditions (2.113) imply

$$V(x) \ge \lambda(\|x\|^2) \tag{2.115}$$

Condition (2.114) leads to the following inequality

$$\nabla V(x)^{\mathrm{T}}(A_i x + a_i) \le -\Gamma_i(x)^{\mathrm{T}}(E_i x + e_i) - \alpha V(x)$$
(2.116)

Since  $\Gamma_i(x)$  is a vector of SOS polynomials and  $E_i x + e_i \ge 0$  for all x in  $\overline{\mathcal{R}}_i$ , we have

$$\Gamma_i(x)^{\mathrm{T}}(E_i x + e_i) \ge 0, \ \forall x \in \overline{\mathcal{R}}_i$$
(2.117)

Therefore (2.116) and (2.117) imply

$$\nabla V(x)^{\mathrm{T}}(A_{i}x + a_{i}) \leq -\alpha V(x) < 0, \ \forall x \in \overline{\mathcal{R}}_{i}, x \neq 0$$
(2.118)

Thus, using Proposition 2.2, the system (2.60) is dissipative with the storage functinon V(x) and the supply rate  $-\alpha V(x)$ . Therefore, considering (2.40),  $S(x,t) = V(x) + \int_0^t \alpha V(x(\tau)) d\tau$  is nonincreasing along the trajectories of (2.60). This implies that the conditions of Theorem 2.1 are satisfied and all trajectories of the PWA system (2.60) in  $\mathcal{X}$  asymptotically converge to x = 0.

**Remark 2.3.** The parameters of the Lyapunov function can be computed by solving the SOS program in Proposition 2.6 using Yalmip [76] and SeDuMi [121].

**Example 2.5.** Consider the PWA system (2.111). There is no quadratic or  $C^1$  PWQ Lyapunov function for this system [66, p.84]. However, by solving the following SOS program we can find a sixth order polynomial Lyapunov function.

 $V(x) - 0.001 ||x||^2 \text{ is SOS.}$ -\nabla V.(A\_1x) - \Gamma\_1(x)(x\_2) - 0.01V(x) \text{ is SOS.} -\nabla V.(A\_2x) - \Gamma\_2(x)(-x\_2) - 0.01V(x) \text{ is SOS.}

where  $\Gamma_1(.)$  and  $\Gamma_2(.)$  are fourth order SOS polynomials. This is a convex problem and can be solved by SOSTOOLS [95] or Yalmip [76]. One feasible solution to the problem is



Figure 2.4: The SOS Lyapunov function

$$V(x) = (-0.0589x_1 - 0.0567x_1^2 - 0.00143x_1x_2 - 0.02x_2^2)$$
  
-0.00269x\_1^3 - 0.00147x\_1x\_2^2)^2 + (0.00102x\_1 - 0.0323x\_2)  
-0.0651x\_1x\_2 - 0.00265x\_1^2x\_2)^2 + (-0.0156x\_1 + 0.0253x\_1^2 - 0.0257x\_2^2)  
+0.00182x\_1^3 - 0.00207x\_1x\_2^2)^2 + (+0.00717x\_1 - 0.00439x\_1^2 - 0.00863x\_2^2)^2 + (-0.00957x\_2 + 0.00474x\_1x\_2)^2 + (-0.0036x\_1^2x\_2 + 0.00118x\_2^3)^2 + (-0.00166x\_1^3 + 0.00232x\_1x\_2^2)^2 + (0.00124x\_1^3)^2 + (0.00104x\_2^3)^2 (2.119)

This SOS Lyapunov function is shown in Fig. 2.4. Trajectories of the system (2.111) and contours of the SOS Lyapunov function are shown in Fig. 2.5. Notice that there is a stable sliding mode in this system.



Figure 2.5: Trajectories of a stable PWA system (black) and contours of the Lyapunov function (gray)

# 2.7 PWA Slab Systems

A PWA slab system can be described by

$$\dot{x} = A_i x + a_i, \quad \text{for } x \in \mathcal{R}_i \tag{2.120}$$

where  $A_i \in \mathbb{R}^{n \times n}$ ,  $a_i \in \mathbb{R}^n$ . It is assumed that  $a_i = 0$  for  $i \in \mathcal{I}(0)$ . Therefore, the origin is an equilibrium point of the system. The slab regions  $\mathcal{R}_i$ ,  $i = 1, \ldots, M$ partitioning a slab subset of the state space  $\mathcal{X} \subset \mathbb{R}^n$  are defined as

$$\mathcal{R}_i = \{ x \mid \sigma_i < C_{\mathcal{R}} x < \sigma_{i+1} \}, \tag{2.121}$$

where  $C_{\mathcal{R}} \in \mathbb{R}^{1 \times n}$  and  $\sigma_i$  for  $i = 1, \ldots, M + 1$  are scalars such that

$$\sigma_1 < \sigma_2 < \ldots < \sigma_{M+1} \tag{2.122}$$

Each slab region can alternatively be described by the following degenerate ellipsoid

$$\mathcal{R}_i = \{x \mid ||L_i x + l_i|| < 1\}$$
(2.123)

where  $L_i = 2C_{\mathcal{R}}/(\sigma_{i+1} - \sigma_i)$  and  $l_i = -(\sigma_{i+1} + \sigma_i)/(\sigma_{i+1} - \sigma_i)$ .

We are interested to know if all possible trajectories in  $\mathcal{X}$  asymptotically converge to the origin. Note that the right-hand-side of (2.120) is not necessarily continuous and therefore there might exist attractive sliding modes. The following proposition provides sufficient conditions for the stability of system (2.120) based on Theorem 2.1.

**Proposition 2.7.** All trajectories of the PWA slab system (2.120) in  $\mathcal{X}$  asymptotically converge to x = 0 if for a given decay rate  $\alpha > 0$ , there exist  $P \in \mathbb{R}^{n \times n}$  and  $\lambda_i \in \mathbb{R}$  for i = 1, ..., M such that

$$P > 0,$$
 (2.124)

$$A_i^T P + P A_i + \alpha P < 0, \ \forall i \in \mathcal{I}(0), \tag{2.125}$$

$$\begin{cases} \lambda_{i} < 0, \\ \begin{bmatrix} A_{i}^{T}P + PA_{i} + \alpha P + \lambda_{i}L_{i}^{T}L_{i} & Pa_{i} + \lambda_{i}l_{i}L_{i}^{T} \\ a_{i}^{T}P + \lambda_{i}l_{i}L_{i} & \lambda_{i}(l_{i}^{2} - 1) \end{bmatrix} < 0, \end{cases} \qquad \text{for } i \notin \mathcal{I}(0) \qquad (2.126)$$

*Proof.* Consider the candidate Lyapunov function  $V(x) = x^{T}Px$  for the PWA slab system (2.120) where P > 0. Consider the following function

$$S(x,t) = V(x) + \int_0^t \alpha V(x(\tau)) d\tau,$$
 (2.127)

where  $\alpha > 0$ . In the following, we will show that S(x, t) is nonincreasing along the trajectories of (2.120):

1. For  $x \in \overline{\mathcal{R}}_i$  where  $i \in \mathcal{I}(0)$ , multiplying the inequality (2.125) by  $x^{\mathrm{T}}$  and x from left and right, respectively, implies

$$\nabla V(x)^{\mathrm{T}} A_{i} x + \alpha V(x) < 0, \text{ for } x \in \overline{\mathcal{R}}_{i}, i \in \mathcal{I}(0)$$
(2.128)

2. For  $x \in \overline{\mathcal{R}}_i$  where  $i \notin \mathcal{I}(0)$ , it follows from the constraint (2.126) that

$$\nabla V(x)^{\mathrm{T}}(A_{i}x+a_{i})+\alpha V(x)+\lambda_{i}(\|L_{i}x-l_{i}\|^{2}-1)<0, \qquad (2.129)$$

Since  $\lambda_i < 0$ , conditions (2.129) and (2.123) imply

$$\nabla V(x)^{\mathrm{T}}(A_{i}x + a_{i}) + \alpha V(x) < 0, \text{ for } x \in \overline{\mathcal{R}}_{i}, i \notin \mathcal{I}(0)$$
(2.130)

Now, it follows from (2.128), (2.130) and Proposition 2.2 that the system (2.120) is dissipative with the storage function V(x) and the supply rate  $-\alpha V(x)$ . Therefore S(x,t) is nonincreasing along the trajectories of (2.120). This implies that the conditions of Theorem 2.1 are satisfied and all trajectories of the PWA slab system (2.120) in  $\mathcal{X}$  asymptotically converge to x = 0.

#### 2.8 PWP Systems

The dynamics of a PWP system can be written as follows.

$$\dot{x}(t) = f_i(x(t)), \text{ if } x(t) \in \mathcal{P}_i$$
(2.131)

where  $x(t) \in \mathbb{R}^n$  denotes the state vector and  $f_i(x) \in \mathbb{R}^n$  are polynomial functions of x. The regions  $\mathcal{P}_i$ ,  $i \in \mathcal{I} = \{1, \ldots, M\}$ , partition a subset of the state space  $\mathcal{X} \subset \mathbb{R}^n$  such that  $\bigcup_{i=1}^M \overline{\mathcal{P}}_i = \mathcal{X}, \ \mathcal{P}_i \cap \mathcal{P}_j = \emptyset, \ i \neq j$ , where  $\overline{\mathcal{P}}_i$  denotes the closure of  $\mathcal{P}_i$ . Each region is described by

$$\mathcal{P}_i = \{x | E_i(x) \succ 0\} \tag{2.132}$$

where  $E_i(x) \in \mathbb{R}^{p_i}$  is a vector polynomial function of x and  $\succ$  represents an elementwise inequality.

**Proposition 2.8.** If for the PWP system (2.131), there exists a polynomial V(x) satisfying

$$V(x) - \lambda(||x||^2)$$
 is SOS. (2.133)

$$-\nabla V(x)^T f_i(x) - \Gamma_i^T(x) E_i(x) - \alpha V(x) \text{ is SOS for all } i.$$
(2.134)

where  $\lambda : \mathbb{R}^+ \to \mathbb{R}^+$  is a strictly increasing polynomial function,  $\lambda(0) = 0$ ,  $\alpha > 0$ and  $\Gamma_i : \mathbb{R}^n \to \mathbb{R}^{p_i \times 1}$  is a vector of SOS polynomials, x = 0 is asymptotically stable. *Proof.* Condition (2.133) implies

$$V(x) \ge \lambda(\|x\|^2) \tag{2.135}$$

Condition (2.134) leads to the following inequality

$$\nabla V(x)^{\mathrm{T}} f_i(x) \le -\Gamma_i^{\mathrm{T}}(x) E_i(x) - \alpha V(x)$$
(2.136)

Since  $\Gamma_i(x)$  is a vector of SOS polynomials and  $E_i(x) \ge 0$  for all x in  $\overline{\mathcal{P}}_i$ , we have

$$\Gamma_i^{\mathrm{T}}(x)E_i(x) \ge 0, \ \forall x \in \overline{\mathcal{P}}_i$$
(2.137)

Therefore (2.136) and (2.137) imply

$$\nabla V(x)^{\mathrm{T}} f_i(x) \le -\alpha V(x) < 0, \ \forall x \in \overline{\mathcal{R}}_i, x \ne 0$$
(2.138)

Thus, using Proposition 2.2, the system (2.131) is dissipative with the storage function V(x) and the supply rate  $-\alpha V(x)$ . Therefore, considerong (2.40),  $S(x,t) = V(x) + \int_0^t \alpha V(x(\tau)) d\tau$  is nonincreasing along the trajectories of (2.131). This implies that the conditions of Theorem 2.1 are satisfied and all trajectories of the PWP system (2.131) in  $\mathcal{X}$  asymptotically converge to x = 0.

## 2.9 Conclusions

In this chapter a general nonsmooth theorem for stability of nonlinear systems was stated and proved. Then, sufficient conditions for stability of PWA slab systems, PWA systems and PWP systems were formulated as convex problems subject to LMIs. The importance of the results of this chapter is to show that sufficient conditions for the stability of PWA and PWP systems can be formed without any need for a-priori information about attractive sliding modes on switching surfaces. This is very important given that such information is very difficult to obtain for complex systems and, even when this information is available, it might lead to unnecessary conservativeness.

# Chapter 3

# Extension of local linear controllers to global piecewise affine controllers for uncertain nonlinear systems

A two-step controller synthesis method is proposed in this chapter for a class of uncertain nonlinear systems described by piecewise affine differential inclusions. In the first step, a robust linear controller is designed for the linear differential inclusion that describes the dynamics of the nonlinear system close to the equilibrium point. In the second step, a stabilizing piecewise affine controller is designed that coincides with the linear controller in a region around the equilibrium point. The proposed method has two objectives: global stability and local performance. It thus enables to use well known techniques in linear control design for local stability and performance while delivering a global piecewise affine controller that is guaranteed to stabilize the nonlinear system. The new method will be applied to numerical examples.

## 3.1 Introduction

Linear control theory provides a variety of well established tools to guarantee robust stability and performance [36]. The controller is, however, valid only locally if the controller is designed for the linearization of a nonlinear system. In fact, the linear controller may not even stabilize the nonlinear system if the initial condition is far from the linearization point. On the other hand, most of the methods in nonlinear control theory address global asymptotic stability but not necessarily performance. Designing a controller that has both a large region of attraction and a good local performance is therefore one of the most interesting research problems in nonlinear control theory [80]. Having this problem in mind, a two-step method is proposed in this chapter to design a piecewise affine (PWA) controller for uncertain nonlinear systems described by piecewise affine differential inclusions (PWADI). The objective of the proposed method is to design a controller to satisfy a local performance requirement and to globally stabilize the nonlinear system. The method extends a linear controller designed for performance to a globally stabilizing PWA controller. One of the main advantages of this method is that it can be employed in many practical problems for which linear controllers currently exist without changing the local performance of the system.

The structure of the proposed method is shown in Figure 3.1. In the first step, a robust linear controller is designed for the linear differential inclusion (LDI) that approximates the local behaviour of the nonlinear system in a neighbourhood of the desired operating point. Then, a PWA controller that coincides with the linear controller in a region around the equilibrium point and globally stabilizes the nonlinear system is designed in the second step. Since the design approach is based on finding a piecewise quadratic Lyapunov function, it is only approximate in the sense that there is no guarantee that a Lyapunov function can be found. If one is found, global stability is guaranteed. Otherwise, the method is inconclusive. In spite of their approximate nature, Lyapunov-based methods for PWA controller design appear to work well in practice and have been widely used in the literature [40, 56, 66, 96, 103, 104].



Figure 3.1: Structure of the proposed PWA controller design method

The main result of this chapter is proved in Theorem 3.2. The contribution of this result is to provide the theoretical framework for extending a local linear controller to a global PWA controller based on piecewise quadratic Lyapunov functions. In previous research, reference [103] has also used piecewise quadratic Lyapunov functions to synthesize PWA controllers. However, the method of [103] does not enable one to extend a local linear controller to a global PWA controller. Futhermore, it is assumed in [103] that there is one equilibrium point for the dynamic equations of each region. The equilibrium points of all regions are then selected a-priori by solving an optimization problem. It is also required that each of the equilibrium points be the extrema of the corresponding sector of any candidate Lyapunov function. By contrast, Theorem 3.2 now shows that it is in fact not necessary to compute those equilibrium points. This has the important advantage of relieving the designer from this tedious and non-intuitive task.

Note that a PWQ function is not necessarily differentiable everywhere and

therefore it is a nonsmooth function. Despite this fact, none of the previously existing approaches to PWA controller design have developed a nonsmooth theory nor have they considered using well-developed nonsmooth analysis theory in the literature e.g. [31]. By contrast, in this chapter, we depart from previous approaches to PWA controller design by using Theorem 2.1, a Lyapunov theorem for nonsmooth Lyapunov functions. The theorem has the advantage of including the standard Lyapunov stability theorem in [70] as its special case for  $C^1$  Lyapunov functions. The proposed PWA controller in this chapter has the additional advantage of coinciding locally with a robust linear controller designed using linear control methods. It therefore combines local performance with global stability. One important application of the proposed method can thus be to extend the region of convergence of existing linear controllers for nonlinear systems.

The rest of this chapter is organized as follows. An illustrative example is employed in section 3.2 to clarify the need for the proposed method. Section 3.3 then explains the proposed method which consists of a robust linear controller design and its PWA extension. Finally, PWA controllers are designed for numerical examples in section 3.4 and conclusions are drawn in section 3.5.

#### **3.2** Illustrative Example

In this section, the following nonlinear system is used to illustrate the design procedure:

$$\dot{x} = 0.5(1 - x^2) + u \tag{3.1}$$

The open loop system has two equilibrium points (figure 3.2), one at x = -1 (unstable) and the other one at x = 1 (stable). The goal is to design a controller so that for any  $x(0) \in \mathcal{X} = [-4, 4]$ , the trajectory of the system asymptotically converges

to  $x^* = 1$ . It is also required that for any  $x(0) \in (0, 2)$ , the following cost function

$$J = \int_0^\infty [Q(x-1)^2 + Ru^2]dt$$
 (3.2)

be minimized where Q = 2 and R = 1.

To achieve this goal, continuous PWA functions  $\sigma_1(x)$  and  $\sigma_2(x)$  (Figure 3.3) are first defined so that

$$\dot{x} \in \mathbf{conv}\{\sigma_1(x) + u, \sigma_2(x) + u\}$$
(3.3)

where **conv** stands for the closed convex hull [107] of a set and  $\sigma_1(x)$  and  $\sigma_2(x)$  are affine in x inside each of the following regions:

$$\mathcal{R}_1 = (-4, -2), \mathcal{R}_2 = (-2, 0), \mathcal{R}_3 = (0, 2), \mathcal{R}_4 = (2, 4)$$
 (3.4)

In  $\mathcal{R}_3$  (where  $x^*$  is located), the dynamics of the system are described by the following LDI,

$$\dot{x} \in \mathbf{conv}\{-1.6(x-1)+u, -0.4(x-1)+u\}$$
 (3.5)

Defining  $z = x - x^* = x - 1$ , we have

$$\dot{z} \in \mathbf{conv}\{-1.6z + u, -0.4z + u\}$$
 (3.6)

An LQR controller can be designed for (3.6) using the design method for robust linear controllers described in subsection 3.3.1. The resulting controller for  $\mathcal{R}_3$  is then described by

$$u = -1.07x + 1.07 \tag{3.7}$$

Figure 3.4 shows the phase plane of the nonlinear system in feedback connection with the linear controller. It can be clearly seen that the system still has two equilibrium points. Therefore, although the closed-loop system locally satisfies the required performance measure, it is not globally stable. In the following sections, a method for extending the designed LQR controller to a PWA controller will be presented. It will be shown in section 3.4 that the resulting PWA controller has the same local performance and, in addition, it is globally stablizing.



Figure 3.2: The phase plane of the open loop system.

# 3.3 Extension of a Linear Controller to a PWA Controller

This section proposes a method to extend a local linear controller to a global PWA controller. The method consists of two steps. In the first step, a robust linear controller will be designed for a nonlinear system that is affine in the input. In this step, the designer can benefit from well established methods for designing robust linear controllers to make the nonlinear system locally stable and to satisfy a performance requirement in a neighbourhood of the desired equilibrium point. In the second step, the objective is to design a PWA controller that coincides with the linear controller in the neighbourhood of the equilibrium point and guarantees global stability of the nonlinear closed-loop system.

Consider the following nonlinear system

$$\dot{x} = f(x) + g(x)u \tag{3.8}$$



Figure 3.3: PWA differential inclusion

where  $x \in \mathcal{X} \subset \mathbb{R}^n$  and  $u \in \mathbb{R}^m$ . Let

$$\dot{x} \in \mathbf{conv}\{\sigma_1(x, u), \dots, \sigma_{\mathcal{K}}(x, u)\}\tag{3.9}$$

where  $\sigma_{\kappa}(x, u)$  is defined as

$$\sigma_{\kappa}(x,u) = A_{i\kappa}x + a_{i\kappa} + B_{i\kappa}u, x \in \overline{\mathcal{R}}_i, \tag{3.10}$$

with  $A_{i\kappa} \in \mathbb{R}^{n \times n}$ ,  $a_{i\kappa} \in \mathbb{R}^n$ ,  $B_{i\kappa} \in \mathbb{R}^{n \times m}$  for  $i = 1, \ldots, M$  and  $\kappa = 1, \ldots, \mathcal{K}$ . The polytopic regions  $\mathcal{R}_i$  are constructed as the intersection of a finite number of half spaces

$$\mathcal{R}_{i} = \{x | E_{i}x + e_{i} \succ 0\}, \text{ for } i = 1, \dots, M$$
(3.11)

where  $E_i \in \mathbb{R}^{p_i \times n}$ ,  $e_i \in \mathbb{R}^{p_i}$  and  $\succ$  represents an elementwise inequality.

The objective is to stabilize system (3.8) to  $x = x^*$  while satisfying a performance requirement for x close to  $x^*$ . The two steps of the proposed method will be presented in the following subsections.



Figure 3.4: The phase plane of the system with the linear controller.

#### 3.3.1 Step 1: Robust linear controller design

The first step is to design a robust linear controller for the LDI describing local behaviour of the nonlinear system. Consider a region  $\mathcal{R}_{i^*}$  such that

$$x^{\star} \in \overline{\mathcal{R}}_{i^{\star}}$$

The dynamics of system (3.8) in this region can be described by the following LDI.

$$\dot{x} \in \mathbf{conv}\{A_{i^{\star}\kappa}x + a_{i^{\star}\kappa} + B_{i^{\star}\kappa}u | \kappa = 1, \dots, \mathcal{K}\}$$
(3.12)

Changing variables to  $z = x - x^*$  and assuming a state feedback control input  $u = K_{i^*}x + k_{i^*}$  yields

$$\dot{z} \in \mathbf{conv}\{(A_{i^{\star}\kappa} + B_{i^{\star}\kappa}K_{i^{\star}})(z + x^{\star}) + a_{i^{\star}\kappa} + B_{i^{\star}\kappa}k_{i^{\star}} | \kappa = 1, \dots, \mathcal{K}\}$$
(3.13)

To make z = 0 an equilibrium point of the system, the following condition must be satisfied.

$$(A_{i^{\star}\kappa} + B_{i^{\star}\kappa}K_{i^{\star}\kappa})x^{\star} + a_{i^{\star}\kappa} + B_{i\kappa}k_{i^{\star}\kappa} = 0, \ \kappa = 1, \dots, \mathcal{K}$$
(3.14)

The closed-loop dynamics of the system can then be written as

$$\dot{z} \in \operatorname{conv}\{(A_{i^{\star}\kappa} + B_{i^{\star}\kappa}K_{i^{\star}\kappa})z \mid \kappa = 1, \dots, \mathcal{K}\}$$
(3.15)

The matrix gain  $K_{i^*}$  can be designed using robust linear control methodologies to satisfy desired design objectives for the differential inclusion (3.15). The affine term of the controller  $k_{i^*}$  can then be computed if the linear equation (3.14) has a solution. The choice of the required performance measure depends on the application. In this work, a robust LQR is designed for the LDI (3.12) using the following result taken from [64].

**Theorem 3.1.** [64] Consider the cost function

$$J = \int_0^\infty (z^T Q z + u^T R u) dt \tag{3.16}$$

where  $Q \ge 0$  and R > 0 for the following LDI

$$\dot{z} \in \operatorname{conv}\{A_{\kappa}z + B_{\kappa}u | \kappa = 1, \dots, \mathcal{K}\}$$

$$(3.17)$$

where  $z \in \mathbb{R}^n$  and  $u \in \mathbb{R}^m$ . If there exist S and Y so that

$$S > 0$$

$$\begin{bmatrix} SA_{\kappa}^{T} + A_{\kappa}S + Y^{T}B_{\kappa}^{T} + B_{\kappa}Y & SQ^{1/2} & Y^{T}R^{1/2} \\ Q^{1/2}S & -I_{n} & 0 \\ R^{1/2}Y & 0 & -I_{m} \end{bmatrix} < 0$$
(3.18)
$$(3.18)$$

for  $\kappa = 1, \ldots, \mathcal{K}$ , then for u = Kz where  $K = YS^{-1}$ , we have

$$J < z(0)^T S^{-1} z(0) \tag{3.20}$$

To avoid the dependency of the upper bound of the cost function on initial conditions of the system, it is proposed in [64] to assume that the initial condition
is a random vector with zero mean and identity covariance, i.e.,

$$\mathbb{E}\{z(0)\} = 0$$
  
$$\mathbb{E}\{z(0)z(0)^T\} = I$$
 (3.21)

It is shown in [64] that **Trace**  $(S^{-1})$  is an upper bound on  $\mathbb{E}\{J\}$ . Therefore it is proposed in the same reference to solve the following optimization problem to minimize the upper bound on the cost function.

 $\max \mathbf{Trace} \ (S)$ 

subject to (3.18) and (3.19) (3.22)

This optimization problem can be solved using SeDuMi [121] and Yalmip [76] to compute the controller gain  $K_{i^*}$ . The affine term  $k_{i^*}$  can then be computed by solving (3.14). The controller in region  $\mathcal{R}_{i^*}$  can then be written as

$$u = \bar{K}_{i^{\star}}\bar{x}, \text{ where } \bar{K}_{i^{\star}} = \begin{bmatrix} K_{i^{\star}} & k_{i^{\star}} \end{bmatrix} \text{ and } \bar{x} = \begin{bmatrix} x \\ 1 \end{bmatrix}$$
 (3.23)

The next step is to find a PWA controller that coincides with the linear controller (3.23) in  $\mathcal{R}_{i^*}$  and guarantees the stability of the closed-loop system in  $\mathcal{X}$ .

#### 3.3.2 Step 2: PWA state feedback design

The second step is to extend the robust linear controller to a PWA state feedback controller that stabilizes the nonlinear system (3.8) at the equilibrium point  $x^*$ . A PWA control input of the following form is considered for this purpose

$$u = K_i x + k_i = \bar{K}_i \bar{x}, \text{ for } x \in \mathcal{R}_i$$
(3.24)

where

$$\bar{K}_i = \left[ \begin{array}{cc} K_i & k_i \end{array} \right] \tag{3.25}$$

The closed loop system is therefore described by

$$\dot{x} = f(x) + g(x)(\bar{K}_i \bar{x}) \text{ for } x \in \mathcal{R}_i, \qquad (3.26)$$

Consider the piecewise quadratic candidate Lyapunov function continuous at the boundaries and defined in  $\mathcal{X}$  by the expression

$$V(x) = \bar{x}^T \bar{P}_i \bar{x}, \text{ for } x \in \overline{\mathcal{R}}_i$$
(3.27)

where  $\bar{P}_i = \bar{P}_i^T \in \mathbb{R}^{(n+1) \times (n+1)}$  and

$$\bar{P}_i = \begin{bmatrix} P_i & q_i \\ q_i^T & r_i \end{bmatrix}$$
(3.28)

with  $P_i \in \mathbb{R}^{n \times n}$ ,  $q_i \in \mathbb{R}^n$  and  $r_i \in \mathbb{R}$ . To simplify the notation, define

$$\bar{A}_{i\kappa} = \begin{bmatrix} A_{i\kappa} & a_{i\kappa} \\ 0 & 1 \end{bmatrix}, \bar{B}_{i\kappa} = \begin{bmatrix} B_{i\kappa} \\ 0 \end{bmatrix}, \bar{F}_{ij} = \begin{bmatrix} F_{ij} & f_{ij} \\ 0 & 1 \end{bmatrix}, \bar{x}^{\star} = \begin{bmatrix} x^{\star} \\ 1 \end{bmatrix} (3.29)$$
$$\bar{I} = \begin{bmatrix} I & 0 \\ -x^{\star^{\mathrm{T}}} & 1 \end{bmatrix} \begin{bmatrix} I & -x^{\star} \\ 0 & 1 \end{bmatrix} = \begin{bmatrix} I & -x^{\star} \\ -x^{\star^{\mathrm{T}}} & x^{\star^{\mathrm{T}}}x^{\star} \end{bmatrix}$$
(3.30)

The following theorem describes sufficient conditions for the existence of a continuous piecewise quadratic Lyapunov function of the form (3.27) and a PWA controller of the form (3.24) that coincides with the robust linear controller in the region where  $x^*$  lies and guarantees global stability.

**Theorem 3.2.** Let there exist matrices  $\bar{P}_i = \bar{P}_i^T$  defined in (3.28),  $\bar{K}_i$  defined in (3.25),  $Z_i$ ,  $\bar{Z}_i$ ,  $\Lambda_{i\kappa}$  and  $\bar{\Lambda}_{i\kappa}$  that verify the following conditions for all i = 1, ..., M,  $\kappa = 1, ..., \mathcal{K}$  and for a given decay rate  $\alpha > 0$ , desired equilibrium point  $x^*$ , linear controller gain  $\bar{K}_{i^*}$  defined in (3.23) and  $\epsilon > 0$ 

• Conditions on the PWA controller:

$$\bar{K}_i = \bar{K}_{i^\star}, \ if \ x^\star \in \overline{\mathcal{R}}_i \tag{3.31}$$

$$(\bar{A}_{i\kappa} + \bar{B}_{i\kappa}\bar{K}_i)\bar{x}^* = 0, \text{ if } x^* \in \overline{\mathcal{R}}_i$$
(3.32)

$$(\bar{A}_{i\kappa} + \bar{B}_{i\kappa}\bar{K}_i)\bar{F}_{ij} = (\bar{A}_{j\kappa} + \bar{B}_{j\kappa}\bar{K}_j)\bar{F}_{ij}, \quad if \quad \overline{\mathcal{R}}_i \bigcap \overline{\mathcal{R}}_j \neq \emptyset$$
(3.33)

• Continuity of the Lyapunov function:

$$\bar{F}_{ij}^{T}(\bar{P}_{i}-\bar{P}_{j})\bar{F}_{ij}=0, \ if \ \overline{\mathcal{R}}_{i}\bigcap\overline{\mathcal{R}}_{j}\neq\emptyset$$
(3.34)

• Positive definiteness of the Lyapunov function:

$$\bar{P}_i \bar{x}^\star = 0, \ if \ x^\star \in \overline{\mathcal{R}}_i \tag{3.35}$$

$$P_i > \epsilon I, \text{ if } x^* \in \overline{\mathcal{R}}_i, E_i x^* + e_i \neq 0$$
(3.36)

$$\begin{aligned}
Z_i \in \mathbb{R}^{n \times n}, & Z_i \succeq 0 \\
P_i - E_i^T Z_i E_i > \epsilon I
\end{aligned}, if x^* \in \overline{\mathcal{R}}_i, E_i x^* + e_i = 0 \tag{3.37}$$

$$\begin{cases} \bar{Z}_i \in \mathbb{R}^{(n+1)\times(n+1)}, \bar{Z}_i \succeq 0\\ \bar{P}_i - \bar{E}_i^T \bar{Z}_i \bar{E}_i > \epsilon \bar{I} \end{cases}, & \text{if } x^* \notin \overline{\mathcal{R}}_i \end{cases}$$
(3.38)

• Monotonicity of the Lyapunov function:

for *i* such that 
$$x^* \in \mathcal{R}_i, E_i x^* + e_i \neq 0$$
,  
 $P_i(A_{i\kappa} + B_{i\kappa}K_i) + (A_{i\kappa} + B_{i\kappa}K_i)^T P_i < -\alpha P_i$ 
(3.39)

for *i* such that 
$$x^* \in \overline{\mathcal{R}}_i, E_i x^* + e_i = 0,$$
  

$$\begin{cases} \Lambda_{i\kappa} \in \mathbb{R}^{n \times n}, \ \Lambda_{i\kappa} \succeq 0 \\ P_i(A_{i\kappa} + B_{i\kappa}K_i) + (A_{i\kappa} + B_{i\kappa}K_i)^T P_i + E_i^T \Lambda_{i\kappa}E_i < -\alpha P_i \end{cases}$$
(3.40)

for *i* such that 
$$x^* \notin \overline{\mathcal{R}}_i$$
,  

$$\begin{cases}
\bar{\Lambda}_{i\kappa} \in \mathbb{R}^{(n+1)\times(n+1)}, \ \bar{\Lambda}_{i\kappa} \succeq 0 \\
\bar{P}_i(\bar{A}_{i\kappa} + \bar{B}_{i\kappa}\bar{K}_i) + (\bar{A}_{i\kappa} + \bar{B}_{i\kappa}\bar{K}_i)^T \bar{P}_i + \bar{E}_i^T \bar{\Lambda}_{i\kappa}\bar{E}_i < -\alpha \bar{P}_i
\end{cases}$$
(3.41)

Then for the nonlinear system (3.26), all trajectories in  $\mathcal{X}$  asymptotically converge to  $x = x^*$ .

*Proof.* Consider the change of coordinates  $z = x - x^*$  or equivalently

$$\begin{cases} \bar{z} = \bar{T}_{zx}\bar{x} \\ \bar{x} = \bar{T}_{zx}^{-1}\bar{z} \end{cases}$$
(3.42)

where

$$\bar{T}_{zx} = \begin{bmatrix} I & -x^* \\ 0 & 1 \end{bmatrix}, \quad \bar{T}_{zx}^{-1} = \begin{bmatrix} I & x^* \\ 0 & 1 \end{bmatrix}$$
(3.43)

With this change of variables, the differential inclusions (3.9) with the control input  $u = K_i x + k_i$  for  $x \in \mathcal{R}_i$  is transformed into

$$\dot{z} \in \mathbf{conv}\{\sigma_1^z(z), \dots, \sigma_{\mathcal{K}}^z(z)\}$$
(3.44)

where  $\sigma_{\kappa}^{z}(z)$  is defined as

$$\sigma_{\kappa}^{z}(z) = A_{i\kappa}^{z} z +, a_{i\kappa}^{z}, \ z \in \overline{\mathcal{R}}_{i}^{z}, \tag{3.45}$$

for  $i = 1, \ldots, M$  and  $\kappa = 1, \ldots, \mathcal{K}$  where

1

$$\bar{A}_{i\kappa}^{z} = \begin{bmatrix} A_{i\kappa}^{z} & a_{i\kappa}^{z} \\ 0 & 0 \end{bmatrix} = (\bar{A}_{i\kappa} + \bar{B}_{i\kappa}\bar{K}_{i})\bar{T}_{zx}^{-1}$$
(3.46)

and therefore

$$A_{i\kappa}^{z} = A_{i\kappa} + B_{i\kappa}K_{i} \tag{3.47}$$

$$a_{i\kappa}^{z} = (A_{i\kappa} + B_{i\kappa}K_{i})x^{\star} + a_{i\kappa} + B_{i\kappa}k_{i}$$
(3.48)

Polytopic regions  $\mathcal{R}^z_i$  can be written as

$$\mathcal{R}_{i}^{z} = \{ z | E_{i}^{z} z + e_{i}^{z} \succ 0 \}, \text{ for } i = 1, \dots, M$$
(3.49)

where  $E_i^z = E_i$  and  $e_i^z = E_i x^* + e_i$ . If  $\overline{\mathcal{R}}_i^z \cap \overline{\mathcal{R}}_j^z \neq \emptyset$  (i.e.  $\overline{\mathcal{R}}_i \cap \overline{\mathcal{R}}_j \neq \emptyset$ ) it follows from (2.64) that

$$\forall x \in \overline{\mathcal{R}}_i \bigcap \overline{\mathcal{R}}_j, \ \exists s \in \mathbb{R}^{n-1}, \ x = F_{ij}s + f_{ij}$$
(3.50)

Thus

$$\forall z \in \overline{\mathcal{R}}_i^z \bigcap \overline{\mathcal{R}}_j^z, \ \exists s \in \mathbb{R}^{n-1}, \ z = F_{ij}s + f_{ij} - x^*$$
(3.51)

and one can define  $F_{ij}^z = F_{ij}, f_{ij}^z = f_{ij} - x^*$  and

$$\bar{F}_{ij}^{z} = \begin{bmatrix} F_{ij}^{z} & f_{ij}^{z} \\ 0 & 1 \end{bmatrix} = \bar{T}_{zx}\bar{F}_{ij}$$
(3.52)

For the candidate Lyapunov function  $V^z(z) = V(x)$ , for  $z \in \mathcal{R}^z_i$  (i.e.  $x \in \mathcal{R}_i$ ) we have

$$V^{z}(z) = V_{i}^{z}(z), (3.53)$$

where

$$V_{i}^{z}(z) = \bar{x}^{\mathrm{T}}\bar{P}_{i}\bar{x} = \bar{z}^{\mathrm{T}}\bar{T}_{zx}^{-\mathrm{T}}\bar{P}_{i}\bar{T}_{zx}^{-1}\bar{z} = \bar{z}^{\mathrm{T}}\bar{P}_{i}^{z}\bar{z}$$
(3.54)

and thus

$$\bar{P}_{i}^{z} = \begin{bmatrix} P_{i}^{z} & q_{i}^{z} \\ q_{i}^{z^{\mathrm{T}}} & r_{i}^{z} \end{bmatrix} = \begin{bmatrix} P_{i} & P_{i}x^{\star} + q_{i} \\ \hline x^{\star^{\mathrm{T}}}P_{i} + q_{i}^{\mathrm{T}} & x^{\star^{\mathrm{T}}}P_{i}x^{\star} + q_{i}^{\mathrm{T}}x^{\star} + x^{\star^{\mathrm{T}}}q_{i} + r_{i} \end{bmatrix}$$
(3.55)

In the following, it is shown that for any  $\kappa \in \{1, \ldots, \mathcal{K}\}$ , the PWA system  $\dot{z} = \sigma_{\kappa}^{z}(z)$  satisfies the conditions of Proposition 2.5.

- Conditions on the vector field:
  - It follows from (3.32) that if  $x^* \in \mathcal{R}_i$  i.e.  $0 \in \mathcal{R}_i^z$

$$\left( \begin{bmatrix} A_{i\kappa} & a_{i\kappa} \\ 0 & 0 \end{bmatrix} + \begin{bmatrix} B_{i\kappa} \\ 0 \end{bmatrix} \begin{bmatrix} K_i & k_i \end{bmatrix} \right) \begin{bmatrix} x^* \\ 1 \end{bmatrix} = 0 \quad (3.56)$$

This is equivalent to  $a_{i\kappa}^{z} = 0$  and therefore (2.91) is satisfied.

- Using (3.33), for *i* and *j* such that  $\overline{\mathcal{R}}_i \bigcap \overline{\mathcal{R}}_j \neq \emptyset$  we have

$$\left[\left(\bar{A}_{i\kappa} + \bar{B}_{i\kappa}\bar{K}_{i}\right) - \left(\bar{A}_{j\kappa} + \bar{B}_{j\kappa}\bar{K}_{j}\right)\right]\bar{T}_{zx}^{-1}\bar{T}_{zx}\bar{F}_{ij} = 0 \qquad (3.57)$$

and therefore from (3.46) and (3.52)

$$(\bar{A}_{i\kappa}^z - \bar{A}_{j\kappa}^z)\bar{F}_{ij}^z = 0, \text{ if } \overline{\mathcal{R}}_i^z \bigcap \overline{\mathcal{R}}_j^z \neq \emptyset$$
 (3.58)

Thus (2.92) is satisfied.

• Continuity of the Lyapunov function: It follows from (3.34) that

$$\bar{F}_{ij}^{\mathrm{T}}\bar{T}_{zx}^{\mathrm{T}}\bar{T}_{zx}^{-\mathrm{T}}(\bar{P}_i - \bar{P}_j)\bar{T}_{zx}^{-1}\bar{T}_{zx}\bar{F}_{ij} = 0, \text{ if } \overline{\mathcal{R}}_i \bigcap \overline{\mathcal{R}}_j \neq \emptyset$$
(3.59)

and from (3.52)

$$\bar{F}_{ij}^{z^{\mathrm{T}}}(\bar{P}_{i}^{z}-\bar{P}_{j}^{z})\bar{F}_{ij}^{z}=0, \text{ if } \overline{\mathcal{R}}_{i}^{z}\bigcap\overline{\mathcal{R}}_{j}^{z}\neq\emptyset$$
(3.60)

Therefore (2.93) is satisfied.

- Positive definiteness of the Lyapunov function:
  - The condition (3.35) can be rewritten as

$$\begin{cases} P_i x^* + q_i = 0\\ q_i^{\mathrm{T}} x^* + r_i = 0 \end{cases}, \quad \text{if } x^* \in \mathcal{R}_i \tag{3.61}$$

and therefore from (3.55),

$$\begin{cases} q_i^z = 0 \\ r_i^z = 0 \end{cases}, & \text{if } 0 \in \mathcal{R}_i^z \\ r_i^z = 0 \end{cases}$$
(3.62)

Thus (2.94) is satisfied.

- It follows from (3.36), (3.55), (3.62) and (3.49) that

$$P_i^z > \epsilon I$$
, if  $0 \in \overline{\mathcal{R}}_i^z$  and  $e_i^z \neq 0$  (3.63)

Therefore (2.95) is satisfied.

- It follows from (3.37), (3.55), (3.62) and (3.49) that

$$P_i^z - E_i^{\mathrm{T}} Z_i E_i > \epsilon I$$
, if  $0 \in \overline{\mathcal{R}}_i^z$  and  $e_i^z = 0$  (3.64)

Therefore (2.96) is satisfied.

- The condition (3.37) can be rewritten as

$$\bar{T}_{zx}^{-\mathrm{T}}\bar{P}_{i}\bar{T}_{zx}^{-1} - \bar{T}_{zx}^{-\mathrm{T}}\bar{E}_{i}^{T}\bar{Z}_{i}\bar{E}_{i}\bar{T}_{zx}^{-1} > \epsilon\bar{T}_{zx}^{-\mathrm{T}}\bar{I}\bar{T}_{zx}^{-1}$$
(3.65)

and then from (3.54), (3.49) and (3.30) that

$$\bar{P}_i^z - \bar{E}_i^{z^{\mathrm{T}}} Z_i \bar{E}_i^z > \epsilon \begin{bmatrix} I & 0 \\ 0 & 0 \end{bmatrix}, \text{ if } 0 \notin \overline{\mathcal{R}}_i^z \qquad (3.66)$$

Therefore (2.97) is satisfied.

- Monotonicity of the Lyapunov function:
  - The condition (3.39) can be rewritten using (3.49), (3.55) and (3.56) as

$$P_i^z A_{i\kappa}^z + A_{i\kappa}^{z^{\mathrm{T}}} P_i^z < -\alpha P_i^z, \quad \text{if } 0 \in \mathcal{R}_i^z, \ e_i^z \neq 0$$
(3.67)

This is equivalent to (2.98).

- It follows from (3.40), (3.49) and (3.55) that

$$P_i^z A_{i\kappa} + A_{i\kappa}^{\mathrm{T}} P_i^z + E_i^{z^{\mathrm{T}}} \Lambda_i E_i^z < -\alpha P_i^z, \text{ if } 0 \in \overline{\mathcal{R}}_i^z \text{ and } e_i^z = 0 \qquad (3.68)$$

Therefore (2.99) is satisfied.

– Multiplying (3.41), by  $\bar{T}_{zx}^{-\mathrm{T}}$  and  $\bar{T}_{zx}^{-1}$  from left and right yields

$$\bar{T}_{zx}^{-\mathrm{T}}\bar{P}_{i}(\bar{A}_{i\kappa}+\bar{B}_{i\kappa}\bar{K}_{i})\bar{T}_{zx}^{-1}+\bar{T}_{zx}^{-\mathrm{T}}(\bar{A}_{i\kappa}+\bar{B}_{i\kappa}\bar{K}_{i})^{T}\bar{P}_{i}\bar{T}_{zx}^{-1} +\bar{T}_{zx}^{-\mathrm{T}}\bar{E}_{i}^{T}\bar{\Lambda}_{i\kappa}\bar{E}_{i}\bar{T}_{zx}^{-1}<-\alpha\bar{T}_{zx}^{-\mathrm{T}}\bar{P}_{i}\bar{T}_{zx}^{-1}$$
(3.69)

Since  $T_{zx}^{-1}T_{zx} = I$ , the condition (3.69) can be rewritten as

$$\bar{T}_{zx}^{-\mathrm{T}}\bar{P}_{i}T_{zx}^{-1}T_{zx}(\bar{A}_{i\kappa}+\bar{B}_{i\kappa}\bar{K}_{i})\bar{T}_{zx}^{-1}+\bar{T}_{zx}^{-\mathrm{T}}(\bar{A}_{i\kappa}+\bar{B}_{i\kappa}\bar{K}_{i})^{T}T_{zx}^{\mathrm{T}}T_{zx}^{-\mathrm{T}}\bar{P}_{i}\bar{T}_{zx}^{-1} +\bar{T}_{zx}^{-\mathrm{T}}\bar{E}_{i}^{T}\bar{\Lambda}_{i\kappa}\bar{E}_{i}\bar{T}_{zx}^{-1} < -\alpha\bar{T}_{zx}^{-\mathrm{T}}\bar{P}_{i}\bar{T}_{zx}^{-1}$$
(3.70)

Note that from (3.43), it follows

$$\bar{T}_{zx}(\bar{A}_{i\kappa} + \bar{B}_{i\kappa}\bar{K}_i) = \begin{bmatrix} I & -x^* \\ 0 & 1 \end{bmatrix} \begin{bmatrix} A_{i\kappa} + B_{i\kappa}K_i & a_{i\kappa} + B_{i\kappa}k_i \\ 0 & 0 \end{bmatrix}$$
$$= \begin{bmatrix} A_{i\kappa} + B_{i\kappa}K_i & a_{i\kappa} + B_{i\kappa}k_i \\ 0 & 0 \end{bmatrix}$$
$$= (\bar{A}_{i\kappa} + \bar{B}_{i\kappa}\bar{K}_i) \qquad (3.71)$$

Now using (3.71), (3.70), (3.46), (3.49) and (3.54), we have

$$\bar{P}_i^z \bar{A}_i^z + \bar{A}_i^{z^{\mathrm{T}}} \bar{P}_i^z + \bar{E}_i^{z^{\mathrm{T}}} \bar{\Lambda}_i \bar{E}_i^z < -\alpha \bar{P}_i^z, \text{ if } 0 \notin \overline{\mathcal{R}}_i^z \qquad (3.72)$$

Therefore (2.100) is satisfied.

Since all the conditions of Proposition 2.5 are satisfied, it can be concluded that for any  $\kappa \in \{1, \ldots, \mathcal{K}\}$ , the PWA system  $\dot{z} = \sigma_{\kappa}^{z}(z)$  is stable at the origin (z = 0) and

$$V_i^z(x) > \epsilon ||z||^2, z \in \overline{\mathcal{R}}_i^z$$
(3.73)

$$\nabla V_i^z(x)^{\mathrm{T}}(A_{i\kappa}^z z + a_{i\kappa}^z) < -\alpha V_i^z(x), z \in \overline{\mathcal{R}}_i^z$$
(3.74)

for i = 1, ..., M.

The differential inclusion (3.44) can be described as

$$\dot{z} = f_i^z(z), \ z \in \mathcal{R}_i^z \tag{3.75}$$

where

$$f_i(z)^z = \sum_{\kappa=1}^{\mathcal{K}} \omega_\kappa(t) (A_{i\kappa}^z z + a_{i\kappa}^z)$$
(3.76)

and  $\omega_{\kappa}(t) \geq 0$  for  $\kappa = 1, ..., \mathcal{K}$  and all  $t \geq 0$  are piecewise smooth functions such that

$$\sum_{\kappa=1}^{\mathcal{K}} \omega_{\kappa}(t) = 1 \tag{3.77}$$

From (3.76) and (3.74), it follows

$$\nabla V_i^z(x)^{\mathrm{T}} f_i(z) < -\alpha V_i^z(x), z \in \overline{\mathcal{R}}_i^z$$
(3.78)

Now using Proposition 2.3, the system (3.75) is dissipative with the storage function  $V^{z}(z)$  and the supply rate  $-\alpha V^{z}(x)$ . Therefore  $S(z,t) = V^{z}(z) + \int_{0}^{t} \alpha V^{z}(z(\tau))d\tau$  is nonincreasing along the trajectories of (3.75). This implies that the conditions of Theorem 2.1 are satisfied and all trajectories of (3.75) (the nonlinear closed loop system (3.26)) asymptotically converge to z = 0 ( $x = x^{*}$ ).

**Remark 3.1.** Theorem 3.2 shows that there is no need to assume that there is one equilibrium point for the dynamic equations of each region and to select them a-priori by solving an optimization problem (such as it was done in [103]).

**Remark 3.2.** The conditions in Theorem 3.2 include bilinear matrix inequalities (BMI) which make the problem nonconvex. Reference [124] showed that the problem of checking the solvability of a BMI is NP-hard. The complexity of the synthesis problem increases with the order of the system, the dimension of the partitioned space and the number of regions. However, PENBMI [71], a recent software package providing algorithms with local optimality guarantees, can be used in practice to search for a local solution to the problem as it will be shown in the next section.

#### **3.4** Numerical Examples

**Example 3.1.** For the illustrative example in section 3.2, a PWA controller is designed to extend the region of convergence of the robust LQR controller. A feasible solution to the synthesis problem described in Theorem 3.2, was calculated using PENBMI [71] and Yalmip [76]. Figure 3.5 depicts the resulting piecewise quadratic Lyapunov function. The designed PWA controller (Figure 3.6) is described by the following gains.

$$\bar{K}_1 = [-4.08 - 0.437], \ \bar{K}_2 = [-3.32 \ 1.07]$$
  
 $\bar{K}_3 = [-1.07 \ 1.07], \ \bar{K}_4 = [2.45 - 5.97]$ 
(3.79)

Note that the PWA controller coincides with the linear LQR controller in (0, 2). Figure 3.7 shows the phase plane of the closed-loop system consisting of the nonlinear system in feedback connection with the PWA controller. Notice that the closed-loop system has now only one equilibrium point in  $\mathcal{X} = [-4, 4]$  and it is stable for all initial conditions in  $\mathcal{X}$ .

**Example 3.2.** Consider the following simple PWA system (adopted from [96] with slight modification)



Figure 3.5: The computed Lyapunov function - Example 3.1

$$\dot{x}_1 = x_2$$
  
 $\dot{x}_2 = -0.1x_2 + g(x_1) + u$  (3.80)

where  $g(x_1)$  is the PWA function depicted in Figure 3.8. It is desired to stabilize the origin  $(x_1 = x_2 = 0)$  for this system. The local performance criterion is

$$J(x,u) = \int_0^\infty 4x_1^2(t) + 4x_2(t)^2 + u(t)^2 dt$$
(3.81)

At first, a PWA controller was designed by applying the synthesis method proposed by [96] using PWLTOOL [57]. Figure 3.9 shows the trajectories of the closed loop system. It can be seen that, in this case, the PWA controller designed by PWLTOOL does not stabilize the origin even locally.

We then employ Theorem 3.2 to stabilize the origin and to extend the following LQR controller with the cost function (3.81) to a PWA controller

$$u = -3.2361x_1 - 3.1376x_2 \tag{3.82}$$



Figure 3.6: The designed PWA controller - Example 3.1

Figure 3.10 depicts the trajectories of the closed loop system. The PWA controller stabilizes the origin while it coincides with the LQR controller (3.82) for the center region  $(-1 < x_1 < 1)$ .

**Example 3.3.** Consider the following second order system

$$\dot{x}_1 = x_2$$
  
 $\dot{x}_2 = -x_1 + 0.5x_2 - 0.5x_1^2x_2 + u$  (3.83)

Figure 3.11 shows the trajectories of the open loop system. A linear controller  $u = -198x_1 - 101x_2$  can extend the region of convergence to the origin as depicted in Figure 3.12. However, there still exist initial conditions in

$$\mathcal{X} = \left\{ \left[ \begin{array}{c} x_1 \\ x_2 \end{array} \right] \left| -30 < x_1 < 30, \ -60 < x_2 < 60 \right\}$$
(3.84)

for which the trajectories of the system do not converge to the origin.

To design a PWA controller, the nonlinear system (3.83) should first be included by a PWADI. This is done by computing upper and lower PWA bounds on



Figure 3.7: The phase plane of the closed-loop system with the PWA controller (solid) and the linear controller (dashed) - Example 3.1

the nonlinear function  $h(x) = 0.5x_1^2x_2$  and then substituting this nonlinear function in (3.83) by its PWA bounds. Figure 3.13 shows the regions (triangles) for which the PWA bounds were computed.

A PWA controller was then designed that satisfies all the conditions of Theorem 3.2. The corresponding piecewise quadratic Lyapunov function is depicted in Figure 3.14. The trajectories in Figure 3.13 clearly show that the PWA controller enlarges the region of convergence.

## **3.5** Conclusions

This chapter proposed a two-step synthesis method to achieve both local performance and global stability for a class of uncertain nonlinear systems. In this method, a local robust linear controller is first designed for a neighborhood of the desired equilibrium point to satisfy a local performance requirement. The local linear controller is then extended to a PWA controller to globally stabilize the nonlinear system.



Figure 3.8: PWA function - Example 3.2



Figure 3.9: Trajectories of the closed-loop system for the PWA controller proposed in [96] - Example 3.2



Figure 3.10: Trajectories of the closed-loop system for the proposed PWA controller - Example 3.2



Figure 3.11: Trajectories of the open loop system - Example 3.3



Figure 3.12: Trajectories of the closed-loop system for the linear controller - Example 3.3



Figure 3.13: Trajectories of the closed-loop system for the PWA controller - Example 3.3



Figure 3.14: Piecewise quadratic Lyapunov function - Example 3.3

The PWA controller locally coincides with the linear controller and therefore has the same local performance.

# Chapter 4

Controller synthesis for piecewise affine slab differential inclusions: a duality-based convex optimization approach

The main contribution of this chapter is to introduce for the first time a duality-based interpretation of piecewise affine (PWA) systems. This is a key concept to enable a convex formulation of PWA controller synthesis for PWA slab differential inclusions using a new convex relaxation. A convex optimization program is also proposed to compute a PWA differential inclusion that *includes* a nonlinear system for which the nonlinearity is a function of one variable. Therefore, by formulating the synthesis problem for PWA differential inclusions, the proposed method can also guarantee stability and performance for the original nonlinear system. Another important contribution of the chapter is to present stability and performance analysis and synthesis to PWA systems whose output is also a PWA function of the state (as opposed to a piecewise-linear

function). To this end, the definition of the regions of a PWA system is generalized in this chapter. These results work even when the PWA systems include sliding modes. Numerical examples illustrate the new approach.

## 4.1 Introduction

Controller synthesis for  $L_2$ -gain performance of PWA systems has attracted growing attention in recent years [40–42]. Reference [40] formulates the  $L_2$ -gain controller synthesis problem for uncertain PWA systems as a set of LMIs based on a piecewise quadratic (PWQ) Lyapunov function provided that the structure of the PWA controller is constrained. Reference [42] proposed a method to design PWL controllers for PWL systems based on a PWQ Lyapunov function to limit the  $L_2$ -gain of the system. The method was later extended to uncertain PWL systems in [41]. However, the approaches in [40, 42] and [41] do not use any S-procedure in the design process, which means that each closed-loop subsystem of the PWL system has to be stable and this makes the proposed methods conservative. Attractive sliding modes are also ignored in [40, 42] and [41]. There is therefore no guarantee for the closed-loop system to be stable in general. In addition, no method is proposed to obtain bounds on the uncertainty for nonlinear systems that are approximated by PWA systems.

A very important subclass of PWA systems is the class of PWA *slab systems* [101], for which the partition of the state space is a function of a scalar variable. The synthesis of PWL controllers for stability and performance of PWA slab systems is formulated in [56] as a set of LMIs. Reference [100] applied PWL  $L_2$ -gain controller synthesis to the problem of inventory control of production systems. Additional constraints were introduced in this chapter to limit the control input. However, for PWA controllers, it is said in [56] that "It doesn't seem that

the condition for stabilizability using this type of input command can be cast as an LMI". Reference [101] showed that by considering an affine term in the controller, the synthesis problem for PWA slab systems can be formulated as a set of LMIs parametrized by a vector. Three different algorithms for controller synthesis have been proposed in [101] and the bisection method has been used to find the controller that maximizes the decay rate of the trajectories.

However, no convex optimization problem has been proposed for PWA controller design for stability and performance without limiting the structure of the controller. To fill this gap in the literature, this chapter formulates PWA controller synthesis for stabilization as a set of LMIs. Then, this formulation is extended to  $L_2$ -gain controller design. These results are based on a new key concept - the *dual parameter set* - that is introduced in this chapter for the first time. In addition, PWA slab differential inclusions (as opposed to equations) are considered here. This enables the design for stability and performance of nonlinear systems that can be *included* by a PWA envelope. Furthermore, for nonlinear systems for which the nonlinearity is a function of one variable, a convex optimization method is proposed in this chapter to compute the PWA envelope that includes the nonlinear system.

The structure of the chapter is as follows. The definition of  $L_2$  gain of a nonlinear system and PWA differential inclusions are introduced in section 4.2 and section 4.3, respectively. A convex optimization method is then described in section 4.3.1 to compute a PWA envelope for a class of nonlinear systems. Stability and performance analysis are presented in section 4.4. Section 4.5 addresses stabilization and  $L_2$ -gain control design. Finally, numerical examples are shown in section 4.6 and conclusions are drawn in section 4.7.

# 4.2 Mathematical Preliminaries

Consider the nonlinear system

$$\begin{cases} \dot{x} = f(x) + g(x)w\\ y = h(x) \end{cases}$$
(4.1)

where f(x) and g(x) are defined almost everywhere and bounded for bounded ||x||.

The  $L_2$  gain from w to y is defined as

$$\sup_{0 < \|w\|_{2} < \infty} \frac{\|y\|_{2}}{\|w\|_{2}} \tag{4.2}$$

where the  $L_2$  norm of a signal z is defined as

$$\|z\|_{2} = \left[\int_{0}^{\infty} z^{\mathrm{T}}(\tau) z(\tau) d\tau\right]^{\frac{1}{2}}$$
(4.3)

and the supremum is taken over all nonzero trajectories assuming x(0) = 0.

**Lemma 4.1.** [7] The nonlinear system (4.1) has finite  $L_2$ -gain less than  $\gamma > 0$  if there exists a locally bounded storage function  $V : \mathbb{R}^n \to \mathbb{R}$ , such that  $V(x) \ge 0$  for all  $x \in \mathbb{R}^n$ , V(0) = 0 and

$$\forall t \ge 0, V(x(t)) \le V(x_0) + \int_0^t W(\tau) d\tau \tag{4.4}$$

for the supply rate  $W(\tau) = -\|y(\tau)\|_2^2 + \gamma^2 \|w(\tau)\|_2^2$ .

## 4.3 Polytopic PWA Slab Differential Inclusions

Polytopic PWA slab differential inclusions are a generalization of polytopic linear differential inclusions in [17]. A polytopic PWA slab differential inclusion is described by

$$\dot{x} = A(x,t)x + a(x,t) + B_u(x,t)u + B_w(x,t)w$$

$$y = C(x,t)x + c(x,t) + D_u(x,t)u + D_w(x,t)w$$
(4.5)

where  $x(t) \in \mathbb{R}^n$  denotes the state,  $u(t) \in \mathbb{R}^{n_u}$  is the control input,  $w(t) \in \mathbb{R}^{n_w}$  is the exogenous input and  $y(t) \in \mathbb{R}^{n_y}$  is the output. The initial state is  $x(0) = x_0$ . It is assumed that system (4.5) satisfies

$$\dot{x} \in \operatorname{conv}\{A_{i\kappa}x + a_{i\kappa} + B_{u_{i\kappa}}u + B_{w_{i\kappa}}w, \kappa = 1, 2\}$$
  
$$y \in \operatorname{conv}\{C_{i\kappa}x + c_{i\kappa} + D_{u_{i\kappa}}u + D_{w_{i\kappa}}w, \kappa = 1, 2\}$$
 for  $(x, w) \in \mathcal{R}_{i}^{\mathcal{X} \times \mathcal{W}}$  (4.6)

where **conv** stands for the convex hull of a set and  $\mathcal{R}_i^{\mathcal{X} \times \mathcal{W}}$ , i = 1, ..., M are M slab regions partitioning the cross product of a slab subset of the state space  $\mathcal{X} \subset \mathbb{R}^n$ and the space of the exogenous input  $\mathcal{W}$  defined as

$$\mathcal{R}_{i}^{\mathcal{X} \times \mathcal{W}} = \{(x, w) \mid \sigma_{i} < C_{\mathcal{R}} x + D_{\mathcal{R}} w < \sigma_{i+1}\},$$
(4.7)

where  $C_{\mathcal{R}} \in \mathbb{R}^{1 \times n}$ ,  $D_{\mathcal{R}} \in \mathbb{R}^{1 \times n_w}$  and  $\sigma_i$  for  $i = 1, \ldots, M + 1$  are scalars such that

$$\sigma_1 < \sigma_2 < \ldots < \sigma_{M+1} \tag{4.8}$$

Each slab region can be described by the following degenerate ellipsoid

$$\mathcal{R}_{i}^{\mathcal{X} \times \mathcal{W}} = \{(x, w) \mid ||L_{i}x + l_{i} + M_{i}w|| < 1\}$$
(4.9)

where  $L_i = 2C_{\mathcal{R}}/(\sigma_{i+1}-\sigma_i)$ ,  $l_i = -(\sigma_{i+1}+\sigma_i)/(\sigma_{i+1}-\sigma_i)$  and  $M_i = 2D_{\mathcal{R}}/(\sigma_{i+1}-\sigma_i)$ .

It is assumed that  $a_{i\kappa} = 0$  and  $c_{i\kappa} = 0$  for  $i \in \mathcal{I}(0,0)$  and  $\kappa = 1, 2$  where

$$\mathcal{I}(x,w) = \{i | (x,w) \in \overline{\mathcal{R}}_i^{\mathcal{X} \times \mathcal{W}}\}$$
(4.10)

The following subsection will formulate the computation of PWA envelopes for a class of nonlinear systems as convex optimization programs. These envelopes will be used to design PWA controllers for nonlinear systems.

#### 4.3.1 PWA envelope for nonlinear systems

In this subsection, a numerical method is proposed to compute a PWA envelope (bounding differential inclusion) for the following nonlinear system

$$\dot{x} = Ax + B_p p + B_u u + B_w w, \ x(0) = x_0,$$

$$q = C_{\mathcal{R}} x + D_{\mathcal{R}} w \qquad (4.11)$$

$$y = Cx + D_p p + D_u u + D_w w$$

where  $x \in \mathbb{R}^n$ ,  $u \in \mathbb{R}^{n_u}$ ,  $w \in \mathbb{R}^{n_w}$ ,  $y \in \mathbb{R}^{n_y}$  and the vector  $p \in \mathbb{R}^{n_p}$  is a nonlinear function of the scalar  $q \in \mathbb{R}$ 

$$p = f(q), \tag{4.12}$$

with f(0) = 0. It is assumed that f(q) is measured at a finite number of sampling points  $q_k$  for  $k = 1, ..., N_s$ .

The objective is to find PWA vector functions  $\delta_1(q)$  and  $\delta_2(q)$  defined as

$$\delta_{\kappa}(q) = A_{q_{i\kappa}}q + a_{q_{i\kappa}}, \text{ for } \sigma_i < q < \sigma_{i+1}, \ \kappa = 1, 2$$

$$(4.13)$$

for  $i = 1, \ldots, M$  such that

$$f(q) \in \mathbf{conv}\{\delta_1(q), \delta_2(q)\}$$
(4.14)

and  $\delta_{\kappa}(0) = 0$ ,  $\kappa = 1, 2$ . It is also required to make the bounding envelope as *tight* as possible. Given  $\sigma_i$  for i = 1, ..., M + 1, the computation of the bounding envelope consisting of  $\delta_1(q)$  and  $\delta_2(q)$  can be formulated as the following optimization problems.

1. Optimization problem to compute  $\delta_1(q)$ :

$$\begin{array}{ll}
\min_{A_{q_{i1}},a_{q_{i1}}} & \Sigma_{k=1,\dots,N_s} \|f(q_k) - \delta_1(q_k))\|^2 \\
\text{s.t.} & \delta_1(q) = A_{q_{i1}}q + a_{q_{i1}}, \text{ for } \sigma_i < q < \sigma_{i+1}, \\
& \delta_1(q_k) < f(q_k), \text{ for } q_k > 0 \\
& \delta_1(q_k) > f(q_k), \text{ for } q_k < 0
\end{array}$$
(4.15)

2. Optimization problem to compute  $\delta_2(q)$ :

$$\begin{aligned} \min_{\substack{A_{q_{i2}}, a_{q_{i2}}}} & \Sigma_{k=1,\dots,N_s} \| f(q_k) - \delta_2(q_k) ) \|^2 \\ \text{s.t.} & \delta_2(q) = A_{q_{i2}}q + a_{q_{i2}}, \text{ for } \sigma_i < q < \sigma_{i+1}, \\ & \delta_2(q_k) > f(q_k), \text{ for } q_k > 0 \\ & \delta_2(q_k) < f(q_k), \text{ for } q_k < 0 \end{aligned}$$
(4.16)

Continuity of  $\delta_1(q)$  and  $\delta_2(q)$  can be written as a set of additional linear constraints on  $A_{q_{i\kappa}}$  and  $a_{q_{i\kappa}}$  as

$$A_{q_{i\kappa}}\sigma_{i+1} + a_{q_{i\kappa}} = A_{q_{(i+1)\kappa}}\sigma_{i+1} + a_{q_{(i+1)\kappa}}, \tag{4.17}$$

for i = 1, ..., M - 1 and  $\kappa = 1, 2$ . The quadratic optimization problems (4.15) and (4.16) (including constraints (4.17) if needed) can be solved efficiently because they are convex. The nonlinear system (4.11) can then be embedded in the PWA differential inclusion (4.6) replacing p = f(q) by the inclusion (4.14) where

$$\begin{split} A_{i\kappa} &= A + B_p A_{q_{i\kappa}}, \ a_{i\kappa} = B_p a_{q_{i\kappa}}, \ B_{u_{i\kappa}} = B_u, \\ B_{w_{i\kappa}} &= B_w, \ C_{i\kappa} = C + D_p A_{q_{i\kappa}}, \ c_{i\kappa} = D_p a_{q_{i\kappa}}, \\ D_{u_{i\kappa}} &= D_u, \ D_{w_{i\kappa}} = D_w \end{split}$$

The next section addresses stability and performance analysis of PWA differential inclusions.

## 4.4 Analysis

In this section, stability and performance analysis of PWA differential inclusions are considered. The concept of the *parameter set* of a PWA differential inclusion is also introduced. This concept will be used to derive equivalent sets of conditions for the analysis.

#### 4.4.1 Stability analysis

In this section, the following PWA slab differential inclusion is considered

$$\dot{x} \in \mathbf{conv}\{A_{i\kappa}x + a_{i\kappa}|\kappa = 1, 2\}, \ x \in \mathcal{R}_i$$

$$(4.18)$$

$$\mathcal{R}_i = \{ x | \| L_i x + l_i \| < 1 \}$$
(4.19)

where  $L_i \in \mathbb{R}^{1 \times n}$ ,  $l_i \in \mathbb{R}$ ,  $\mathcal{R}_i \cap \mathcal{R}_j = \emptyset$  for  $i \neq j$  and  $\bigcup_{i=1}^M \overline{\mathcal{R}}_i = \mathcal{X}$ . It is assumed that  $a_{i\kappa} = 0$  for  $i \in \mathcal{I}(0)$  and  $\kappa = 1, 2$ .

**Definition 4.1.** The parameter set of the differential inclusion (4.18) is defined as

$$\Omega = \left\{ \left[ \begin{array}{cc} A_{i\kappa} & a_{i\kappa} \\ L_i & l_i \end{array} \right] \middle| i = 1, \dots, M, \ \kappa = 1, 2 \right\}$$
(4.20)

We are interested to know if all possible trajectories in  $\mathcal{X}$  asymptotically converge to the origin. Note that the right-hand-side of (4.18) is not necessarily continuous and therefore there might exist attractive sliding modes. This prevents us from using standard Lyapunov theorems. The following proposition provides sufficient conditions for the stability of system (4.18) based on Theorem 2.1.

**Proposition 4.1.** All trajectories of the PWA slab differential inclusion (4.18) in  $\mathcal{X}$  asymptotically converge to x = 0 if for a given decay rate  $\alpha > 0$ , there exist  $P = P^T \in \mathbb{R}^{n \times n}$  and  $\lambda_{i\kappa} \in \mathbb{R}$  for i = 1, ..., M and  $\kappa = 1, 2$  such that

$$P > 0, \tag{4.21}$$

$$A_{i\kappa}^T P + P A_{i\kappa} + \alpha P < 0, \ \forall i \in \mathcal{I}(0),$$
(4.22)

$$\begin{cases} \lambda_{i\kappa} < 0, \\ \begin{bmatrix} A_{i\kappa}^{T}P + PA_{i\kappa} + \alpha P + \lambda_{i\kappa}L_{i}^{T}L_{i} & Pa_{i\kappa} + \lambda_{i\kappa}l_{i}L_{i}^{T} \\ a_{i\kappa}^{T}P + \lambda_{i\kappa}l_{i}L_{i} & \lambda_{i\kappa}(l_{i}^{2} - 1) \end{bmatrix} < 0, \end{cases}$$
(4.23)

for  $i \notin \mathcal{I}(0)$ .

*Proof.* Consider the candidate Lyapunov function  $V(x) = x^{T}Px$  for the differential inclusion (4.18) where P > 0. It follows from Proposition 2.7 and the inequalities (4.19), (4.21),(4.22) and (4.23) that

$$\nabla V(x)^{\mathrm{T}}(A_{i1}x + a_{i1}) + \alpha V(x) < 0, \text{ for } x \in \overline{\mathcal{R}}_i,$$
(4.24)

 $\operatorname{and}$ 

$$\nabla V(x)^{\mathrm{T}}(A_{i2}x + a_{i2}) + \alpha V(x) < 0, \text{ for } x \in \overline{\mathcal{R}}_i$$
(4.25)

Therefore by performing a convex combination of (4.24) and (4.25)

$$\nabla V(x)^{\mathrm{T}} f + \alpha V(x) < 0, \text{ for } x \in \mathcal{R}_i, \forall f \in \mathbf{conv} \{ A_{i\kappa} x + a_{i\kappa} | i \in \mathcal{I}(x), \kappa = 1, 2 \}$$

$$(4.26)$$

Now, it follows from (4.26) and Proposition 2.2 that the differential inclusion (4.18) is dissipative with the storage function V(x) and the supply rate  $-\alpha V(x)$ . Therefore the following function

$$S(x,t) = V(x) + \int_0^t \alpha V(x(\tau)) d\tau, \qquad (4.27)$$

is nonincreasing along the trajectories of (4.18). This implies that the conditions of Theorem 2.1 are satisfied and all trajectories of the differential inclusion (4.18) in  $\mathcal{X}$  asymptotically converge to x = 0.

In the following, the new concept of the *dual parameter set* of the differential inclusion (4.18) is introduced.

**Definition 4.2.** The dual parameter set of (4.18) is defined as

$$\Omega^{T} = \left\{ \begin{bmatrix} A_{i\kappa}^{T} & L_{i}^{T} \\ a_{i\kappa}^{T} & l_{i} \end{bmatrix} \middle| i = 1, \dots, M, \kappa = 1, 2 \right\}$$
(4.28)

The importance of the dual parameter set is that if we write the LMIs in Proposition 4.1 for  $\Omega^{T}$ , the resulting LMIs are stability conditions equivalent to those of Proposition 4.1. This is shown in the following.

**Proposition 4.2.** All trajectories of the PWA slab differential inclusion (4.18) in  $\mathcal{X}$  asymptotically converge to x = 0 if for a given decay rate  $\alpha > 0$ , there exist  $Q = Q^T \in \mathbb{R}^{n \times n}$  and  $\mu_{i\kappa} \in \mathbb{R}$  for i = 1, ..., M and  $\kappa = 1, 2$  such that

$$Q > 0, \tag{4.29}$$

$$A_{i\kappa}Q + QA_{i\kappa}^T + \alpha Q < 0, \ \forall i \in \mathcal{I}(0), \ \kappa = 1,2$$

$$(4.30)$$

$$\begin{cases} \mu_{i\kappa} < 0 \\ \begin{bmatrix} A_{i\kappa}Q + QA_{i\kappa}^{T} + \alpha Q + \mu_{i\kappa}a_{i\kappa}a_{i\kappa}^{T} & QL_{i}^{T} + \mu_{i\kappa}l_{i}a_{i\kappa} \\ L_{i}Q + \mu_{i\kappa}l_{i}a_{i\kappa}^{T} & \mu_{i\kappa}(l_{i}^{2} - 1) \end{bmatrix} < 0, \end{cases}$$

$$(4.31)$$

for  $i \notin \mathcal{I}(0)$  and  $\kappa = 1, 2$ .

*Proof.* The conditions of Proposition 4.1 and Proposition 4.2 will be shown to be equivalent with the change of variables  $Q = P^{-1}$ ,  $\mu_{i\kappa} = \frac{1}{\lambda_{i\kappa}}$ . Multiplying (4.21) and (4.22) by Q from the left and right leads to (4.29) and (4.30) respectively. To show that (4.31) is equivalent to (4.23), we multiply the matrix inequality in (4.23) by the following matrix from both sides

$$\left[\begin{array}{cc} Q & 0\\ 0 & 1 \end{array}\right] \tag{4.32}$$

to get the following inequality

$$\begin{bmatrix} \begin{pmatrix} QA_{i\kappa}^{\mathrm{T}} + A_{i\kappa}Q + \alpha Q \\ + \frac{1}{\mu_{i\kappa}}QL_{i}^{\mathrm{T}}L_{i}Q \end{pmatrix} & a_{i\kappa} + \frac{1}{\mu_{i\kappa}}l_{i}QL_{i}^{\mathrm{T}} \\ a_{i\kappa}^{\mathrm{T}} + \frac{1}{\mu_{i\kappa}}l_{i}L_{i}Q & \frac{1}{\mu_{i\kappa}}(l_{i}^{2} - 1) \end{bmatrix} < 0$$

$$(4.33)$$

Using the Schur complement, inequality (4.33) is equivalent to

$$\frac{1}{\mu_{i\kappa}}(l_{i}^{2}-1) < 0$$
(4.34)
$$QA_{i\kappa}^{T} + A_{i\kappa}Q + \alpha Q + \frac{1}{\mu_{i\kappa}}QL_{i}^{T}L_{i}Q - (a_{i\kappa} + \frac{1}{\mu_{i\kappa}}l_{i}QL_{i}^{T})$$

$$\mu_{i\kappa}(l_{i}^{2}-1)^{-1}(a_{i\kappa}^{T} + \frac{1}{\mu_{i\kappa}}l_{i}L_{i}Q) < 0$$
(4.35)

Replacing  $(l_i^2 - 1)^{-1}$  by the following expression

$$(l_i^2 - 1)^{-1} = -1 + l_i^2 (l_i^2 - 1)^{-1}$$
(4.36)

leads to the following inequality after some manipulations.

$$A_{i\kappa}Q + QA_{i\kappa}^{\mathrm{T}} + \alpha Q + \mu_{i\kappa}a_{i\kappa}a_{i\kappa}^{\mathrm{T}} - (QL_{i}^{\mathrm{T}} + \mu_{i\kappa}l_{i}a_{i\kappa})$$
$$\frac{1}{\mu_{i\kappa}}(l_{i}^{2} - 1)^{-1}(L_{i}Q + \mu_{i\kappa}l_{i}a_{i\kappa}^{\mathrm{T}}) < 0$$
(4.37)

From (4.34) and the fact that  $\mu_{i\kappa} < 0$ , it follows that

$$\mu_{i\kappa}(l_i^2 - 1) < 0 \tag{4.38}$$

Using the Schur complement once again, inequalities (4.37) and (4.38) are equivalent to (4.31). Therefore, all the constraints of Proposition 4.2 and Proposition 4.1 are equivalent and all trajectories of the PWA slab differential inclusion (4.18) in  $\mathcal{X}$ asymptotically converge to x = 0.

Proposition 4.2 will be used in section 4.5 to formulate the PWA controller synthesis problem as a convex problem. In the next subsection, the  $L_2$ -gain analysis of PWA differential inclusions is discussed.

#### 4.4.2 L<sub>2</sub>-gain analysis

In this subsection, the L<sub>2</sub>-gain analysis of the following system is considered

$$\dot{x} \in \mathbf{conv} \{ A_{i\kappa} x + a_{i\kappa} + B_{w_{i\kappa}} w, \kappa = 1, 2 \}, \ (x, w) \in \mathcal{R}_{i}^{\mathcal{X} \times \mathcal{W}}$$
$$y \in \mathbf{conv} \{ C_{i\kappa} x + c_{i\kappa} + D_{w_{i\kappa}} w, \kappa = 1, 2 \}$$
$$\mathcal{R}_{i}^{\mathcal{X} \times \mathcal{W}} = \{ (x, w) | \| L_{i} x + l_{i} + M_{i} w \| < 1 \}$$

$$(4.39)$$

It is assumed that  $a_{i\kappa} = 0$ ,  $c_{i\kappa} = 0$  for  $i \in \mathcal{I}(0,0)$  and  $\kappa = 1, 2$ . It is also assumed that the PWA function  $C_{i\kappa}x + c_{i\kappa} + D_{w_{i\kappa}}w$  is continuous in x for  $\kappa = 1$  and 2. Similar to the case of stability analysis, the *parameter set* of (4.39) is defined in the following. **Definition 4.3.** The parameter set of the differential inclusion (4.39) is defined as

$$\Phi = \left\{ \begin{bmatrix} A_{i\kappa_1} & a_{i\kappa_1} & B_{w_{i\kappa_1}} \\ L_i & l_i & M_i \\ C_{i\kappa_2} & c_{i\kappa_2} & D_{w_{i\kappa_2}} \end{bmatrix} \middle| i = 1, \dots, M, \ \kappa_1 = 1, 2, \ \kappa_2 = 1, 2 \right\}$$
(4.40)

The following proposition describes sufficient conditions for the differential inclusion (4.39) to have a finite L<sub>2</sub>-gain less than  $\gamma$  from w to y.

**Proposition 4.3.** For a given  $\gamma > 0$ , the PWA slab differential inclusion (4.39) has a finite  $L_2$ -gain less than  $\sqrt{2\gamma}$  from w to y if there exists  $P = P^T \in \mathbb{R}^{n \times n}$  such that

1. P > 0

2. for  $i \in \mathcal{I}(0,0)$ ,  $\kappa_1 = 1, 2$  and  $\kappa_2 = 1, 2$ ,

$$\begin{bmatrix} A_{i\kappa_{1}}^{T}P + PA_{i\kappa_{1}} + C_{i\kappa_{2}}^{T}C_{i\kappa_{2}} & * \\ B_{w_{i\kappa_{1}}}^{T}P + D_{w_{i\kappa_{2}}}^{T}C_{i\kappa_{2}} & -\gamma^{2}I + D_{w_{i\kappa_{2}}}^{T}D_{w_{i\kappa_{2}}} \end{bmatrix} < 0$$
(4.41)

3. there exists  $\lambda_{i\kappa_1\kappa_2} < 0$  for  $i \notin \mathcal{I}(0,0)$ ,  $\kappa_1 = 1,2$  and  $\kappa_2 = 1,2$  such that

$$\begin{pmatrix}
A_{i\kappa_{1}}^{T}P + PA_{i\kappa_{1}} \\
+C_{i\kappa_{2}}^{T}C_{i\kappa_{2}} \\
+\lambda_{i\kappa_{1}\kappa_{2}}L_{i}^{T}L_{i}
\end{pmatrix} * * * \\
a_{i\kappa_{1}}^{T}P + c_{i\kappa_{2}}^{T}C_{i\kappa_{2}} + \lambda_{i\kappa_{1}\kappa_{2}}l_{i}L_{i} \quad \lambda_{i\kappa_{1}\kappa_{2}}(l_{i}^{2} - 1) + c_{i\kappa_{2}}^{T}c_{i\kappa_{2}} * \\
\begin{pmatrix}
B_{w_{i\kappa_{1}}}^{T}P \\
+D_{w_{i\kappa_{2}}}^{T}C_{i\kappa_{2}} \\
+\lambda_{i\kappa_{1}\kappa_{2}}M_{i}^{T}L_{i}
\end{pmatrix} D_{w_{i\kappa_{2}}}^{T}c_{i\kappa_{2}} + \lambda_{i\kappa_{1}\kappa_{2}}l_{i}M_{i}^{T} \begin{pmatrix}
-\gamma^{2}I \\
+D_{w_{i\kappa_{2}}}^{T}D_{w_{i\kappa_{2}}} \\
+\lambda_{i\kappa_{1}\kappa_{2}}M_{i}^{T}L_{i}
\end{pmatrix} (4.42)$$

*Proof.* Consider the storage function  $V(x) = x^{T}Px$  for the differential inclusion (4.39) where P > 0.

1. For  $i \in \mathcal{I}(0, 0)$ , inequality (4.41) implies

$$\nabla V(x)^{\mathrm{T}} f_1 + \|y_1\|^2 - \gamma^2 \|w\|^2 < 0$$
(4.43)

$$\nabla V(x)^{\mathrm{T}} f_2 + \|y_1\|^2 - \gamma^2 \|w\|^2 < 0$$
(4.44)

$$\nabla V(x)^{\mathrm{T}} f_1 + \|y_2\|^2 - \gamma^2 \|w\|^2 < 0$$
(4.45)

$$\nabla V(x)^{\mathrm{T}} f_2 + \|y_2\|^2 - \gamma^2 \|w\|^2 < 0$$
(4.46)

where

$$f_1 = A_{i1}x + a_{i1} + B_{w_{i1}}w \tag{4.47}$$

$$f_2 = A_{i2}x + a_{i2} + B_{w_{i2}}w \tag{4.48}$$

$$y_1 = C_{i1}x + c_{i1} + D_{w_{i1}}w aga{4.49}$$

$$y_2 = C_{i2}x + c_{i2} + D_{w_{i2}}w aga{4.50}$$

From (4.43) and (4.44), it follows

$$\nabla V(x)^{\mathrm{T}} f + \|y_1\|^2 - \gamma^2 \|w\|^2 < 0$$
(4.51)

for any  $f \in \operatorname{conv}\{A_{i\kappa}x + B_{w_{i\kappa}}w | i \in \mathcal{I}(0,0), \kappa = 1,2\}$ . It also follows from (4.45) and (4.46),

$$\nabla V(x)^{\mathrm{T}} f + \|y_2\|^2 - \gamma^2 \|w\|^2 < 0$$
(4.52)

Consider  $y \in \mathbf{conv}\{y_1, y_2\}$  so that

$$y = \alpha y_1 + (1 - \alpha)y_2 \tag{4.53}$$

where  $0 \le \alpha \le 1$ . One can write

$$\|y\|^{2} \leq 2(\alpha^{2}\|y_{1}\|^{2} + (1-\alpha)^{2}\|y_{2}\|^{2})$$
(4.54)

From (4.51) and (4.52), one has

$$\|y\|^{2} \leq 2[\alpha^{2} + (1-\alpha)^{2}](-\nabla V(x)^{\mathrm{T}}f + \gamma^{2}\|w\|^{2})$$
(4.55)

It is easy to verify that for  $0 \le \alpha \le 1$ ,

$$\alpha^2 + (1 - \alpha)^2 \le 1 \tag{4.56}$$

Thus,

$$\|y\|^{2} \leq 2(-\nabla V(x)^{\mathrm{T}}f + \gamma^{2}\|w\|^{2})$$
(4.57)

or

$$2\nabla V(x)^{\mathrm{T}}f + \|y\|^{2} - 2\gamma^{2}\|w\|^{2} \le 0$$
(4.58)

2. For  $i \notin \mathcal{I}(0,0)$ , it follows from the inequality (4.42) that

$$\nabla V(x)^{\mathrm{T}} f_{1} + \|y_{1}\|^{2} - \gamma^{2} \|w\|^{2} + \lambda_{i11}(\|L_{i}x + l_{i} + M_{i}w\|^{2} - 1) < 0$$
 (4.59)

$$\nabla V(x)^{\mathrm{T}} f_{2} + \|y_{1}\|^{2} - \gamma^{2} \|w\|^{2} + \lambda_{i21}(\|L_{i}x + l_{i} + M_{i}w\|^{2} - 1) < 0$$
 (4.60)

$$\nabla V(x)^{\mathrm{T}} f_1 + \|y_2\|^2 - \gamma^2 \|w\|^2 + \lambda_{i12}(\|L_i x + l_i + M_i w\|^2 - 1) < 0$$
 (4.61)

$$\nabla V(x)^{\mathrm{T}} f_2 + \|y_2\|^2 - \gamma^2 \|w\|^2 + \lambda_{i22}(\|L_i x + l_i + M_i w\|^2 - 1) < 0$$
 (4.62)

Since  $\lambda_{i\kappa_1\kappa_2} < 0$  for  $\kappa_1 = 1, 2$  and  $\kappa_2 = 1, 2$ , the expression (4.9) and the inequalities (4.59)-(4.62) imply that inequalities (4.43)-(4.46) are satisfied for  $x \in \overline{\mathcal{R}}_i$  and  $i \notin \mathcal{I}(0,0)$ . Using the same arguments as above, one can conclude that the inequality (4.58) is satisfied  $x \in \overline{\mathcal{R}}_i$  and  $i \notin \mathcal{I}(0,0)$ .

Therefore the PWA differential inclusion (4.39) is dissipative with the storage function 2V(x) and the supply rate  $W(x, w) = 2\gamma^2 ||w||^2 - ||y||^2$  and the L<sub>2</sub>-gain of (4.39) from w to y is less than  $\sqrt{2\gamma}$ .

The dual parameter set is defined in the following. Note that, in this chapter, we consider polytopic PWA slab differential inclusions. Therefore  $l_i = l_i^{T}$  is a scalar. **Definition 4.4.** The dual parameter set of (4.39) is defined as

$$\Phi^{T} = \left\{ \begin{bmatrix} A_{i\kappa_{1}}^{T} & L_{i}^{T} & C_{i\kappa_{2}}^{T} \\ a_{i\kappa_{1}}^{T} & l_{i} & c_{i\kappa_{2}}^{T} \\ B_{w_{i\kappa_{1}}}^{T} & M_{i}^{T} & D_{w_{i\kappa_{2}}}^{T} \end{bmatrix} \middle| i = 1, \dots, M, \ \kappa_{1} = 1, 2, \ \kappa_{2} = 1, 2 \right\}$$
(4.63)

The following proposition shows that the L<sub>2</sub>-gain LMIs for  $\Phi$  in Proposition 4.3 are equivalent to the L<sub>2</sub>-gain LMIs for  $\Phi^{T}$ .

**Proposition 4.4.** For a given  $\gamma > 0$ , the PWA slab differential inclusion (4.39) has a finite  $L_2$ -gain less than  $\sqrt{2\gamma}$  from w to y if there exists  $Q = Q^T \in \mathbb{R}^{n \times n}$  such that

1. Q > 0,

2. for  $i \in \mathcal{I}(0,0)$ ,  $\kappa_1 = 1, 2$  and  $\kappa_2 = 1, 2$ ,

$$\begin{bmatrix} A_{i\kappa_{1}}Q + QA_{i\kappa_{1}}^{T} + B_{w_{i\kappa_{1}}}B_{w_{i\kappa_{1}}}^{T} & * \\ C_{i\kappa_{2}}Q + D_{w_{i\kappa_{2}}}B_{w_{i\kappa_{1}}}^{T} & -\gamma^{2}I + D_{w_{i\kappa_{2}}}D_{w_{i\kappa_{2}}}^{T} \end{bmatrix} < 0$$
(4.64)

3. there exists  $\mu_{i\kappa_1\kappa_2} < 0$  for  $i \notin \mathcal{I}(0,0)$ ,  $\kappa_1 = 1, 2$  and  $\kappa_2 = 1, 2$  such that

$$\begin{bmatrix} \begin{pmatrix} A_{i\kappa_{1}}Q + QA_{i\kappa_{1}}^{T} \\ +B_{w_{i\kappa_{1}}}B_{w_{i\kappa_{1}}}^{T} \\ +\mu_{i\kappa_{1}\kappa_{2}}a_{i\kappa_{1}}a_{i\kappa_{1}}^{T} \end{pmatrix} * & * \\ \mu_{i\kappa_{1}\kappa_{2}}a_{i\kappa_{1}}a_{i\kappa_{1}}^{T} \end{pmatrix} & * & * \\ L_{i}Q + M_{i}B_{w_{i\kappa_{1}}}^{T} + \mu_{i\kappa_{1}\kappa_{2}}l_{i}a_{i\kappa_{1}}^{T} & \mu_{i\kappa_{1}\kappa_{2}}(l_{i}^{2}-1) + M_{i}M_{i}^{T} & * \\ \begin{pmatrix} C_{i\kappa_{2}}Q + D_{w_{i\kappa_{2}}}B_{w_{i\kappa_{1}}}^{T} \\ +\mu_{i\kappa_{1}\kappa_{2}}c_{i\kappa_{2}}a_{i\kappa_{1}}^{T} \end{pmatrix} & D_{w_{i\kappa_{2}}}M_{i}^{T} + \mu_{i\kappa_{1}\kappa_{2}}l_{i}c_{i\kappa_{2}} & \begin{pmatrix} -\gamma^{2}I \\ +D_{w_{i\kappa_{2}}}D_{w_{i\kappa_{2}}}^{T} \\ +\mu_{i\kappa_{1}\kappa_{2}}c_{i\kappa_{2}}c_{i\kappa_{2}}^{T} \end{pmatrix} \\ (4.65)$$

*Proof.* In the following, it will be shown that the conditions of Proposition 4.4 and that of Proposition 4.3 are equivalent with the change of variables  $Q = \gamma^2 P^{-1}$  and  $\mu_{i\kappa_1\kappa_2} = \frac{1}{\lambda_{i\kappa_1\kappa_2}}$ . Multiplying (4.41) from both sides by the following matrix

$$\left[\begin{array}{cc} Q & 0 \\ 0 & I \end{array}\right] \tag{4.66}$$

leads to the following inequality.

$$\begin{bmatrix} \gamma^2 Q A_{i\kappa_1}^{\mathrm{T}} + \gamma^2 A_{i\kappa_1} Q + Q C_{i\kappa_2}^{\mathrm{T}} C_{i\kappa_2} Q & * \\ \gamma^2 B_{w_{i\kappa_1}}^{\mathrm{T}} + D_{w_{i\kappa_2}}^{\mathrm{T}} C_{i\kappa_2} Q & -\gamma^2 I + D_{w_{i\kappa_2}}^{\mathrm{T}} D_{w_{i\kappa_2}} \end{bmatrix} < 0$$
(4.67)

It follows from applying the Schur complement that

$$-\gamma^{2}I + D_{w_{i\kappa_{2}}}^{\mathrm{T}} D_{w_{i\kappa_{2}}} < 0$$

$$\gamma^{2}QA_{i\kappa_{1}}^{\mathrm{T}} + \gamma^{2}A_{i\kappa_{1}}Q + QC_{i\kappa_{2}}^{T}C_{i\kappa_{2}}Q - (\gamma^{2}B_{w_{i\kappa_{1}}} + QC_{i\kappa_{2}}^{\mathrm{T}}D_{w_{i\kappa_{2}}})$$

$$(-\gamma^{2}I + D_{w_{i\kappa_{2}}}^{\mathrm{T}} D_{w_{i\kappa_{2}}})^{-1}(\gamma^{2}B_{w_{i\kappa_{1}}}^{\mathrm{T}} + D_{w_{i\kappa_{2}}}^{\mathrm{T}}C_{i\kappa_{2}}Q) < 0$$

$$(4.69)$$

Using the matrix inversion lemma [68], we have

$$(-\gamma^{2}I + D_{w_{i\kappa_{2}}}^{\mathrm{T}}D_{w_{i\kappa_{2}}})^{-1} = -\frac{1}{\gamma^{2}}(I - \frac{1}{\gamma^{2}}D_{w_{i\kappa_{2}}}^{\mathrm{T}}D_{w_{i\kappa_{2}}})^{-1}$$
$$= -\frac{1}{\gamma^{2}}\left(I + D_{w_{i\kappa_{2}}}^{\mathrm{T}}(\gamma^{2}I - D_{w_{i\kappa_{2}}}D_{w_{i\kappa_{2}}}^{\mathrm{T}})^{-1}D_{w_{i\kappa_{2}}}\right)$$
(4.70)

Replacing (4.70) into (4.69), after some manipulations we get

$$A_{i\kappa_{1}}Q + QA_{i\kappa_{1}}^{\mathrm{T}} + B_{w_{i\kappa_{1}}}B_{w_{i\kappa_{1}}}^{\mathrm{T}} - (QC_{i\kappa_{2}}^{\mathrm{T}} + B_{w_{i\kappa_{1}}}D_{w_{i\kappa_{2}}}^{\mathrm{T}})$$
$$(-\gamma^{2}I + D_{w_{i\kappa_{2}}}D_{w_{i\kappa_{2}}}^{\mathrm{T}})^{-1}(C_{i\kappa_{2}}Q + D_{w_{i\kappa_{2}}}B_{w_{i\kappa_{2}}}^{\mathrm{T}}) < 0$$
(4.71)

From (4.68) and the fact that nonzero eigenvalues of  $D_{w_{i\kappa_2}}^{\mathrm{T}} D_{w_{i\kappa_2}}$  and  $D_{w_{i\kappa_2}} D_{w_{i\kappa_2}}^{\mathrm{T}}$  are equal, one can conclude

$$-\gamma^2 I + D_{w_{i\kappa_2}} D_{w_{i\kappa_2}}^{\mathrm{T}} < 0 \tag{4.72}$$

Applying the Schur complement to (4.71) and (4.72) yields inequality (4.64).

The steps of the proof for inequality (4.42) are similar. If one writes (4.42) as

$$\left[\begin{array}{cc} \psi_{11} & * \\ \psi_{21} & \psi_{22} \end{array}\right] < 0 \tag{4.73}$$

where

$$\psi_{11} = \begin{bmatrix} A_{i\kappa_1}^{\mathrm{T}} & 0\\ a_{i\kappa_1}^{\mathrm{T}} & 1 \end{bmatrix} \begin{bmatrix} P & 0\\ 0 & -\frac{\lambda_{i\kappa_1\kappa_2}}{2} \end{bmatrix} + \begin{bmatrix} P & 0\\ 0 & -\frac{\lambda_{i\kappa_1\kappa_2}}{2} \end{bmatrix} \begin{bmatrix} A_{i\kappa_1} & a_{i\kappa_1}\\ 0 & 1 \end{bmatrix} + \begin{bmatrix} C_{i\kappa_2}^{\mathrm{T}} & L_i^{\mathrm{T}}\\ c_{i\kappa_2}^{\mathrm{T}} & l_i \end{bmatrix} \begin{bmatrix} I & 0\\ 0 & \lambda_{i\kappa_1\kappa_2}I \end{bmatrix} \begin{bmatrix} C_{i\kappa_2} & c_{i\kappa_2}\\ L_i & l_i \end{bmatrix}$$
(4.74)

$$\psi_{21} = \begin{bmatrix} B_{w_{i\kappa_{1}}}^{\mathrm{T}} & 0 \end{bmatrix} \begin{bmatrix} P & 0 \\ 0 & -\frac{\lambda_{i\kappa_{1}\kappa_{2}}}{2} \end{bmatrix} + \begin{bmatrix} D_{w_{i\kappa_{2}}}^{\mathrm{T}} & M_{i}^{\mathrm{T}} \end{bmatrix} \begin{bmatrix} I & 0 \\ 0 & \lambda_{i\kappa_{1}\kappa_{2}}I \end{bmatrix}$$
(4.75)

$$\psi_{22} = -\gamma^2 I + \begin{bmatrix} D_{w_{i\kappa_2}}^{\mathrm{T}} & M_i^{\mathrm{T}} \end{bmatrix} \begin{bmatrix} I & 0 \\ 0 & \lambda_{i\kappa_1\kappa_2}I \end{bmatrix} \begin{bmatrix} D_{w_{i\kappa_2}} \\ M_i \end{bmatrix}$$
(4.76)

and multiply it from both sides by the following matrix

$$\begin{bmatrix} Q & 0 \\ 0 & \mu_{i\kappa_1\kappa_2} \end{bmatrix} = 0$$

$$\begin{bmatrix} 0 & \mu_{i\kappa_1\kappa_2} \end{bmatrix} = 0$$

$$\begin{bmatrix} 0 & I \end{bmatrix}$$

$$(4.77)$$

Then, using the following matrix inversion

$$\begin{pmatrix} -\gamma^{2}I + \begin{bmatrix} D_{w_{i\kappa_{2}}}^{\mathrm{T}} & M_{i}^{\mathrm{T}} \end{bmatrix} \begin{bmatrix} I & 0 \\ 0 & \lambda_{i\kappa_{1}\kappa_{2}}I \end{bmatrix} \begin{bmatrix} D_{w_{i\kappa_{2}}} \\ M_{i} \end{bmatrix} \end{pmatrix}^{-1} = \\ -\frac{1}{\gamma^{2}} \begin{pmatrix} I + \begin{bmatrix} D_{w_{i\kappa_{2}}}^{\mathrm{T}} & M_{i}^{\mathrm{T}} \end{bmatrix} (\gamma^{2} \begin{bmatrix} I & 0 \\ 0 & \mu_{i\kappa_{1}\kappa_{2}}I \end{bmatrix} \\ - \begin{bmatrix} D_{w_{i\kappa_{2}}} \\ M_{i} \end{bmatrix} \begin{bmatrix} D_{w_{i\kappa_{2}}}^{\mathrm{T}} & M_{i}^{\mathrm{T}} \end{bmatrix})^{-1} \begin{bmatrix} D_{w_{i\kappa_{2}}} \\ M_{i} \end{bmatrix} \end{pmatrix}$$
(4.78)

and the Schur complement, after some manipulation, we get (4.65). Therefore all the constraints of Propositions 4.4 and 4.3 are equivalent and the PWA slab differential inclusion (4.39) has a finite L<sub>2</sub>-gain less than  $\gamma$  from w to y.

Proposition 4.4 is an important result, which enables us to formulate the  $L_2$ gain synthesis as a convex optimization problem in the next section.

# 4.5 Controller Synthesis

In this section, PWA controller synthesis for PWA slab differential inclusions will be formulated as a convex optimization program.

#### 4.5.1 Stabilizability

Consider the following PWA slab differential inclusion.

$$\dot{x} \in \operatorname{conv}\{A_{i\kappa}x + a_{i\kappa} + B_{u_{i\kappa}}u|\kappa = 1, 2\}, \ x \in \mathcal{R}_i$$

$$\mathcal{R}_i = \{x \mid ||L_ix + l_i|| < 1\}$$
(4.79)

We seek a PWA control signal of the form  $u = K_i x + k_i$  for  $x \in \mathcal{R}_i$  to stabilize (4.79) to the origin. It follows from Proposition 4.2 that there exists a control signal

$$u = K_i x + k_i, \ x \in \mathcal{R}_i \tag{4.80}$$

such that all trajectories of the PWA slab differential inclusion (4.79) in  $\mathcal{X}$  asymptotically converge to x = 0 if for a given decay rate  $\alpha > 0$ , there exist  $Q \in \mathbb{R}^{n \times n}$ and  $\mu_i \in \mathbb{R}$  such that

$$Q > 0, \tag{4.81}$$

$$A_{i\kappa}Q + QA_{i\kappa}^{\mathrm{T}} + B_{u_{i\kappa}}Y_i + Y_i^{\mathrm{T}}B_{u_{i\kappa}}^{\mathrm{T}} + \alpha Q < 0, \qquad (4.82)$$

for  $i\in\mathcal{I}(0),\kappa=1,2$  , and

$$\mu_i < 0 \tag{4.83}$$

$$\begin{bmatrix}
\begin{pmatrix}
A_{i\kappa}Q + QA_{i\kappa}^{\mathrm{T}} \\
+B_{u_{i\kappa}}Y_{i} + Y_{i}^{\mathrm{T}}B_{u_{i\kappa}}^{\mathrm{T}} \\
+\alpha Q + \mu_{i}a_{i\kappa}a_{i\kappa}^{\mathrm{T}} \\
+a_{i\kappa}Z_{i}^{\mathrm{T}}B_{u_{i\kappa}}^{\mathrm{T}} + B_{u_{i\kappa}}Z_{i}a_{i\kappa}^{\mathrm{T}} \\
+B_{u_{i\kappa}}W_{i}B_{u_{i\kappa}}^{\mathrm{T}}
\end{pmatrix} * \\
\begin{pmatrix}
L_{i}Q + \mu_{i}l_{i}a_{i\kappa}^{\mathrm{T}} \\
+l_{i}Z_{i}^{\mathrm{T}}B_{u_{i\kappa}}^{\mathrm{T}}
\end{pmatrix} \mu_{i\kappa}(l_{i}^{2} - 1)
\end{bmatrix} \leq 0, \quad (4.84)$$

for  $i \notin \mathcal{I}(0)$  and  $\kappa = 1, 2$  where

$$Y_i = K_i Q \tag{4.85}$$

$$Z_i = \mu_i k_i \tag{4.86}$$

$$W_i = \mu_i k_i k_i^{\mathrm{T}} \tag{4.87}$$

The main problem in designing a PWA controller using constraints (4.81)-(4.87) is the equality constraint (4.87) because it prevents the problem to be formulated as a convex optimization. We propose two methods to overcome this difficulty.

1. Convex relaxation: The first approach to formulate PWA controller synthesis as a convex program comes from the observation that considering  $\mu_i < 0$  and (4.87), we have

$$W_i \le 0 \tag{4.88}$$

Therefore the term  $B_{u_{i\kappa}}W_iB_{u_{i\kappa}}^{\mathrm{T}}$  in (4.84) is always negative semi-definite and can be omitted to make the problem convex. This idea leads to the following proposition.

**Proposition 4.5.** There exists a PWA controller of the from (4.80) such that all trajectories of the PWA slab differential inclusion (4.79) in  $\mathcal{X}$  asymptotically converge to x = 0 if for a given decay rate  $\alpha > 0$ , there exist  $Q = Q^T \in \mathbb{R}^{n \times n}$  and  $\mu_i \in \mathbb{R}$  such that for all  $i \in \mathcal{I}(0)$  and  $\kappa = 1, 2$ 

$$Q > 0, \tag{4.89}$$

$$A_{i\kappa}Q + QA_{i\kappa}^{T} + B_{u_{i\kappa}}Y_i + Y_i^{T}B_{u_{i\kappa}}^{T} + \alpha Q < 0, \qquad (4.90)$$

and for all  $i \notin \mathcal{I}(0)$  and  $\kappa = 1, 2$ 

$$\mu_{i} < 0$$

$$\left[ \begin{pmatrix} A_{i\kappa}Q + QA_{i\kappa}^{T} \\ B_{u_{i\kappa}}Y_{i} + Y_{i}^{T}B_{u_{i\kappa}}^{T} \\ +\alpha Q + \mu_{i}a_{i\kappa}a_{i\kappa}^{T} \\ +a_{i\kappa}Z_{i}^{T}B_{u_{i\kappa}}^{T} + B_{u_{i\kappa}}Z_{i}a_{i\kappa}^{T} \end{pmatrix} * \\ \begin{pmatrix} L_{i}Q + \mu_{i}l_{i}a_{i\kappa}^{T} \\ +l_{i}Z_{i}^{T}B_{u_{i\kappa}}^{T} \end{pmatrix} \mu_{i\kappa}(l_{i}^{2} - 1) \right] \leq 0, \quad (4.92)$$

2. *Trace heuristic:* Equality constraint (4.87) can be written as the following rank minimization.

min Rank 
$$X_i$$
  
s.t.  $X_i = \begin{bmatrix} W_i & Z_i \\ Z_i^{\mathrm{T}} & \mu_i \end{bmatrix} \le 0$  (4.93)

A well-known heuristic to solve the rank minimization problem (4.93) is to maximize the trace instead of minimizing the rank (for a negative semi-definite matrix) [39]. Using this approach, the following proposition formulates sufficient conditions for the existence of a PWA controller for differential inclusion (4.79) as a convex optimization problem.

**Proposition 4.6.** There exists a PWA controller of the form (4.80) such that all trajectories of the PWA slab differential inclusion (4.79) in  $\mathcal{X}$  asymptotically converge to x = 0 if for a given decay rate  $\alpha > 0$ , there exist  $Q = Q^T \in \mathbb{R}^{n \times n}$  and  $\mu_i \in \mathbb{R}$  such that the following optimization problem

$$\max \sum_{i=1}^{M} Trace X_i$$
(4.94)

subject to

$$X_{i} = \begin{bmatrix} W_{i} & Z_{i} \\ Z_{i}^{T} & \mu_{i} \end{bmatrix} \leq 0, \forall i \notin \mathcal{I}(0,0),$$

$$(4.95)$$

and (4.81)-(4.84) has a feasible solution such that

$$\mu_i W_i = Z_i Z_i^T \tag{4.96}$$

For both Propositions 4.5 and 4.6, the PWA controller gains can be computed as

$$K_i = Y_i Q^{-1} (4.97)$$

$$k_{i} = \begin{cases} 0 & \text{if } i \in \mathcal{I}(0) \\ \frac{1}{\mu_{i}} Z_{i} & \text{otherwise} \end{cases}$$
(4.98)

In the next subsection,  $L_2$ -gain PWA controller synthesis is formulated as a convex optimization problem.
#### 4.5.2 L<sub>2</sub>-gain synthesis

In this subsection, the objective is to design a PWA control signal of the form (4.80) to limit the L<sub>2</sub>-gain from w to y for the following differential inclusion

$$\dot{x} \in \operatorname{conv} \{ A_{i\kappa} x + a_{i\kappa} + B_{u_{i\kappa}} u + B_{w_{i\kappa}} w | \kappa = 1, 2 \},$$

$$y \in \operatorname{conv} \{ C_{i\kappa} x + c_{i\kappa} + D_{u_{i\kappa}} u + D_{w_{i\kappa}} w \},$$

$$for (x, w) \in \mathcal{R}_{i}^{\mathcal{X} \times \mathcal{W}} = \{ (x, w) | \| L_{i} x + l_{i} + M_{i} w \| < 1 \}$$

$$(4.99)$$

Similar to the case of stabilizability, the same convex relaxations can be used to yield the following propositions.

**Proposition 4.7.** For a given  $\gamma > 0$ , there exists a PWA controller of the form (4.80) such that the PWA slab differential inclusion (4.99) has a finite  $L_2$ -gain less than  $\sqrt{2}\gamma$  from w to y if there exist  $Q = Q^T \in \mathbb{R}^{n \times n}$  and  $\mu_i \in \mathbb{R}$ , such that for all  $i \in \mathcal{I}(0,0)$  and  $\kappa_1 = 1, 2$  and  $\kappa_2 = 1, 2$ 

 $Q > 0, \tag{4.100}$ 

$$\begin{bmatrix} \begin{pmatrix} A_{i\kappa_{1}}Q + B_{u_{i\kappa_{1}}}Y_{i} \\ +QA_{i\kappa_{1}}^{T} + Y_{i}^{T}B_{u_{i\kappa_{1}}}^{T} \\ +B_{w_{i\kappa_{1}}}B_{w_{i\kappa_{1}}}^{T} \end{pmatrix} * \\ \begin{pmatrix} C_{i\kappa_{2}}Q + D_{u_{i\kappa_{2}}}Y_{i} \\ +D_{w_{i\kappa_{2}}}B_{w_{i\kappa_{1}}}^{T} \end{pmatrix} -\gamma^{2}I + D_{w_{i\kappa_{2}}}D_{w_{i\kappa_{2}}}^{T} \end{bmatrix} < 0$$
(4.101)

93

and for all  $i \notin \mathcal{I}(0,0)$ ,  $\mu_i < 0$ ,  $\kappa_1 = 1, 2$  and  $\kappa_2 = 1, 2$ 

**Proposition 4.8.** For given  $\gamma > 0$ , there exists a PWA controller of the form (4.80) such that the PWA slab differential inclusion (4.99) has finite  $L_2$ -gain less than  $\sqrt{2}\gamma$  from w to y if there exist  $Q = Q^T \in \mathbb{R}^{n \times n}$  and  $\mu_i \in \mathbb{R}$  such that the following optimization problem

$$\max \sum_{i=1}^{M} Trace X_i$$
(4.103)

subject to (4.100)-(4.102) and

$$\mu_i < 0, \ X_i = \begin{bmatrix} W_i & Z_i \\ Z_i^T & \mu_i \end{bmatrix} \le 0,$$
(4.104)

has a feasible solution that satisfies

$$\mu_i W_i = Z_i Z_i^T \tag{4.105}$$

Note that the PWA controller gains can be computed using (4.97) and (4.98).



Figure 4.1: The computed bounding envelope for a scalar function

# 4.6 Numerical Examples

**Example 4.1.** In this example, we consider the surge model of a jet engine taken from [72]. The model is described by the following state equations.

$$\begin{cases} \dot{x}_1 = -x_2 - \frac{3}{2}x_1^2 - \frac{1}{2}x_1^3 \\ \dot{x}_2 = u \end{cases}$$
(4.106)

Using the proposed method in subsection 4.3.1, a bounding envelope is computed for the nonlinear function  $f(x_1) = -\frac{3}{2}x_1^2 - \frac{1}{2}x_1^3$  which is shown in Fig. 4.1.

By substituting the PWA bounds in (4.106), we get a differential inclusion with

$$\mathcal{R}_{1} = \begin{pmatrix} -4 & -2.5 \end{pmatrix}, \qquad \mathcal{R}_{2} = \begin{pmatrix} -2.5 & -1 \end{pmatrix}, \\ \mathcal{R}_{3} = \begin{pmatrix} -1 & 1.5 \end{pmatrix}, \qquad \mathcal{R}_{4} = \begin{pmatrix} 1.5 & 4.0 \end{pmatrix}, \\ A_{11} = \begin{bmatrix} -19 & -1 \\ 0 & 0 \end{bmatrix}, \qquad A_{21} = \begin{bmatrix} 2.013 & -1 \\ 0 & 0 \end{bmatrix}, \\ A_{31} = \begin{bmatrix} -5.938 & -1 \\ 0 & 0 \end{bmatrix}, \qquad A_{41} = \begin{bmatrix} 1.413 & -1 \\ 0 & 0 \end{bmatrix},$$



The approximation error of the nonlinear function is considered as a disturbance input (w) and the objective is to limit the  $L_2$ -gain from w to  $x_1$ . Using the computed PWA approximation, the nonlinear system (4.106) can be described by the following differential inclusion

$$\dot{x} \in conv\{A_{i\kappa}x + a_{i\kappa} + B_uu + B_ww | \kappa = 1, 2\}, x \in \mathcal{R}_i$$

$$y = Cx + D_ww + D_uu$$
(4.107)

where  $i = 1, ..., 4, \kappa = 1, 2$  and

$$B_w = \begin{bmatrix} 1\\0 \end{bmatrix}, \ C = \begin{bmatrix} 1&0 \end{bmatrix}, \ D_u = 0, \ D_w = 0$$
(4.108)

Using Proposition 4.7, the following PWA controller was obtained for  $\gamma = 0.2$  using



Figure 4.2: Trajectories of the closed loop nonlinear system

SeDuMi [121] and Yalmip [76].

$$K_{1} = \begin{bmatrix} 1036.3 & -40.37 \end{bmatrix}, k_{1} = 280.40$$

$$K_{2} = \begin{bmatrix} 769.20 & -30.27 \end{bmatrix}, k_{2} = 0.03$$

$$K_{3} = \begin{bmatrix} 899.15 & -34.97 \end{bmatrix}, k_{3} = 0$$

$$K_{4} = \begin{bmatrix} 597.57 & -23.57 \end{bmatrix}, k_{4} = -17.39$$
(4.109)

Figure 4.2 shows the trajectories of the nonlinear system (4.106) in closed loop connection with the PWA controller. The contours of the storage function are also shown in Figure 4.2. Note that the PWA controller designed for the differential inclusion is guaranteed to stabilize the nonlinear system.

**Example 4.2.** In this example, we consider a nonlinear system with a discontinuous vector field. The model is described by the following state equations.

$$\dot{x}_{1} = f(x_{1}) - 2 \operatorname{sgn}(x_{1}) x_{2} + w$$
  
$$\dot{x}_{2} = 2|x_{1}| - 2x_{2} + u$$
  
$$y = x_{1} + x_{2} - w$$
 (4.110)



Figure 4.3: The computed bounding envelope for a scalar function

where  $f(x_1) = -x_1 - 0.5x_1^3$ . Using the proposed method in [109], a bounding envelope is computed for the nonlinear function  $f(x_1)$  which is shown in Fig. 4.3. Trajectories of the open loop system are shown in Fig. 4.4. Notice that a sliding mode exists on  $x_1 = 0$ .

The objective is to design a PWA controller u to limit the  $L_2$ -gain from the disturbance w to the output y. Substituting  $f(x_1)$  by its PWA bounds in (4.110), one gets a PWA differential inclusion with

$$\mathcal{R}_1 = \begin{pmatrix} -4 & -2 \end{pmatrix}, \qquad \mathcal{R}_2 = \begin{pmatrix} -2 & 0 \end{pmatrix}, \\ \mathcal{R}_3 = \begin{pmatrix} 0 & 2 \end{pmatrix}, \qquad \mathcal{R}_4 = \begin{pmatrix} 2 & 4 \end{pmatrix}$$

Using the PWA approximation proposed in subsection 4.3.1, the nonlinear system (4.110) can be described by the following differential inclusion

$$\dot{x} \in conv\{A_{i\kappa}x + a_{i\kappa} + B_uu + B_ww\}, x \in \mathcal{R}_i$$
$$y = Cx + D_ww + D_uu$$
(4.111)



Figure 4.4: Trajectories of the closed loop nonlinear system

where  $i = 1, \ldots, 4$ ,  $\kappa = 1, 2$  and

$$B_w = \begin{bmatrix} 1\\ 0 \end{bmatrix}, \ C = \begin{bmatrix} 1 & 0 \end{bmatrix}, \ D_u = 0, \ D_w = 0$$
(4.112)

Using Proposition 4.4, one can solve a set of LMIs using SeDuMi [121] and Yalmip [76] to compute  $\gamma = 1.2$  for the open loop system. The following PWA controller can then be calculated using Proposition 4.7 to achieve  $\gamma = 1$  for the closed loop system.

$$K_{1} = \begin{bmatrix} -22.4985 & -83.6140 \end{bmatrix}, k_{1} = 0.7609$$

$$K_{2} = \begin{bmatrix} -31.6286 & -117.2289 \end{bmatrix}, k_{2} = 0$$

$$K_{3} = \begin{bmatrix} -25.6282 & -96.8814 \end{bmatrix}, k_{3} = 0$$

$$K_{4} = \begin{bmatrix} -35.2165 & -114.3215 \end{bmatrix}, k_{4} = 0.0489$$
(4.113)

Figure 4.5 shows the trajectories of the nonlinear system (4.110) in closed loop connection with the PWA controller. Note that the PWA controller designed for the differential inclusion is guaranteed to limit the  $L_2$  gain of the nonlinear system.



Figure 4.5: Trajectories of the closed loop nonlinear system

# 4.7 Conclusions

In this chapter, an interesting duality relation was revealed in the LMIs describing sufficient conditions for the stability of PWA slab differential inclusions. This concept was then employed to find the duality relation for the  $L_2$ -gain design. As a result, the definition of the regions of a PWA slab system was extended, the  $L_2$ -gain controller design was formulated as a set of LMIs and this design method was extended to PWA slab systems with an output that is also a PWA function of the state. The new method presented in the chapter enables stability and performance analysis, as well as controller synthesis, for a large class of nonlinear systems as a solution of convex optimization problems.

# Chapter 5

# Backstepping Controller Synthesis for Piecewise Polynomial Systems: A Sum of Squares Approach

This chapter addresses backstepping controller synthesis for piecewise polynomial (PWP) systems. The main contribution of the chapter is to formulate controller design for a large class of PWP systems as a convex problem. Integrator backstepping is proposed as the principal design step in constructing Lyapunov functions for PWP systems in *strict feedback form*. The controller synthesis problem is divided into two cases. The first case consists of the construction of a sum of squares (SOS) Lyapunov function for PWP systems with discontinuous vector fields. The second case addresses the construction of a piecewise polynomial Lyapunov function for PWP systems with continuous vector fields. After constructing a (piecewise) polynomial Lyapunov function, controller synthesis for a PWP system can be formulated as an SOS program, which is a convex optimization problem and can be solved efficiently using available software [76]. The new synthesis method is applied to two numerical examples to illustrate its effectiveness.

# 5.1 Introduction

PWP or spline approximation of curves and surfaces has been widely used in many different scientific contexts and engineering applications [1,34]. However, the lack of practical methods to check the sign of polynomials has prevented PWP systems to be commonly used in the field of control systems. One of the first attempts to design controllers for PWP systems was made in [87]. Paul proposed in [87] to partition the state space of a nonlinear system that is affine in the input into cells and to approximate the dynamics of the system in each cell by a model that is polynomial in the state. A controller is then designed for each cell using feedback linearization. A global controller is then formed by joining the individual cell controllers. The proposed method was employed in [87] to design controllers for a few examples of nonlinear systems. However, there is no guarantee for the stability of the closed loop system because a switched system consisting of stable subsystems can be unstable in general.

Recently, the class of discrete-time PWP systems was defined in [45] and a new method was proposed to address the constrained finite-time optimal control problem for this class of systems. This seems to be the first systematic approach to controller synthesis for discrete-time PWP systems. However, according to the authors of [45], the method suffers from excessive computational burden.

For continuous time PWP systems, a stability analysis was proposed in [93] and [85] using piecewise polynomial Lyapunov functions. The advantage of the proposed method is that the analysis problem is formulated as a sum of squares (SOS) programming which is a convex optimization problem. There exist numerical tools such as SOSTOOLS [95] to solve SOS programming problems efficiently. However, systems with infinitely fast switching or sliding modes are excluded from the discussion in [93] and [85].

The main contribution of this chapter is to propose a backstepping technique

to construct control Lyapunov functions for a class of PWP systems. The proposed method formulates the control synthesis problem for PWP systems in *strict feedback form* as an SOS feasibility problem. The synthesis of PWP controllers is formulated for two cases. The first case addresses the construction of (SOS) Lyapunov functions [85] for PWP systems with discontinuous vector fields. The second case deals with the construction of piecewise polynomial Lyapunov functions for PWP systems with continuous vector fields. After constructing a (piecewise) polynomial Lyapunov function, controller synthesis for a PWP system can be formulated as an SOS program, which is a convex optimization problem and therefore can be solved efficiently.

The chapter is organized as follows. Integrator backstepping is addressed in section 5.2. Controller design for PWP systems in strict feedback form is then described in section 5.3. Finally, a numerical example is demonstrated in section 5.4.

## 5.2 Integrator Backstepping

Before introducing the recursive backstepping controller design, integrator backstepping is presented in this section for its simplicity. Consider the following PWP system

$$\dot{x} = f_i(x) + g_i(x)z, \ x \in \mathcal{P}_i \tag{5.1}$$

where  $x \in \mathbb{R}^n$ ,  $z \in \mathbb{R}^{n_z}$  and  $\mathcal{P}_i$  for i = 1, ..., M is defined in section 2.8. Assume that there exist a stabilizing polynomial controller  $z = \gamma(x)$  and a Lyapunov function V(x) which proves the stability of the closed loop system. Consider adding an integrator to this system, which yields the following PWP system

$$\dot{x} = f_i(x) + g_i(x)z, \ x \in \mathcal{P}_i$$
  
$$\dot{z} = u$$
(5.2)

The objective is to design a controller to stabilize the augmented system (5.2). In the following, two approaches to this problem are discussed. The first approach is to construct an SOS Lyapunov function for the case of a PWP system with discontinuous vector fields. The second approach builds a piecewise polynomial Lyapunov function for the case where the vector field of the PWP system (5.2) is continuous.

#### 5.2.1 PWP systems with discontinuous vector fields

Consider the PWP system (5.1). Assume that there exists a polynomial control  $z = \gamma(x)$  where  $\gamma(x) \in \mathbb{R}^{n_z}$  is a vector of polynomials so that  $\gamma(0) = 0$  and V(x) is an SOS Lyapunov function for the closed loop system verifying

$$\begin{cases} V(x) - \lambda(x) \text{ is SOS} \\ -\nabla V(x)^{\mathrm{T}}(f_i(x) + g_i(x)\gamma(x)) - \Gamma_i(x)^{\mathrm{T}}E_i(x) - \alpha V(x) \text{ is SOS} \end{cases}$$
(5.3)

for i = 1, ..., M and any  $\alpha > 0$ , where  $\lambda(x)$  is a positive definite SOS polynomial,  $\Gamma_i(x)$  is an SOS vector function and  $E_i(x)$  is defined in section 2.8. Consider now the following candidate Lyapunov function for system (5.2),

$$V_{\gamma}(x,z) = V(x) + \frac{1}{2}(z - \gamma(x))^{\mathrm{T}}(z - \gamma(x))$$
(5.4)

Note that  $V_{\gamma}(x, z)$  is a positive definite function. To compute a PWP controller, one can write

$$\nabla V_{\gamma}(x,z)^{\mathrm{T}} \begin{bmatrix} f_i(x) + g_i(x)z \\ u \end{bmatrix} = \nabla V(x)^{\mathrm{T}} (f_i(x) + g_i(x)z) + (z - \gamma(x))^{\mathrm{T}} [u - \frac{d\gamma(x)}{dx} (f_i(x) + g_i(x)z)]$$

$$=\nabla V(x)^{\mathrm{T}}(f_{i}(x) + g_{i}(x)\gamma(x)) + \nabla V(x)^{\mathrm{T}}g_{i}(x)(z - \gamma(x))$$

$$+ (z - \gamma(x))^{\mathrm{T}}[u - \frac{d\gamma(x)}{dx}(f_{i}(x) + g_{i}(x)z)]$$

$$=\nabla V(x)^{\mathrm{T}}(f_{i}(x) + g_{i}(x)\gamma(x)) + (z - \gamma(x))^{\mathrm{T}}[u + g_{i}^{\mathrm{T}}(x)\nabla V(x)$$

$$- \frac{d\gamma(x)}{dx}(f_{i}(x) + g_{i}(x)z)]$$
(5.5)

Using the following expression

$$u(x,z) = -g_i^{\rm T}(x)\nabla V(x) + \frac{d\gamma(x)}{dx}(f_i(x) + g_i(x)z) - \frac{\alpha}{2}(z - \gamma(x)), \qquad (5.6)$$

and the SOS constraints (5.3) leads to the following inequality

$$\nabla V_{\gamma}(x,z)^{\mathrm{T}} \begin{bmatrix} f_i(x) + g_i(x)z \\ u(x,z) \end{bmatrix} \leq -\alpha V_{\gamma}(x,z) - \Gamma_i(x)^{\mathrm{T}} E_i(x)$$
(5.7)

Therefore it follows from (2.137) that for  $x \in \overline{\mathcal{P}}_i$  and  $i = 1, \dots, M$ 

$$\nabla V_{\gamma}(x,z)^{\mathrm{T}} \left[ \begin{array}{c} f_{i}(x) + g_{i}(x)z\\ u(x,z) \end{array} \right] \leq -\alpha V_{\gamma}(x,z)$$
(5.8)

From (5.8) and based on Proposition 2.2, it follows that the PWP system (5.2) with the following controller

$$u(x,z) = -g_i^{\mathrm{T}}(x)\nabla V(x) + \frac{d\gamma(x)}{dx}(f_i(x) + g_i(x)z) - \frac{\alpha}{2}(z - \gamma(x)), \ x \in \mathcal{P}_i$$
(5.9)

is dissipative with the storage function  $V_{\gamma}(x, z)$  and supply rate  $-\alpha V_{\gamma}(x, z)$ . Therefore it follows from Theorem 2.1 that the PWP system (5.2) with controller (5.9) is asymptotically stable.

In summary, integrator backstepping consists of two steps:

- Lyapunov function construction: The candidate Lyapunov function (5.4) was constructed using a known Lyapunov function V(x) and polynomial controller  $\gamma(x)$  for the PWP system (5.1).
- Controller synthesis: The control law (5.9) was designed to make the candidate Lyapunov function (5.4) decreasing with time.

#### 5.2.2 PWP systems with continuous vector fields

Assume that the vector field of the PWP system (5.1) is continuous for  $x \in \mathcal{X}$  and there exists a continuous PWP control  $z = \gamma(x)$  where

$$\gamma(x) = \gamma_i(x), \ x \in \mathcal{P}_i \tag{5.10}$$

where  $\gamma_i(x) \in \mathbb{R}^{n_z}$  is a vector polynomial so that the continuous piecewise polynomial

$$V(x) = V_i(x), \ x \in \mathcal{P}_i \tag{5.11}$$

where  $V_i(x)$  is a polynomial function verifying the following constraints

$$\begin{cases} V_i(x) - \Lambda_i^{\mathrm{T}}(x)E_i(x) - \lambda(x) \text{ is SOS} \\ -\nabla V_i(x)^{\mathrm{T}}(f_i(x) + g_i(x)\gamma_i(x)) - \Gamma_i^{\mathrm{T}}(x)E_i(x) - \alpha V_i(x) \text{ is SOS} \end{cases}$$
(5.12)

for i = 1, ..., M where  $\alpha$  is a positive scalar,  $\Lambda_i(x)$  and  $\Gamma_i(x)$  are SOS vector functions and  $E_i(x)$  is defined in section 2.8. In the following, the objective is to construct a PWP Lyapunov function and a PWP controller for the PWP system (5.2).

• Lyapunov function construction: Consider now the following candidate Lyapunov function for system (5.2)

$$V_{\gamma}(x,z) = V_{\gamma_i}(x,z), \ x \in \mathcal{P}_i \tag{5.13}$$

where

$$V_{\gamma_i}(x,z) = V_i(x) + \frac{1}{2}(z - \gamma_i(x))^{\mathrm{T}}(z - \gamma_i(x))$$
(5.14)

Note that  $V_{\gamma}(x, z)$  is a continuous piecewise polynomial function because V(x)and  $\gamma(x)$  are continuous piecewise polynomial functions. To compute a PWP controller, we write

$$\nabla V_{\gamma_i}(x,z)^{\mathrm{T}} \begin{bmatrix} f_i(x) + g_i(x)z \\ u \end{bmatrix} = \nabla V_i(x)^{\mathrm{T}} (f_i(x) + g_i(x)z) + (z - \gamma_i(x))^{\mathrm{T}} [u - \frac{d\gamma_i(x)}{dx} (f_i(x) + g_i(x)z)] = \nabla V_i(x)^{\mathrm{T}} (f_i(x) + g_i(x)\gamma_i(x)) + (z - \gamma_i(x))^{\mathrm{T}} [u + g_i(x)^{\mathrm{T}} \nabla V_i(x) - \frac{d\gamma_i(x)}{dx} (f_i(x) + g_i(x)z)]$$
(5.15)

• Controller synthesis: Using the following expression

$$u(x,z) = -g_i(x)^{\mathrm{T}} \nabla V_i(x) + \frac{d\gamma_i(x)}{dx} (f_i(x) + g_i(x)z) - \frac{\alpha}{2} (z - \gamma_i(x)), \quad (5.16)$$

the SOS constraint (5.12) and also (2.137) leads to the following inequality

$$\nabla V_{\gamma_i}(x,z)^{\mathrm{T}} \begin{bmatrix} f_i(x) + g_i(x)z \\ u \end{bmatrix} \le -\alpha V_{\gamma_i}(x,z)$$
(5.17)

for  $x \in \overline{\mathcal{P}}_i$  and  $i = 1, \ldots, M$ . Therefore if the following controller

$$u(x,z) = -g_i(x)^{\mathrm{T}} \nabla V_i(x) + \frac{d\gamma_i(x)}{dx} (f_i(x) + g_i(x)z) - \frac{\alpha}{2} (z - \gamma_i(x)), \ x \in \mathcal{P}_i$$
(5.18)

is a continuous function for  $x \in \mathcal{X}$ , based on Proposition 2.3, the PWP system (5.2) with controller (5.18) is dissipative with the storage function  $V_{\gamma}(x, z)$  and supply rate  $-\alpha V_{\gamma}(x, z)$ . Therefore it follows from Theorem 2.1 that the PWP system (5.2) with controller (5.18) is asymptotically stable. However, there is no guarantee that the control input in (5.18) is continuous.

The more general case of recursive backstepping controller design for PWP systems is formulated as a set of SOS programs in the next section.

## 5.3 Recursive Backstepping Controller Design

In this section, a recursive PWP controller synthesis method is proposed for *strict feedback* PWP systems of the following form

$$\dot{x}_{1} = f_{1i_{1}}(x_{1}) + g_{1i_{1}}(x_{1})x_{2}, \qquad \text{for } x_{1} \in \mathcal{P}_{1i_{1}}$$

$$\dot{x}_{2} = f_{2i_{2}}(x_{1}, x_{2}) + g_{2i_{2}}(x_{1}, x_{2})x_{3}, \qquad \text{for } \begin{bmatrix} x_{1} \\ x_{2} \end{bmatrix} \in \mathcal{P}_{2i_{2}}$$

$$\vdots \qquad (5.19)$$

$$\dot{x}_{k} = f_{ki_{k}}(x_{1}, x_{2}, \dots, x_{k}) + g_{ki_{k}}(x_{1}, x_{2}, \dots, x_{k})u, \quad \text{for } \begin{bmatrix} x_{1} \\ x_{2} \\ \vdots \\ x_{k} \end{bmatrix} \in \mathcal{P}_{ki_{k}}$$

where the state vector of the system (5.19) is divided into k subvectors:

$$x = \begin{bmatrix} x_1 \\ x_2 \\ \vdots \\ x_k \end{bmatrix} \in \mathbb{R}^n, \ x_j \in \mathbb{R}^{n_j},$$
(5.20)

with  $\sum_{j=1}^{k} n_j = n$ . For each  $j \in \{1, 2, ..., k\}$ , the regions  $\mathcal{P}_{ji_j}$  for  $i_j = 1, ..., M_j$  are disjoint sets defined as

$$\mathcal{P}_{ji_j} = \left\{ \begin{bmatrix} x_1 \\ \vdots \\ x_j \end{bmatrix} \middle| E_{ji_j}(x_1, \dots, x_j) \succ 0 \right\}$$
(5.21)

where  $E_{jij}(x_1, \ldots, x_j) \in \mathbb{R}^{p_j}$  is a vector polynomial function and  $\succ$  denotes an elementwise inequality. For a given j, the regions  $\mathcal{P}_{jij}$  for  $i_j = 1, \ldots, M_j$  partition the projection of the state space  $\mathcal{X} \subset \mathbb{R}^n$  onto the  $(x_1, \ldots, x_j)$  space.

Assumption 5.1. It is assumed that for  $1 \le j_1 < j_2$ , the projection of each region  $\mathcal{P}_{j_2 i_{j_2}}$  for  $i_{j_2} = 1, \ldots, M_{j_2}$  on the  $(x_1, \ldots, x_{j_1})$  space is a subset of only one of the

regions  $\mathcal{P}_{j_1i_{j_1}}$  for  $i_{j_1} = 1, \ldots, M_{j_1}$ . In other words, for each  $1 < j_2 \leq k, j_1 < j_2$  and  $i_{j_2} \in \{1, \ldots, M_{j_2}\}$  there exists a unique number  $i(j_1, j_2, i_{j_2})$  in  $\{1, \ldots, M_{j_1}\}$  such that

$$\begin{bmatrix} x_1 \\ \vdots \\ x_{j_2} \end{bmatrix} \in \mathcal{P}_{j_2 i_{j_2}} \Rightarrow \begin{bmatrix} x_1 \\ \vdots \\ x_{j_1} \end{bmatrix} \in \mathcal{P}_{j_1 i(j_1, j_2, i_{j_2})}$$
(5.22)

Assumption 5.2. It is also assumed that

$$f_{ji_j^*}(0,\ldots,0) = 0, \ \forall i_j^* \in \mathcal{I}_j(0,\ldots,0)$$
 (5.23)

where

$$\mathcal{I}_{j}(x_{1},\ldots,x_{j}) := \left\{ i_{j} \left| \left[ \begin{array}{c} x_{1} \\ \vdots \\ x_{j} \end{array} \right] \in \mathcal{P}_{ji_{j}} \right\}$$
(5.24)

In what follows the stabilization problem for PWP systems in strict feedback form is discussed for two cases of PWP systems: discontinuous and continuous vector fields.

#### 5.3.1 PWP systems with discontinuous vector fields

To design a PWP controller for (5.19), we start from the following subsystem

$$\dot{x}_1 = f_{1i_1}(x_1) + g_{1i_1}(x_1)x_2, \text{ for } x_1 \in \mathcal{P}_{1i_1},$$
(5.25)

(5.26)

with  $i_1 = 1, ..., M_1$ . It is assumed that there exist a polynomial Lyapunov function  $V_1(x_1)$  and a polynomial controller  $x_2 = \gamma_1(x_1)$  such that for  $i_1 = 1, ..., M_1$ 

$$\begin{cases} \gamma_1(0) = 0 \\ V_1(0) = 0 \\ V_1(x_1) - \lambda(x_1) \text{ is SOS} \\ -\nabla V_1(x_1)^{\mathrm{T}}(f_{1i_1}(x_1) + g_{1i_1}(x_1)\gamma_1(x_1)) - \Gamma_{1i_1}(x_1)^{\mathrm{T}}E_{1i_1}(x_1) - \alpha V_1(x_1) \text{ is SOS} \\ \Gamma_{1i_1}(x_1) \in \mathbb{R}^{p_1} \text{ is SOS} \end{cases}$$

where  $\alpha > 0$  and  $\lambda(x_1)$  is a positive definite polynomial.

Next, a polynomial controller should be designed for the following subsystem

$$\dot{x}_{1} = f_{1i_{1}}(x_{1}) + g_{1i_{1}}(x_{1})x_{2}, \text{ for } x_{1} \in \mathcal{P}_{1i_{1}}$$

$$\dot{x}_{2} = f_{2i_{2}}(x_{1}, x_{2}) + g_{2i_{2}}(x_{1}, x_{2})x_{3}, \text{ for } \begin{bmatrix} x_{1} \\ x_{2} \end{bmatrix} \in \mathcal{P}_{2i_{2}}$$
(5.27)

Note that if  $f_{2i_2}(x_1, x_2) = 0$  and  $g_{2i_2}(x_1, x_2) = 1$ , this would be an integrator backstepping problem. The design process consists of two steps.

1. Lyapunov function construction: Consider the following Lyapunov function

$$V_2(x_1, x_2) = V_1(x_1) + \frac{1}{2}(x_2 - \gamma_1(x_1))^{\mathrm{T}}(x_2 - \gamma_1(x_1))$$
(5.28)

2. Controller synthesis: The synthesis problem can be formulated as the following SOS program.

Find 
$$x_3 = \gamma_2(x_1, x_2), \ \Gamma_{2i_2}(x_1, x_2)$$
  
s.t.  $-\nabla_{x_1} V_2(x_1, x_2)^{\mathrm{T}} (f_{1i(1,2,i_2)}(x_1) + g_{1i(1,2,i_2)}(x_1)x_2)$   
 $-\nabla_{x_2} V_2(x_1, x_2)^{\mathrm{T}} (f_{2i_2}(x_1, x_2) + g_{2i_2}(x_1, x_2)x_3)$   
 $-\Gamma_{2i_2}(x_1, x_2)^{\mathrm{T}} E_{2i_2}(x_1, x_2) - \alpha V_2(x_1, x_2) \text{ is SOS},$   
 $\Gamma_{2i_2}(x_1, x_2) \text{ is SOS}$   
 $\gamma_2(0, 0) = 0$  (5.29)

where  $i_2 = 1, ..., M_2$  and  $\gamma_2(x_1, x_2)$  is a polynomial function of  $x_1$  and  $x_2$ . If this SOS program is feasible then the procedure can be repeated for the next steps by adding the dynamics of  $x_3$  and so on.

Assume that all the SOS programs in the backstepping procedure are feasible and we reach the last step with the following candidate Lyapunov function.

$$V_k(x_1, \dots, x_k) = V_{k-1}(x_1, \dots, x_{k-1}) + \frac{1}{2}(x_k - \gamma_{k-1}(x_1, \dots, x_{k-1}))^{\mathrm{T}}(x_k - \gamma_{k-1}(x_1, \dots, x_{k-1}))$$
(5.30)

where  $\gamma_{k-1}(x_1, \ldots, x_{k-1})$  is a polynomial function. Note that  $V_k(x_1, \ldots, x_k)$  is also a polynomial function. The final controller  $u = \gamma_k(x_1, \ldots, x_k)$  will not be used to construct another SOS Lyapunov function. Therefore it does not have to be continuously differentiable. One can hence search for a PWP control of the form

$$u = \gamma_{ki_k}(x_1, \dots, x_k), \text{ for } \begin{bmatrix} x_1 \\ \vdots \\ x_k \end{bmatrix} \in \mathcal{P}_{ki_k}$$
(5.31)

for  $i_k = 1, ..., M_k$ . In case that all  $g_{ki_k}$  for  $i_k = 1, ..., M_k$  are invertible, this step of the controller synthesis can be converted to integrator backstepping. In general, this step can be formulated as the following SOS program

Find 
$$u = \gamma_{ki_k}(x_1, \dots, x_k), \ \Gamma_{ki_k}(x_1, \dots, x_k)$$
  
s.t.  $-\nabla_{x_1} V_k(x_1, \dots, x_k)^{\mathrm{T}} (f_{1i(1,k,i_k)}(x_1) + g_{1i(1,k,i_k)}(x_1)x_2)$   
 $-\nabla_{x_2} V_k(x_1, \dots, x_k)^{\mathrm{T}} (f_{2i(2,k,i_k)}(x_1, x_2) + g_{2i(2,k,i_k)}(x_1, x_2)x_3)$   
 $-\dots - \nabla_{x_k} V_k(x_1, \dots, x_k)^{\mathrm{T}} (f_{ki_k}(x_1, \dots, x_k) + g_{ki_k}(x_1, \dots, x_k)u)$   
 $-\Gamma_{ki_k}(x_1, \dots, x_k)^{\mathrm{T}} E_{ki_k}(x_1, \dots, x_k) - \alpha V_k(x_1, \dots, x_k)$  is SOS,  
 $\Gamma_{ki_k}(x_1, \dots, x_k)$  is SOS (5.32)

for  $i_k = 1, ..., M_k$ . The following theorem shows that if the SOS program (5.32) is feasible then the PWP controller (5.31) stabilizes the PWP system (5.19).

**Theorem 5.1.** Let there exist polynomial functions  $V_1(x_1)$  and  $\gamma_1(x_1)$  satisfying (5.26) and let  $V_j(x_1, \ldots, x_j)$  for  $j = 2, \ldots, k$  be defined as

$$V_{j}(x_{1},...,x_{j}) = V_{j-1}(x_{1},...,x_{j-1}) + \frac{1}{2}(x_{j} - \gamma_{j-1}(x_{1},...,x_{j-1}))^{T}(x_{j} - \gamma_{j-1}(x_{1},...,x_{j-1}))$$
(5.33)

where

$$\gamma_j(\overbrace{0,\ldots,0}^{j \text{ arguments}}) = 0, \ j = 1,\ldots,k-1$$
(5.34)

Also assume that the PWP control (5.31) satisfies the conditions of the SOS program (5.32). Then the PWP control (5.31) makes the trajectories of the PWP system (5.19) in  $\mathcal{X}$  asymptotically converge to the origin.

*Proof.* It follows from (5.26) that  $V_1(x_1) \ge \lambda(x_1)$  and since  $\lambda(x_1)$  is positive definite,

$$V_1(x_1) > 0$$
, if  $x_1 \neq 0$  (5.35)

From (5.33) we have

$$V_k(x_1, \dots, x_k) = V_1(x_1) + \sum_{j=2}^k \frac{1}{2} (x_j - \gamma_{j-1}(x_1, \dots, x_{j-1}))^{\mathrm{T}} (x_j - \gamma_{j-1}(x_1, \dots, x_{j-1}))$$
(5.36)

Therefore  $V_k(x_1, \ldots, x_k)$  is nonnegative. Now assume for some  $x_1, x_2, \ldots$  and  $x_k$ , we have  $V_k(x_1, \ldots, x_k) = 0$ . It follows from (5.36) that

$$V_1(x_1) = 0 (5.37)$$

and

$$x_j = \gamma_{j-1}(x_1, \dots, x_{j-1}), \ j = 2, \dots, k$$
 (5.38)

Now, from (5.34) and positive definiteness of  $V_1(x_1)$  it follows that  $x_1 = 0, x_2 = 0, ...$ and  $x_k = 0$ . Therefore  $V_k(x_1, ..., x_k)$  is a positive definite function.

From (5.21) and (5.32), it follows that

$$\nabla_x V_k(x_1, \dots, x_k)^{\mathrm{T}} \dot{x} \le -\alpha V_k(x_1, \dots, x_k), \text{ for } x \in \mathcal{P}_{ki_k}, i_k = 1, \dots, M_k \qquad (5.39)$$

Now, from Proposition 2.2 it follows that the PWP system (5.19) is dissipative with the storage function  $V_k(x_1, \ldots, x_k)$  and supply rate  $-\alpha V_k(x_1, \ldots, x_k)$ . From the fact that  $V_k(x_1, \ldots, x_k)$  is a positive definite function and Theorem 2.1 it then follows that the trajectories of the PWP system (5.19) in  $\mathcal{X}$  asymptotically converge to the origin.

#### 5.3.2 PWP systems with continuous vector fields

In this section, it is assumed that the vector field of PWP system (5.19) is continuous for  $x \in \mathcal{X}$ . It is also assumed that for the following subsystem

$$\dot{x}_1 = f_{1i_1}(x_1) + g_{1i_1}(x_1)x_2, \text{ for } x_1 \in \mathcal{P}_{1i_1},$$
(5.40)

with  $i_1 = 1, ..., M_1$ , there exist a continuous piecewise polynomial Lyapunov function  $V_1(x_1)$  and a continuous PWP controller  $x_2 = \gamma_1(x_1)$  with

$$\begin{cases} V_1(x_1) = V_{1i_1}(x_1) \\ , \text{ for } x_1 \in \mathcal{P}_{i_1}, \\ \gamma_1(x_1) = \gamma_{1i_1}(x_1) \end{cases}$$
(5.41)

such that  $\gamma_{1i_1}(x_1)$  and  $V_{1i_1}(x_1)$  are polynomials and for  $i_1 = 1, \ldots, M_1$ 

$$\begin{cases} V_{1}(0) = 0 \\ \gamma_{1}(0) = 0 \\ V_{1i_{1}}(x_{1}) - \Lambda_{1i_{1}}(x_{1})^{\mathrm{T}} E_{1i_{1}}(x_{1}) - \lambda(x_{1}) \text{ is SOS} \\ -\nabla V_{1i_{1}}(x_{1})^{\mathrm{T}} (f_{1i_{1}}(x_{1}) + g_{1i_{1}}(x_{1})\gamma_{1i_{1}}(x_{1})) - \Gamma_{1i_{1}}(x_{1})^{\mathrm{T}} E_{1i_{1}}(x_{1}) - \alpha V_{1i_{1}} \text{ is SOS} \\ \Lambda_{1i_{1}}(x_{1}) \text{ and } \Gamma_{1i_{1}}(x_{1}) \text{ are SOS} \end{cases}$$

$$(5.42)$$

where  $\alpha > 0$  and  $\lambda(x_1)$  is a positive definite polynomial.

Then, a PWP controller should be designed for the following subsystem

$$\begin{cases} \dot{x}_1 = f_{1i_1}(x_1) + g_{1i_1}(x_1)x_2, \text{ for } x_1 \in \mathcal{P}_{1i_1} \\ \dot{x}_2 = f_{2i_2}(x_1, x_2) + g_{2i_2}(x_1, x_2)x_3, \text{ for } \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} \in \mathcal{P}_{2i_2} \end{cases}$$

$$(5.43)$$

Considering the following PWP Lyapunov function

$$V_2(x_1, x_2) = V_{2i_2}(x_1, x_2), \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} \in \mathcal{P}_{2i_2}$$
(5.44)

where

$$V_{2i_2}(x_1, x_2) = V_{1i(1,2,i_2)}(x_1) + \frac{1}{2}(x_2 - \gamma_{1i(1,2,i_2)}(x_1))^{\mathrm{T}}(x_2 - \gamma_{1i(1,2,i_2)}(x_1))$$
(5.45)

the synthesis problem can be formulated as the following SOS program

Find 
$$x_3 = \gamma_{2i_2}(x_1, x_2), \Gamma_{2i_2}(x_1, x_2), c_{i_{21}i_{22}}(x_1, x_2)$$
  
s.t.  $-\nabla_{x_1} V_{2i_2}(x_1, x_2)^{\mathrm{T}}(f_{1i(1,2,i_2)}(x_1) + g_{1i(1,2,i_2)}(x_1)x_2)$   
 $-\nabla_{x_2} V_{2i_2}(x_1, x_2)^{\mathrm{T}}(f_{2i_2}(x_1, x_2) + g_{2i_2}(x_1, x_2)x_3)$   
 $-\Gamma_{2i_2}(x_1, x_2)^{\mathrm{T}} E_{2i_2}(x_1, x_2) - \alpha V_{2i_2}(x_1, x_2)$  is SOS,  
 $\Gamma_{2i_2}(x_1, x_2)$  is SOS  
 $\gamma_{2i_{21}}(x_1, x_2) - \gamma_{2i_{22}}(x_1, x_2) = c_{i_{21}i_{22}}(x_1, x_2) E_{2i_{21}i_{22}}(x_1, x_2)$  (5.46)

for  $i_2 = 1, \ldots, M_2$  and all  $i_{21}$  and  $i_{22}$  in  $\{1, \ldots, M_2\}$  such that  $\mathcal{P}_{2i_{21}}$  and  $\mathcal{P}_{2i_{22}}$  are neighboring cells and  $E_{2i_{21}i_{22}}(x_1, x_2) = 0$  contains their boundary, i.e.

$$\overline{\mathcal{P}}_{2i_{21}} \cap \overline{\mathcal{P}}_{2i_{22}} \subset \left\{ \left[ \begin{array}{c} x_1 \\ x_2 \end{array} \right] \middle| E_{2i_{21}i_{22}}(x_1, x_2) = 0 \right\}$$
(5.47)

In addition,  $\gamma_{2i_2}(x_1, x_2)$  and  $c_{i_{21}i_{22}}(x_1, x_2)$  are polynomial functions.

If this SOS program is feasible then the procedure can be repeated for the next steps by adding the dynamics of  $x_3$  and so on until  $x_k$ . If all SOS programs in the backstepping procedure are feasible, a continuous PWP controller

$$u = \gamma_{ki_k}(x_1, \dots, x_k), \text{ for } \begin{bmatrix} x_1 \\ \vdots \\ x_k \end{bmatrix} \in \mathcal{P}_{ki_k}$$
(5.48)

can be designed using the final SOS program

Find 
$$u = \gamma_{ki_k}(x_1, \dots, x_k), \ \Gamma_{ki_k}(x_1, \dots, x_k), c_{i_{k1}i_{k2}}(x_1, \dots, x_k)$$
  
s.t.  $-\nabla_{x_1}V_{ki_k}(x_1, \dots, x_k)^{\mathrm{T}}(f_{1i(1,k,i_k)}(x_1) + g_{1i(1,k,i_k)}(x_1)x_2)$   
 $-\nabla_{x_2}V_{ki_k}(x_1, \dots, x_k)^{\mathrm{T}}(f_{2i(2,k,i_k)}(x_1, x_2) + g_{2i(2,k,i_k)}(x_1, x_2)x_3)$   
 $-\dots - \nabla_{x_k}V_{ki_k}(x_1, \dots, x_k)^{\mathrm{T}}(f_{ki_k}(x_1, \dots, x_k) + g_{ki_k}(x_1, \dots, x_k)u)$   
 $-\Gamma_{ki_k}(x_1, \dots, x_k)^{\mathrm{T}}E_{ki_k}(x_1, \dots, x_k) - \alpha V_{ki_k}(x_1, \dots, x_k) \text{ is SOS},$   
 $\Gamma_{ki_k}(x_1, \dots, x_k) \text{ is SOS}$   
 $f_{ki_{k1}}(x_1, x_2, \dots, x_k) + g_{ki_{k1}}(x_1, x_2, \dots, x_k)\gamma_{ki_{k1}}(x_1, \dots, x_k)$   
 $-f_{ki_{k2}}(x_1, x_2, \dots, x_k) + g_{ki_{k2}}(x_1, x_2, \dots, x_k)\gamma_{ki_{k2}}(x_1, \dots, x_k)$   
 $= c_{i_{k1}i_{k2}}(x_1, \dots, x_k)E_{ki_{k1}i_{k2}}(x_1, \dots, x_k)$ 
(5.49)

for  $i_k = 1, \ldots, M_k$  and all  $i_{k1}$  and  $i_{k2}$  in  $\{1, \ldots, M_k\}$  such that  $\mathcal{P}_{ki_{k1}}$  and  $\mathcal{P}_{ki_{k2}}$ are neighboring cells where  $\gamma_{ki_k}(x_1, \ldots, x_k)$  and  $c_{i_{k1}i_{k2}}(x_1, \ldots, x_k)$  are polynomial functions.

**Remark 5.1.** The equality constraint in (5.49) is equivalent to the continuity of the vector fields of the closed loop system. Therefore, if the vector field of the open loop system is discontinuous, the controller should be discontinuous to make the resulting vector field continuous. Note that the final controller will not be used to construct a Lyapunov function.

**Theorem 5.2.** Let there exist a PWP function  $V_1(x_1)$  satisfying (5.42) and  $V_j(x_1, \ldots, x_j)$ for  $j = 2, \ldots, k$  is defined as

$$V_{j}(x_{1},...,x_{j}) = V_{j-1}(x_{1},...,x_{j-1}) + \frac{1}{2}(x_{j} - \gamma_{j-1}(x_{1},...,x_{j-1}))^{T}(x_{j} - \gamma_{j-1}(x_{1},...,x_{j-1}))$$
(5.50)

where

$$\gamma_j(\overbrace{0,...,0}^{j \text{ arguments}}) = 0, \ j = 1,...,k-1$$
 (5.51)

Also assume that the PWP control (5.48) satisfies the conditions of the SOS program (5.49). Then the PWP control (5.48) makes the trajectories of the PWP system (5.19) in  $\mathcal{X}$  asymptotically converge to the origin.

*Proof.* It follows from (5.42) and (5.21) that  $V_1(x_1) \ge \lambda(x_1)$  and since  $\lambda(x_1)$  is positive definite,

$$V_1(x_1) > 0$$
, if  $x_1 \neq 0$  (5.52)

From (5.50) we have

$$V_k(x_1, \dots, x_k) = V_1(x_1) + \sum_{j=2}^k \frac{1}{2} (x_j - \gamma_{j-1}(x_1, \dots, x_{j-1}))^{\mathrm{T}} (x_j - \gamma_{j-1}(x_1, \dots, x_{j-1}))$$
(5.53)

Therefore  $V_k(x_1, \ldots, x_k) \ge 0$ . Now assume for some  $x_1, x_2, \ldots$  and  $x_k$ , we have  $V_k(x_1, \ldots, x_k) = 0$ . It follows from (5.53) that

$$V_1(x_1) = 0 \tag{5.54}$$

and

$$x_j = \gamma_{j-1}(x_1, \dots, x_{j-1}), \ j = 2, \dots, k$$
 (5.55)

Now, from (5.51) and positive definiteness of  $V_1(x_1)$  it follows that  $x_1 = 0, x_2 = 0, ...$ and  $x_k = 0$ . Therefore  $V_k(x_1, ..., x_k)$  is a positive definite function.

From (5.21) and (5.49), it follows that

$$\nabla_x V_k(x_1, \dots, x_k)^{\mathrm{T}} \dot{x} \le -\alpha V_k(x_1, \dots, x_k), \text{ for } x \in \mathcal{P}_{ki_k}, i_k = 1, \dots, M_k$$
 (5.56)

Now, from Proposition 2.3 it follows that the PWP system (5.19) is dissipative with the storage function  $V_k(x_1, \ldots, x_k)$  and supply rate  $-\alpha V_k(x_1, \ldots, x_k)$ . From the fact that  $V_k(x_1, \ldots, x_k)$  is a positive definite function and Theorem 2.1 it then follows that the trajectories of the PWP system (5.19) in  $\mathcal{X}$  asymptotically converge to the origin.

# 5.4 Numerical Examples

**Example 5.1.** Consider the following PWA system:

$$\dot{x}_{1} = \begin{cases} -0.25x_{1} + 0.05x_{2} & \text{if } x_{1} < 0.2 \\ 0.1x_{1} + 0.05x_{2} - 0.07 & \text{if } 0.2 < x_{1} < 0.6 \\ -0.2x_{1} + 0.05x_{2} + 0.11 & \text{if } x_{1} > 0.6 \end{cases}$$
  
$$\dot{x}_{2} = -20x_{1} - 30x_{2} + 24 + 20u \qquad (5.57)$$

The objective is to stabilize the system to  $x_{cl} = \begin{bmatrix} 0.6429 & 0.3714 \end{bmatrix}^T$ . Consider  $V_1(x_1) = \frac{1}{2}(x_1 - 0.6429)^2$  and the following system

$$\dot{x}_{1} = \begin{cases} -0.25x_{1} + 0.05x_{2} & \text{if } x_{1} < 0.2 \\ 0.1x_{1} + 0.05x_{2} - 0.07 & \text{if } 0.2 < x_{1} < 0.6 \\ -0.2x_{1} + 0.05x_{2} + 0.11 & \text{if } x_{1} > 0.6 \end{cases}$$
(5.58)

The following expression for  $x_2$  can stabilize this system to  $x_1 = 0.6429$ 

$$x_2 = \gamma(x_1) = 0.3714 - 4.8344(x_1 - 0.6429)$$
(5.59)

Considering the following candidate Lyapunov function

$$V_2(x_1, x_2) = \frac{1}{2}(x_1 - 0.6429)^2 + \frac{1}{2}(x_2 - 0.3714 + 4.8344(x_1 - 0.6429))^2, \quad (5.60)$$

the following PWA control input can be computed for the whole PWA system using the method presented in subsection 5.3.2.

$$u = \begin{cases} -0.35009 - 0.1216x_1 + 1.2572x_2, & x_1 < 0.2 \\ -0.34175 - 0.20165x_1 + 1.2603x_2, & 0.2 < x_1 < 0.6 \\ -0.3784 - 0.13739x_1 + 1.2567x_2, & x_1 > 0.6 \end{cases}$$
(5.61)



Figure 5.1: The trajectory of the closed-loop PWA system for  $x(0) = [0.1 \ 0.5]^T$ The trajectory of the closed-loop PWA system for  $x(0) = [0.1 \ 0.5]^T$  is shown in Fig. 5.1.

**Example 5.2.** Consider the single-link flexible-joint robot in Fig. 5.2. The dynamic equations of the robot are given by [115]

$$\dot{x}_1 = x_2 \tag{5.62}$$

$$\dot{x}_2 = -\frac{MgL}{I}\sin(x_1) - \frac{K}{I}(x_1 - x_3)$$
(5.63)

$$\dot{x}_3 = x_4 \tag{5.64}$$

$$\dot{x}_4 = -\frac{T_f}{J} + \frac{K}{J}(x_1 - x_3) + \frac{1}{J}u$$
(5.65)

where  $x_1 = \theta_1$ ,  $x_2 = \dot{\theta}_1$ ,  $x_3 = \theta_2$  and  $x_4 = \dot{\theta}_2$ . u is the motor torque and  $T_f = f_2(x_4)$ denotes the motor friction which is described by [119]

$$T_f = b_m x_4 + \operatorname{sgn}(x_4) \left( F_{cm} + (F_{sm} - F_{cm}) \exp(-\frac{x_4^2}{c_m^2}) \right)$$
(5.66)



Figure 5.2: Single-link flexible-joint robot

The numerical values of the parameters are given as follows

| M = 0.25 kg               | L = 1m            | $I = \frac{ML^2}{3}$ |
|---------------------------|-------------------|----------------------|
| K = 7.47 Nm/rad           | $J = 0.216 kgm^2$ | $g = 9.8 m/s^2$      |
| $c_m = 1.2 rad/sec$       | $F_{cm} = 1.2 Nm$ | $F_{sm} = 1.75 Nm$   |
| $b_m = 0.17 Nm/(rad/sec)$ |                   |                      |

. . . 0

Fig. 5.3 depicts the state response of the open loop nonlinear model of the robot with the initial condition  $x_0 = [\pi \ 0 \ 0.8\pi \ 0]^T$ . It can be seen that the system converges to a limit cycle. The limit cycles due to friction forces are investigated in [119]. In this example, the objective is to stabilize the nonlinear model at the origin.

To build a PWP model, there are two nonlinear functions that should be approximated by PWP curves. The function  $f_1(x_1) = \sin(x_1)$  is approximated by the



Figure 5.3: State variables of the nonlinear model - open loop

following function for  $x_1 \in [-\pi, \pi]$ 

$$\hat{f}_{1}(x_{1}) = \begin{cases} 0.4031x_{1}^{2} + 1.2464x_{1} - 0.0211 & -\pi \leq x_{1} \leq -\frac{2\pi}{7} \\ 0.908x_{1} & -\frac{2\pi}{7} \leq x_{1} \leq \frac{2\pi}{7} \\ -0.4031x_{1}^{2} + 1.2464x_{1} + 0.0211 & \frac{2\pi}{7} \leq x_{1} \leq \pi \end{cases}$$
(5.67)

The nonlinear function  $T_f = f_2(x_4)$  in (5.66) is approximated by the following PWP function for  $x_4 \in [-8, 8]$ 

$$\hat{f}_2(x_4) = \begin{cases} -0.0057x_4^3 + 0.0873x_4^2 - 0.2472x_4 + 1.8056 & x_4 > 0\\ -0.0057x_4^3 - 0.0873x_4^2 - 0.2472x_4 - 1.8056 & x_4 < 0 \end{cases}$$
(5.68)

Next, the PWP approximation of the nonlinear model (5.62)-(5.65) can be written in the strict feedback form (5.19). To start the controller synthesis procedure in subsection 5.3.2, we first consider the following system

$$\dot{x}_1 = x_2 \tag{5.69}$$



Figure 5.4: PWP approximation of  $f_1(x_1)$ 



Figure 5.5: PWP approximation of  $f_2(x_4)$ 

with

$$\mathcal{P}_{11} = \{ x_1 | x_1 \in \mathbb{R} \}$$
 (5.70)

The linear controller  $x_2 = -2x_1$  is considered in this step to make the quadratic Lyapunov function  $V_1(x_1) = \frac{1}{2}x_1^2$  decreasing with time.

In the second step, the following PWP system is considered

$$\dot{x}_1 = x_2$$
  
$$\dot{x}_2 = -\frac{MgL}{I}\hat{f}_1(x_1) - \frac{K}{I}(x_1 - x_3)$$
(5.71)

with the regions defined as

$$\mathcal{P}_{21} = \left\{ \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} \middle| -\pi < x_1 < -\frac{2\pi}{7}, x_2 \in \mathbb{R} \right\}$$
(5.72)

$$\mathcal{P}_{22} = \left\{ \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} \middle| -\frac{2\pi}{7} < x_1 < \frac{2\pi}{7}, x_2 \in \mathbb{R} \right\}$$
(5.73)

$$\mathcal{P}_{23} = \left\{ \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} \middle| \frac{2\pi}{7} < x_1 < \pi, x_2 \in \mathbb{R} \right\}$$
(5.74)

Each of the following polynomials, using (5.47), describes a set that contains the common boundaries of the corresponding regions

$$E_{212}(x_1, x_2) = x_1 + \frac{2\pi}{7}$$
(5.75)

$$E_{223}(x_1, x_2) = x_1 - \frac{2\pi}{7} \tag{5.76}$$

Considering the Lyapunov function  $V_2(x_1, x_2) = \frac{1}{2}x_1^2 + \frac{1}{2}(x_2 + 2x_1)^2$  and solving the SOS feasibility problem (5.46) for the PWP system (5.71), the following controller is computed

$$x_{3} = \gamma_{2}(x_{1}, x_{2}) = \begin{cases} 0.26137 + 0.85161x_{1} - 0.1x_{2} & \begin{bmatrix} x_{1} \\ x_{2} \end{bmatrix} \in \mathcal{P}_{21} \\ 0.56043x_{1} - 0.1x_{2} & \begin{bmatrix} x_{1} \\ x_{2} \end{bmatrix} \in \mathcal{P}_{22} \\ -0.26137 + 0.85161x_{1} - 0.1x_{2} & \begin{bmatrix} x_{1} \\ x_{2} \end{bmatrix} \in \mathcal{P}_{23} \end{cases}$$
(5.77)

For the next step, the following PWP system is considered

$$\dot{x}_1 = x_2$$
  

$$\dot{x}_2 = -\frac{MgL}{I}\hat{f}_1(x_1) - \frac{K}{I}(x_1 - x_3)$$
  

$$\dot{x}_3 = x_4$$
  
(5.78)

with the following regions

$$\mathcal{P}_{31} = \left\{ \left[ \begin{array}{c} x_1 \\ x_2 \\ x_3 \end{array} \right] \middle| -\pi < x_1 < -\frac{2\pi}{7}, x_2 \in \mathbb{R}, x_3 \in \mathbb{R} \right\}$$
(5.79)

$$\mathcal{P}_{32} = \left\{ \left[ \begin{bmatrix} x_1 \\ x_2 \\ x_3 \end{bmatrix} \middle| -\frac{2\pi}{7} < x_1 < \frac{2\pi}{7}, x_2 \in \mathbb{R}, x_3 \in \mathbb{R} \right\}$$
(5.80)

$$\mathcal{P}_{33} = \left\{ \left[ \begin{array}{c} x_1 \\ x_2 \\ x_3 \end{array} \right] \middle| \frac{2\pi}{7} < x_1 < \pi, x_2 \in \mathbb{R}, x_3 \in \mathbb{R} \right\}$$
(5.81)

and the following polynomials for the common boundaries

$$E_{312}(x_1, x_2, x_3) = x_1 + \frac{2\pi}{7}$$
(5.82)

$$E_{323}(x_1, x_2, x_3) = x_1 - \frac{2\pi}{7}$$
(5.83)

Considering the Lyapunov function  $V_3(x_1, x_2, x_3) = \frac{1}{2}x_1^2 + \frac{1}{2}(x_2 + 2x_1)^2 + \frac{1}{2}(x_3 - \gamma_2(x_1, x_2))^2$  and solving the corresponding SOS feasibility problem for the PWP system (5.78), the following controller is computed

$$x_4 = \gamma_3(x_1, x_2, x_3) = \begin{cases} 2.1731 - 77.5789x_1 - 75.8241x_2 - 80x_3 & \begin{bmatrix} x_1 \\ x_2 \\ x_3 \end{bmatrix} \in \mathcal{P}_{31} \\ -80x_1 - 75.8241x_2 - 80x_3 & \begin{bmatrix} x_1 \\ x_2 \\ x_3 \end{bmatrix} \in \mathcal{P}_{32} \end{cases}$$

$$\left(\begin{array}{c} -2.1731 - 77.5789x_1 - 75.8241x_2 - 80x_3 \quad \begin{bmatrix} x_1\\ x_2\\ x_3 \end{bmatrix} \in \mathcal{P}_{33} \\ (5.84) \end{array}\right)$$

For the next step, the following PWP system is considered

$$\dot{x}_{1} = x_{2}$$

$$\dot{x}_{2} = -\frac{MgL}{I}\hat{f}_{1}(x_{1}) - \frac{K}{I}(x_{1} - x_{3})$$

$$\dot{x}_{3} = x_{4}$$

$$\dot{x}_{4} = -\frac{\hat{f}_{2}(x_{4})}{J} + \frac{K}{J}(x_{1} - x_{3}) + \frac{1}{J}u$$
(5.85)

with the following regions

$$\mathcal{P}_{41} = \left\{ \begin{bmatrix} x_1 \\ x_2 \\ x_3 \\ x_4 \end{bmatrix} \middle| -\pi < x_1 < -\frac{2\pi}{7}, x_2 \in \mathbb{R}, x_3 \in \mathbb{R}, x_4 > 0 \right\}$$
(5.86)

$$\mathcal{P}_{42} = \left\{ \begin{bmatrix} x_1 \\ x_2 \\ x_3 \\ x_4 \end{bmatrix} \middle| -\frac{2\pi}{7} < x_1 < \frac{2\pi}{7}, x_2 \in \mathbb{R}, x_3 \in \mathbb{R}, x_4 > 0 \right\}$$
(5.87)

$$\mathcal{P}_{43} = \left\{ \left[ \begin{array}{c} x_1 \\ x_2 \\ x_3 \\ x_4 \end{array} \right] \left| \begin{array}{c} 2\pi \\ 7 \end{array} < x_1 < \pi, x_2 \in \mathbb{R}, x_3 \in \mathbb{R}, x_4 > 0 \right\}$$
(5.88)

$$\mathcal{P}_{44} = \left\{ \left[ \begin{array}{c} x_1 \\ x_2 \\ x_3 \\ x_4 \\ x_4 \\ \end{array} \right] \middle| -\pi < x_1 < -\frac{2\pi}{7}, x_2 \in \mathbb{R}, x_3 \in \mathbb{R}, x_4 < 0 \right\}$$
(5.89)

$$\mathcal{P}_{45} = \left\{ \begin{bmatrix} x_1 \\ x_2 \\ x_3 \\ x_4 \end{bmatrix} \middle| -\frac{2\pi}{7} < x_1 < \frac{2\pi}{7}, x_2 \in \mathbb{R}, x_3 \in \mathbb{R}, x_4 < 0 \right\}$$
(5.90)

$$\mathcal{P}_{46} = \left\{ \left[ \begin{array}{c} x_1 \\ x_2 \\ x_3 \\ x_4 \end{array} \right] \middle| \frac{2\pi}{7} < x_1 < \pi, x_2 \in \mathbb{R}, x_3 \in \mathbb{R}, x_4 < 0 \right\}$$
(5.91)

and the following polynomials for the common boundaries

$$E_{412}(x_1, x_2, x_3, x_4) = x_1 + \frac{2\pi}{7}$$
(5.92)

$$E_{414}(x_1, x_2, x_3, x_4) = x_4 \tag{5.93}$$

$$E_{423}(x_1, x_2, x_3, x_4) = x_1 - \frac{2\pi}{7}$$
(5.94)

$$E_{425}(x_1, x_2, x_3, x_4) = x_4 \tag{5.95}$$

$$E_{436}(x_1, x_2, x_3, x_4) = x_4 \tag{5.96}$$

$$E_{445}(x_1, x_2, x_3, x_4) = x_1 + \frac{2\pi}{7}$$
(5.97)

$$E_{456}(x_1, x_2, x_3, x_4) = x_1 - \frac{2\pi}{7}$$
(5.98)

The structure of the regions of the PWP system (5.85) is shown in Figure 5.2. This



Figure 5.6: The structure of the regions of the PWP system (5.85)

figure shows that, for example, the image of the region  $\mathcal{P}_{42}$  on the  $(x_1, x_2)$  space is a subset of  $\mathcal{P}_{22}$ .

Considering the Lyapunov function  $V_4(x_1, x_2, x_3, x_4) = \frac{1}{2}x_1^2 + \frac{1}{2}(x_2 + 2x_1)^2 + \frac{1}{2}(x_3 - \gamma_2(x_1, x_2))^2 + \frac{1}{2}(x_4 - \gamma_3(x_1, x_2, x_3))^2$  and solving the SOS feasibility problem (5.49) for the PWP system (5.85), the following PWA controller is computed

$$u = \begin{cases} 737.0 - 9786x_1 - 11040x_2 - 1311x_3 - 162.7x_4 & \begin{bmatrix} x_1 & x_2 & x_3 & x_4 \end{bmatrix}^T \in \mathcal{P}_{41} \\ -10610x_1 - 11040x_2 - 13110x_3 - 162.7x_4 & \begin{bmatrix} x_1 & x_2 & x_3 & x_4 \end{bmatrix}^T \in \mathcal{P}_{42} \\ -736.2 - 9786x_1 - 11040x_2 - 13110x_3 - 162.7x_4 & \begin{bmatrix} x_1 & x_2 & x_3 & x_4 \end{bmatrix}^T \in \mathcal{P}_{43} \\ 736.2 - 9786x_1 - 11040x_2 - 13110x_3 - 162.5x_4 & \begin{bmatrix} x_1 & x_2 & x_3 & x_4 \end{bmatrix}^T \in \mathcal{P}_{44} \\ -10610x_1 - 11040x_2 - 13110x_3 - 162.5x_4 & \begin{bmatrix} x_1 & x_2 & x_3 & x_4 \end{bmatrix}^T \in \mathcal{P}_{45} \\ -737.0 - 9786x_1 - 11040x_2 - 13110x_3 - 162.5x_4 & \begin{bmatrix} x_1 & x_2 & x_3 & x_4 \end{bmatrix}^T \in \mathcal{P}_{45} \\ (5.99) \end{cases}$$



PWA controller (5.99) with the initial condition  $x_0 = [\pi \ 0 \ 0.8\pi \ 0]^T$ . The system converges to the origin in 4 seconds. However, by examining the transient changes of the control input in Fig. 5.8, we realize that the input is very large. Another possible controller is a PWP controller. Next, a PWP controller of third order in  $x_1$ , first order in  $x_2$ , first order in  $x_3$  and third order in  $x_4$  is designed. Considering the Lyapunov function  $V_4(x_1, x_2, x_3, x_4) = \frac{1}{2}x_1^2 + \frac{1}{2}(x_2 + 2x_1)^2 + \frac{1}{2}(x_3 - \gamma_2(x_1, x_2))^2 + \frac{1}{2}(x_4 - \gamma_3(x_1, x_2, x_3))^2$  and solving the SOS feasibility problem (5.49) for the PWP system (5.85), the following PWP controller is computed

$$u = 11.46 + 832.0x_{1} - 44.25x_{2} - 169.9x_{1}^{2} + 23.74x_{1}x_{2} - 10.66x_{1}^{3} - 10.42x_{1}^{2}x_{2}$$
  
- 1490x<sub>3</sub> + 24.93x<sub>1</sub>x<sub>3</sub> - 10.94x\_{1}^{2}x\_{3} - 17.53x\_{4} - 0.08227x\_{1}x\_{4} - 0.3943x\_{2}x\_{4}  
- 0.4231x<sub>3</sub>x<sub>4</sub> + 0.01364x\_{4}^{2} + 0.1917x\_{1}^{2}x\_{4} + 0.3465x\_{1}x\_{2}x\_{4} + 0.3670x\_{1}x\_{3}x\_{4}  
- 0.008023x<sub>1</sub>x<sub>4</sub><sup>2</sup> - 0.01344x<sub>2</sub>x<sub>4</sub><sup>2</sup> - 0.01427x\_{3}x\_{4}^{2} - 0.001407x\_{4}^{3}  
for  $\begin{bmatrix} x_{1} & x_{2} & x_{3} & x_{4} \end{bmatrix}^{T} \in \mathcal{P}_{41}$   
(5.100)

$$u = 966.8x_{1} - 70.87x_{2} - 0.0071x_{1}^{2} - 0.009167x_{1}x_{2} - 4.058x_{1}^{3} - 3.835x_{1}^{2}x_{2} - 1518x_{3}$$
  
- 0.01287x\_{1}x\_{3} - 4.11x\_{1}^{2}x\_{3} - 17.92x\_{4} - 0.6414x\_{1}x\_{4} - 0.6152x\_{2}x\_{4} - 0.6545x\_{3}x\_{4}  
+ 0.009x\_{4}^{2} + 0.06297x\_{1}^{2}x\_{4} + 0.1x\_{1}x\_{2}x\_{4} + 0.11x\_{1}x\_{3}x\_{4} - 0.01316x\_{1}x\_{4}^{2}  
- 0.01344x\_{2}x\_{4}^{2} - 0.01427x\_{3}x\_{4}^{2} - 0.001407x\_{4}^{3}  
for  $\begin{bmatrix} x_{1} & x_{2} & x_{3} & x_{4} \end{bmatrix}^{T} \in \mathcal{P}_{42}$   
(5.101)



Figure 5.7: State variables of the nonlinear model - PWA controller



Figure 5.8: Control input - PWA controller

$$u = -10.69 + 832.0x_{1} - 44.25x_{2} + 169.9x_{1}^{2} - 23.75x_{1}x_{2} - 10.66x_{1}^{3} - 10.42x_{1}^{2}x_{2}$$
  

$$-1490x_{3} - 24.95x_{1}x_{3} - 10.95x_{1}^{2}x_{3} - 17.54x_{4} - 0.63x_{1}x_{4} - 0.2946x_{2}x_{4}$$
  

$$-0.3062x_{3} * x_{4} + 0.01371x_{4}^{2} - 0.4293x_{1}^{2}x_{4} - 0.2568x_{1}x_{2}x_{4} - 0.279x_{1}x_{3}x_{4}$$
  

$$-0.0184x_{1}x_{4}^{2} - 0.01344x_{2}x_{4}^{2} - 0.01427x_{3}x_{4}^{2} - 0.001407x_{4}^{3}$$
  

$$for \begin{bmatrix} x_{1} & x_{2} & x_{3} & x_{4} \end{bmatrix}^{T} \in \mathcal{P}_{43}$$
  
(5.102)

$$u = 10.69 + 832x_1 - 44.25x_2 - 169.9x_1^2 + 23.74x_1x_2 - 10.66x_1^3 - 10.42x_1^2x_2 - 1490x_3 + 24.93x_1x_3 - 10.94x_1^2x_3 - 17.36x_4 + 0.634x_1x_4 + 0.2984x_2x_4 + 0.31x_3x_4 - 0.01362x_4^2 - 0.4328x_1^2x_4 - 0.26x_1x_2x_4 - 0.2825x_1x_3x_4 - 0.01852x_1x_4^2 - 0.01349x_2x_4^2 - 0.01431x_3x_4^2 + 0.001045x_4^3 for  $\begin{bmatrix} x_1 & x_2 & x_3 & x_4 \end{bmatrix}^T \in \mathcal{P}_{44}$  (5.103)$$

$$u = 966.8x_{1} - 70.87x_{2} - 0.0071x_{1}^{2} - 0.009167x_{1}x_{2} - 4.058x_{1}^{3} - 3.8347x_{1}^{2}x_{2}$$
  
- 1518x<sub>3</sub> - 0.01287x<sub>1</sub>x<sub>3</sub> - 4.11x\_{1}^{2}x\_{3} - 17.74x\_{4} + 0.6477x\_{1}x\_{4} + 0.6215x\_{2}x\_{4}  
+ 0.6612x<sub>3</sub>x<sub>4</sub> - 0.008877x\_{4}^{2} + 0.0624x\_{1}^{2}x\_{4} + 0.09985x\_{1}x\_{2}x\_{4} + 0.1086x\_{1}x\_{3}x\_{4}  
- 0.01324x<sub>1</sub>x\_{4}^{2} - 0.01349x\_{2}x\_{4}^{2} - 0.01431x\_{3}x\_{4}^{2} + 0.001045x\_{4}^{3}  
for  $\begin{bmatrix} x_{1} & x_{2} & x_{3} & x_{4} \end{bmatrix}^{T} \in \mathcal{P}_{45}$   
(5.104)

$$u = -11.47 + 832x_1 - 44.25x_2 + 169.9x_1^2 - 23.75x_1x_2 - 10.66x_1^3 - 10.42x_1^2x_2$$
  
- 1490x<sub>3</sub> - 24.95x<sub>1</sub>x<sub>3</sub> - 10.95x\_1^2x\_3 - 17.35x\_4 + 0.08544x\_1x\_4 + 0.3976x\_2x\_4  
+ 0.4266x\_3x\_4 - 0.01358x\_4^2 + 0.1946x\_1^2x\_4 + 0.3493x\_1x\_2x\_4 + 0.37x\_1x\_3x\_4  
- 0.008x\_1x\_4^2 - 0.01349x\_2x\_4^2 - 0.01431x\_3x\_4^2 + 0.001x\_4^3  
for  $\begin{bmatrix} x_1 & x_2 & x_3 & x_4 \end{bmatrix}^T \in \mathcal{P}_{46}$ 

(5.105)
Fig. 5.9 shows the states of the nonlinear system in feedback connection with the PWP controller with the initial condition  $x_0 = [\pi \ 0 \ 0.8\pi \ 0]^T$ . The system converges to the origin a bit slower in comparison to the case of the PWA controller. However, Fig. 5.10 shows that in this case the control input is much smaller.

#### 5.5 Conclusions

In this chapter, the strict feedback form for PWP systems was introduced. Using backstepping, controller synthesis for this large class of PWP systems was formulated as an SOS program, which is a convex optimization problem. The synthesis problem was addressed in two cases: SOS Lyapunov functions for PWP systems with discontinuous vector fields and PWP Lyapunov functions for PWP systems with continuous vector fields. One of the main advantages of the proposed method is that it addresses PWP systems with discontinuous vector fields regardless of possible attractive sliding modes.





Figure 5.9: State variables of the nonlinear model - PWP controller

Figure 5.10: Control input - PWA controller

### Chapter 6

# Sampled-Data Piecewise Affine Systems: A Time-Delay Approach

This chapter addresses stability analysis of sampled-data piecewise-affine (PWA) systems consisting of a continuous-time plant in feedback connection with a discretetime emulation of a continuous-time state feedback controller. The sampled-data system is considered as a continuous-time system with a variable delay. Conditions under which the trajectories of the sampled-data closed-loop system will converge to an attracting invariant set are then presented. It is also shown that when the sampling period converges to zero, these conditions coincide with sufficient conditions for non-fragility of the stabilizing continuous-time PWA state feedback controller.

#### 6.1 Introduction

The research work on continuous-time PWA systems has concentrated on Lyapunovbased controller synthesis methods [56, 66, 96, 101, 103]. However, none of these approaches would be applicable directly to controller synthesis for computer-controlled or sampled-data PWA systems. This is the scenario mostly encountered in applications given the flexibility of control implementation in a microprocessor.

Although linear sampled-data systems are a well-studied matter [27], controller emulation for systems with possible discontinuities at the switching, such as sampled-data PWA systems, has not had many research contributions. In fact, only recently these systems have started to be addressed in the literature in references such as [6, 62, 63, 99, 122, 123, 129]. The approach by [123] established that, under certain conditions, the controllable subspaces of a continuous-time switched linear system and its discrete-time counterpart are the same. Canonical forms of switched linear systems based on controllability are presented in [122]. The reference [129] considers stability analysis of switched systems that can switch between a set of continuous-time plants and a set of discrete-time plants but does not handle sampled-data systems involving a cascade of a discrete-time system between a sample-and-hold and a continuous-time system. Furthermore, it does not address controller design. The approach by [6, 62, 63] was probably the first where the term "sampled-data PWA systems" is used, although the systems described in this work do not posses the typical structure of a continuous-time plant being controlled by a discrete-time controller. The problem addressed in [6, 62, 63] is one where the controller is continuous-time and the switching events are the ones controlled by the system logic inside a computer. In other words, in these systems it is assumed that the designer has command over the switching times of the system, which does not occur often in practice. For this class of systems, reference [6] presents a probabilistic analysis of controllability. The preliminary study of [62, 63] is interesting as it highlights important limitations of current discrete-time PWA control methodologies when applied to the control of a physical continuous-time system. As mentioned in [62] unexpected phenomena such as chattering can occur, depending on the switching times. This increases the interest in studying computer implementations of controllers designed in continuous-time. Reference [99] addresses the classical structure of a sampled-data system whereby the system is continuoustime and the controller is being implemented (emulated) in discrete-time inside a computer. However, the sampling time must be constant.

This chapter departs considerably from previous research by addressing stability analysis of sampled-data PWA systems using a time delay approach. In fact, the discrete-time PWA controller is seen as a continuous-time PWA controller with a delay that varies with time. Using a Lyapunov-Krasovskii functional, LMI conditions are derived as sufficient conditions for convergence of the sampled-data PWA system trajectories to an attracting invariant set. One of the advantages of the proposed method is that it can be applied to sampled-data PWA systems with variable sampling time as opposed to [99] that deals with a constant sample time. A very important and interesting property of the LMI conditions proposed in this chapter is that when the sampling time converges to zero, these conditions coincide with LMI conditions for non-fragility of the continuous-time PWA controller. Therefore, to implement a continuous-time PWA controller in discrete-time, it is required that the controller be robust to variations in the controller parameters. This in itself is a very interesting result.

The chapter starts by the stability analysis of the sampled-data system when a continuous-time controller is emulated in discrete-time. A numerical example is included to show the performance of the proposed method. Finally, the chapter closes by stating the conclusions.

#### 6.2 Stability of Sampled-Data PWA Systems

Consider a PWA controller of the following form

$$u(t) = K_i x(t) + k_i, \ x(t) \in \mathcal{R}_i$$
(6.1)

for the PWA system

$$\dot{x} = A_i x + a_i + Bu, \quad \text{for } x \in \mathcal{R}_i \tag{6.2}$$

with the region  $\mathcal{R}_i$  defined as

$$\mathcal{R}_i = \{ x | E_i x + e_i \succ 0 \}, \tag{6.3}$$

The closed-loop system is assumed to be asymptotically stable. It is also assumed that the vector field of the open loop PWA system (3.10) with u(t) = 0 is continuous across the boundaries of two or more regions and  $a_i = 0$  for  $i \in \mathcal{I}(0)$ .

If the PWA controller (6.1) is implemented as a digital controller and is connected to the PWA system (6.2) through a sample-and-hold, the closed-loop system can be described by

$$\dot{x}(t) = A_i x(t) + a_i + B(K_j x(t_k) + k_j), \tag{6.4}$$

for  $x(t) \in \mathcal{R}_i$  and  $x(t_k) \in \mathcal{R}_j$  where  $t_k$  for  $k \in \mathbb{N}$  is the sampling time and  $t_k \leq t < t_{k+1}$ . The closed-loop system (6.4) can be rewritten as

$$\dot{x}(t) = A_i x(t) + a_i + B(K_i x(t_k) + k_i) + Bw,$$
(6.5)

for  $x(t) \in \mathcal{R}_i$  and  $x(t_k) \in \mathcal{R}_j$  where

$$w(t) = (K_j - K_i)x(t_k) + (k_j - k_i), \ x(t) \in \mathcal{R}_i, \ x(t_k) \in \mathcal{R}_j$$
(6.6)

The input w(t) is a result of the fact that x(t) and  $x(t_k)$  are not necessarily in the same region.

Following [82], the time elapsed since the last sampling time will be denoted by

$$\rho(t) := t - t_k, \ t_k \le t < t_{k+1} \tag{6.7}$$

and  $\tau_M(\tau_D)$  is defined as the maximum (minimum) interval between sampling times.

 $\tau_D \le t_{k+1} - t_k \le \tau_M, \forall k \in \mathbb{N}$ (6.8)

Consider a Lyapunov-Krasovskii functional of the form

$$V(x_s, \rho) := V_1(x) + V_2(x_s) + V_3(x_s, \rho)$$
(6.9)

where

$$x_s(t) := \begin{bmatrix} x(t) \\ x(t_k) \end{bmatrix}, t_k \le t < t_{k+1}$$

$$V_{1}(x) := x^{T} P x$$
  

$$V_{2}(x_{s}) := \int_{-\tau_{M}}^{0} \int_{t+r}^{t} \dot{x}^{T}(s) R \dot{x}(s) ds dr$$
  

$$V_{3}(x_{s}, \rho) := (\tau_{M} - \rho) (x(t) - x(t_{k}))^{T} X(x(t) - x(t_{k}))$$

and P, R and X are positive definite matrices. Therefore, the Lyapunov-Krasovskii functional  $V(x_s, \rho)$  is positive definite. At the sampling times,  $V(x_s, \rho)$  does not increase because  $V_2(x_s)$  and  $V_3(x_s, \rho)$  are non-negative right before each sampling time and they become zero right after the sampling time [82]. It can be shown that  $V(x_s, \rho)$  satisfies the following inequality

$$\lambda_{\min}(P) \|x\|^2 \le V(x_s, \rho) \le \sigma_a \|x_s\|^2 + \sigma_b$$
(6.10)

where

$$\sigma_a = \lambda_{\max}(P) + 2(\tau_M - \rho)\lambda_{\max}(X) + \frac{\tau_M^2}{2}\lambda_{\max}(\bar{R}),$$
  
$$\sigma_b = \frac{\tau_M^2}{2}\lambda_{\max}(\bar{R}),$$

where  $\lambda_{\min}(.)$  and  $\lambda_{\max}(.)$  mean the minimum and maximum eigenvalues of a matrix, respectively, and

$$\bar{R} = \arg\max_{i,j} \lambda_{\max}(\bar{R}_{ij}) \tag{6.11}$$

$$\bar{R}_{ij} = \begin{bmatrix} A_i^{\mathrm{T}} \\ K_j^{\mathrm{T}} B^{\mathrm{T}} \\ a_i^{\mathrm{T}} + k_j^{\mathrm{T}} B^{\mathrm{T}} \end{bmatrix} R \begin{bmatrix} A_i & BK_j & a_i + Bk_j \end{bmatrix}$$
(6.12)

The main result of this chapter is now presented.

**Theorem 6.1.** For the sampled-data PWA system (6.5), assume there exist symmetric positive matrices P, R, X and matrices  $N_i$  for i = 1, ..., M such that

• for all  $i \in \mathcal{I}(0)$ ,

$$\Omega_i + \tau_M M_{1i} + \tau_M M_{2i} < 0 \tag{6.13}$$

$$\begin{bmatrix} \Omega_{i} + \tau_{M} M_{1i} & \tau_{M} \begin{bmatrix} N_{i} \\ 0 \end{bmatrix} \\ \tau_{M} \begin{bmatrix} N_{i}^{T} & 0 \end{bmatrix} - \tau_{M} R \end{bmatrix} < 0$$
(6.14)

• for all  $i \notin \mathcal{I}(0)$ ,  $\bar{\Lambda}_i \succ 0$ ,

$$\overline{\Omega}_{i} + \tau_{M} \overline{M}_{1i} + \tau_{M} \overline{M}_{2i} < 0 \qquad (6.15)$$

$$\overline{\Omega}_{i} + \tau_{M} \overline{M}_{1i} \qquad \tau_{M} \begin{bmatrix} N_{i} \\ 0 \\ 0 \end{bmatrix} \\ \tau_{M} \begin{bmatrix} N_{i}^{T} & 0 & 0 \end{bmatrix} - \tau_{M} R \qquad (6.16)$$

where

$$\begin{split} \Omega_{i} &= \begin{bmatrix} \Psi_{i} & \begin{bmatrix} P \\ 0 \end{bmatrix} B \\ B^{T} \begin{bmatrix} P & 0 \end{bmatrix} & -\gamma I \end{bmatrix}, \\ \Psi_{i} &= \begin{bmatrix} P \\ -0 \end{bmatrix} \begin{bmatrix} A_{i} & BK_{i} \end{bmatrix} + \begin{bmatrix} A_{i}^{T} \\ K_{i}^{T}B^{T} \end{bmatrix} \begin{bmatrix} P & 0 \end{bmatrix} \\ &- \begin{bmatrix} I \\ -I \end{bmatrix} X \begin{bmatrix} I & -I \end{bmatrix} \\ &- N_{i} \begin{bmatrix} I - I \end{bmatrix} - \begin{bmatrix} I \\ -I \end{bmatrix} N_{i}^{T} + \eta I_{2n \times 2n}, \end{split}$$

$$M_{1i} = \begin{bmatrix} A_i^T \\ K_i^T B^T \\ B^T \end{bmatrix} R \begin{bmatrix} A_i & BK_i & B \end{bmatrix},$$
  

$$M_{2i} = \begin{bmatrix} I \\ -I \\ 0 \end{bmatrix} X \begin{bmatrix} A_i & BK_i & B \end{bmatrix} + \begin{bmatrix} A_i^T \\ K_i^T B^T \\ B^T \end{bmatrix} X \begin{bmatrix} I & -I & 0 \end{bmatrix},$$

$$\begin{split} \overline{\Omega}_{i} &= \left[ \begin{array}{ccc} \overline{\Psi}_{i} & \left[ \begin{array}{c} P \\ 0 \\ 0 \end{array} \right] B \\ B^{T} \left[ \begin{array}{c} P & 0 & 0 \end{array} \right] & -\gamma I \end{array} \right], \\ \overline{\Psi}_{i} &= \left[ \begin{array}{c} P \\ 0 \\ 0 \end{array} \right] \left[ \begin{array}{c} A_{i} & BK_{i} & Bk_{i} + a_{i} \end{array} \right] \\ &+ \left[ \begin{array}{c} A_{i}^{T} \\ K_{i}^{T}B^{T} \\ k_{i}^{T}B^{T} + a_{i}^{T} \end{array} \right] \left[ \begin{array}{c} P & 0 & 0 \end{array} \right] - \left[ \begin{array}{c} I \\ -I \\ 0 \end{array} \right] X \left[ \begin{array}{c} I & -I & 0 \end{array} \right] \\ &- \left[ \begin{array}{c} N_{i} \\ 0 \end{array} \right] \left[ \begin{array}{c} I - I & 0 \end{array} \right] - \left[ \begin{array}{c} I \\ -I \\ 0 \end{array} \right] \left[ \begin{array}{c} N_{i}^{T} & 0 \end{array} \right] \\ &+ \left[ \begin{array}{c} \eta I_{2n \times 2n} & 0 \\ 0 & 0 \end{array} \right] + \left[ \begin{array}{c} E_{i}^{T} & 0 \\ 0 & 0 \\ e_{i}^{T} & 1 \end{array} \right] \overline{\Lambda}_{i} \left[ \begin{array}{c} E_{i} & 0 & e_{i} \\ 0 & 0 & 1 \end{array} \right], \end{split}$$

$$\overline{M}_{1i} = \begin{bmatrix} A_i^T \\ K_i^T B^T \\ k_i^T B^T + a_i^T \\ B^T \end{bmatrix} R \begin{bmatrix} A_i & BK_i & Bk_i + a_i & B \end{bmatrix},$$

$$\overline{M}_{2i} = \begin{bmatrix} I \\ -I \\ 0 \\ 0 \end{bmatrix} X \begin{bmatrix} A_i & BK_i & Bk_i + a_i & B \end{bmatrix} + \begin{bmatrix} A_i^T \\ K_i^T B^T \\ k_i^T B^T + a_i^T \\ B^T \end{bmatrix} X \begin{bmatrix} I & -I & 0 & 0 \end{bmatrix}$$

Let there be constants  $\Delta_K$  and  $\Delta_k$  such that

$$\|w\| \le \Delta_K \|x(t_k)\| + \Delta_k \tag{6.17}$$

Define

$$\mu_{\theta} = \frac{\sqrt{\gamma}\Delta_k}{\sqrt{\theta\eta} - \sqrt{\gamma}\Delta_K} \tag{6.18}$$

and the region

$$\Phi_{\theta} = \{ x_s | \ \| x_s \| \le \mu_{\theta} \}$$
(6.19)

for some positive constant  $\theta < 1$  that verifies

$$\Delta_K < \sqrt{\frac{\theta \eta}{\gamma}} \tag{6.20}$$

Then, all the trajectories of the system (6.5) in  $\mathcal{X}$  converge to the following invariant set

$$\Omega = \{ x_s | V(x_s, \rho) \le \sigma_a \mu_\theta^2 + \sigma_b \}$$
(6.21)

*Proof.* The proof is divided into two parts.

1. First, it is shown that the inequalities (6.13), (6.14), (6.15) and (6.16) are sufficient conditions for the following inequality to hold

$$\dot{V}(x_s,\rho) \le -\eta x_s^{\mathrm{T}} x_s + \gamma w^{\mathrm{T}} w \tag{6.22}$$

for  $t_k < t < t_{k+1}$ . Since  $V_1(x) = x^{\mathrm{T}} P x$ , one has

$$\dot{V}_1(x) = \dot{x}^{\mathrm{T}} P x + x^{\mathrm{T}} P \dot{x}$$
(6.23)

 $V_2(x_s)$  can be written in the following form

$$V_2(x_s) = \int_{-\tau_M}^0 g(t, r) dr$$
 (6.24)

where

$$g(t,r) = \int_{t+r}^{t} \dot{x}^{\mathrm{T}}(s) R \dot{x}(s) ds \qquad (6.25)$$

Thus, since  $\dot{\rho} = 1$  for  $t_k < t < t_{k+1}$ ,

$$\dot{V}_2(x_s) = \int_{-\tau_M}^0 \frac{\partial}{\partial t} g(t, r) dr$$
(6.26)

The expression

$$\frac{\partial}{\partial t}g(t,r) = \dot{x}^{\mathrm{T}}(t)R\dot{x}(t) - \dot{x}^{\mathrm{T}}(t+r)R\dot{x}(t+r)$$
(6.27)

then yields

$$\dot{V}_2(x_s) = \tau_M \dot{x}^{\mathrm{T}}(t) R \dot{x}(t) - \int_{t-\tau_M}^t \dot{x}^{\mathrm{T}}(s) R \dot{x}(s) ds$$
 (6.28)

From (6.8) one has  $\rho \leq \tau_M$  and considering the fact that R is positive definite, this leads to

$$\dot{V}_2(x_s) \le \tau_M \dot{x}^{\mathrm{T}}(t) R \dot{x}(t) - \int_{t-\rho}^t \dot{x}^{\mathrm{T}}(s) R \dot{x}(s) ds$$
(6.29)

Since R is positive definite, for any matrix  $N_i \in \mathbb{R}^{n \times 2n}$  one has

$$\begin{bmatrix} \dot{x}^{\mathrm{T}}(s) & x_{s}^{\mathrm{T}}(t)N_{i} \end{bmatrix} \begin{bmatrix} R & -I \\ -I & R^{-1} \end{bmatrix} \begin{bmatrix} \dot{x}(s) \\ N_{i}^{\mathrm{T}}x_{s}(t) \end{bmatrix} \ge 0$$
(6.30)

and therefore

$$-\dot{x}(s)^{\mathrm{T}}R\dot{x}(s) \le x_{s}^{\mathrm{T}}(t)N_{i}R^{-1}N_{i}^{\mathrm{T}}x_{s}(t) - 2x_{s}^{\mathrm{T}}(t)N_{i}\dot{x}(s)$$
(6.31)

Integrating both sides from  $t - \rho$  to t and using (6.7) yields,

$$-\int_{t-\rho}^{t} \dot{x}(s)^{\mathrm{T}} R \dot{x}(s) ds \leq \rho x_{s}^{\mathrm{T}}(t) N_{i} R^{-1} N_{i}^{\mathrm{T}} x_{s}(t) -2 x_{s}^{\mathrm{T}}(t) N_{i} \begin{bmatrix} I & -I \end{bmatrix} x_{s}(t)$$

$$(6.32)$$

It follows from (6.29) and (6.32) that

$$\dot{V}_{2}(x_{s}) \leq \tau_{M} \dot{x}^{\mathrm{T}} R \dot{x} + \rho x_{s}^{\mathrm{T}} N_{i} R^{-1} N_{i}^{\mathrm{T}} x_{s}$$
$$-2x_{s}^{\mathrm{T}} N_{i} \begin{bmatrix} I & -I \end{bmatrix} x_{s}$$
(6.33)

For  $V_3(x_s, \rho)$ , since  $\dot{\rho} = 1$  for  $t_k < t < t_{k+1}$ , one can write

$$\dot{V}_{3}(x_{s},\rho) = -(x(t) - x(t_{k}))^{\mathrm{T}}X(x(t) - x(t_{k})) + 2(\tau_{M} - \rho)(x(t) - x(t_{k}))^{\mathrm{T}}X\dot{x}(t)$$
(6.34)

From (6.23), (6.33) and (6.34), it follows that a sufficient condition for (6.22) is the following inequality

$$\dot{x}^{\mathrm{T}} P x + x^{\mathrm{T}} P \dot{x} + \tau_{M} \dot{x}^{\mathrm{T}} R \dot{x} + \rho x_{s}^{\mathrm{T}} N_{i} R^{-1} N_{i}^{\mathrm{T}} x_{s}$$

$$-2 x_{s}^{\mathrm{T}} N_{i} \begin{bmatrix} I & -I \end{bmatrix} x_{s}$$

$$-x_{s}^{\mathrm{T}} \begin{bmatrix} I \\ -I \end{bmatrix} X \begin{bmatrix} I & -I \end{bmatrix} x_{s}$$

$$+2 (\tau_{M} - \rho) x_{s}^{\mathrm{T}} \begin{bmatrix} I \\ -I \end{bmatrix} X \dot{x}$$

$$+\eta x_{s}^{\mathrm{T}} x_{s} - \gamma w^{\mathrm{T}} w \leq 0 \qquad (6.35)$$

For  $i \in \mathcal{I}(0)$ , one has

$$\dot{x} = \begin{bmatrix} A_i & BK_i \end{bmatrix} x_s + Bw, \tag{6.36}$$

for  $x(t) \in \mathcal{R}_i$  and  $x(t_k) \in \mathcal{R}_j$ . Replacing  $\dot{x}$  from (6.36) into (6.35) yields

$$x_{s}^{\mathrm{T}}\left(\left[\begin{array}{c}P\\0\end{array}\right]\left[\begin{array}{c}A_{i} \quad BK_{i}\end{array}\right]+\left[\begin{array}{c}A_{i}^{\mathrm{T}}\\K_{i}^{\mathrm{T}}B^{\mathrm{T}}\end{array}\right]\left[\begin{array}{c}P \quad 0\end{array}\right]$$
$$+\tau_{M}\left[\begin{array}{c}A_{i}^{\mathrm{T}}\\K_{i}^{\mathrm{T}}B^{\mathrm{T}}\end{array}\right]R\left[\begin{array}{c}A_{i} \quad BK_{i}\end{array}\right]+\rho N_{i}R^{-1}N_{i}^{\mathrm{T}}$$
$$-N_{i}\left[\begin{array}{c}I & -I\end{array}\right]-\left[\begin{array}{c}I\\-I\end{array}\right]N_{i}^{\mathrm{T}}$$
$$-\left[\begin{array}{c}I\\-I\end{array}\right]X\left[\begin{array}{c}I & -I\end{array}\right]$$
$$+(\tau_{M}-\rho)\left[\begin{array}{c}I\\-I\end{array}\right]X\left[\begin{array}{c}A_{i} \quad BK_{i}\end{array}\right]$$

$$+(\tau_{M}-\rho)\begin{bmatrix}A_{i}^{\mathrm{T}}\\K_{i}^{\mathrm{T}}B^{\mathrm{T}}\end{bmatrix}X[I - I] + \eta I x_{s}$$

$$+x_{s}^{\mathrm{T}}\begin{bmatrix}P\\0\end{bmatrix}Bw + w^{\mathrm{T}}B^{\mathrm{T}}[P 0]x_{s}$$

$$+\tau_{M}x_{s}^{\mathrm{T}}\begin{bmatrix}A_{i}^{\mathrm{T}}\\K_{i}^{\mathrm{T}}B^{\mathrm{T}}\end{bmatrix}RBw + \tau_{M}w^{\mathrm{T}}B^{\mathrm{T}}R[A_{i} BK_{i}]x_{s}$$

$$+(\tau_{M}-\rho)x_{s}^{\mathrm{T}}\begin{bmatrix}I\\-I\end{bmatrix}XBw + (\tau_{M}-\rho)w^{\mathrm{T}}B^{\mathrm{T}}X[I - I]x_{s}$$

$$+\tau_{M}w^{\mathrm{T}}B^{\mathrm{T}}RBw - \gamma w^{\mathrm{T}}w < 0 \qquad (6.37)$$

Since (6.37) is affine in  $\rho$ , if it holds for  $\rho = 0$  and  $\rho = \tau_M$  then it is satisfied for any  $\rho \in [0, \tau_M]$ . For  $\rho = 0$ , the inequality (6.37) can be written as (6.13). Using Schur complement for  $\rho = \tau_M$ , the inequality (6.37) can be converted to (6.14). For  $i \notin \mathcal{I}(0)$ , one has

$$\dot{x} = \left[\begin{array}{cc} A_i & BK_i & a_i + Bk_i \end{array}\right] \bar{x}_s + Bw, \ x \in \mathcal{R}_i \tag{6.38}$$

where

$$\bar{x}_s = \begin{bmatrix} x_s \\ 1 \end{bmatrix} \tag{6.39}$$

It follows from (2.62) that

$$\begin{bmatrix} x^{\mathrm{T}} & 1 \end{bmatrix} \begin{bmatrix} E_i^{\mathrm{T}} & 0 \\ e_i^{\mathrm{T}} & 1 \end{bmatrix} \bar{\Lambda}_i \begin{bmatrix} E_i & e_i \\ 0 & 1 \end{bmatrix} \begin{bmatrix} x \\ 1 \end{bmatrix} > 0, \ x \in \mathcal{R}_i$$
(6.40)

where  $\bar{\Lambda}_i \succ 0$ . Using (6.38) and (6.40), a sufficient condition for (6.37) when  $x \in \mathcal{R}_i$  with  $i \notin \mathcal{I}(0)$  can be written as

$$\begin{split} \bar{x}_{s}^{\mathrm{T}} \left( \begin{bmatrix} P\\0\\0\\0 \end{bmatrix} \begin{bmatrix} A_{i} & BK_{i} & a_{i} + Bk_{i} \end{bmatrix} + \begin{bmatrix} A_{i}^{\mathrm{T}}\\K_{i}^{\mathrm{T}}B^{\mathrm{T}}\\a_{i}^{\mathrm{T}} + k_{i}^{\mathrm{T}}B^{\mathrm{T}} \end{bmatrix} \begin{bmatrix} P & 0 & 0 \end{bmatrix} \\ &+ \tau_{M} \begin{bmatrix} A_{i}^{\mathrm{T}}\\K_{i}^{\mathrm{T}}B^{\mathrm{T}}\\a_{i}^{\mathrm{T}} + k_{i}^{\mathrm{T}}B^{\mathrm{T}} \end{bmatrix} R \begin{bmatrix} A_{i} & BK_{i} & a_{i} + Bk_{i} \end{bmatrix} + \rho \begin{bmatrix} N_{i}\\0 \end{bmatrix} R^{-1} \begin{bmatrix} N_{i}^{\mathrm{T}} & 0 \end{bmatrix} \\ &- \begin{bmatrix} N_{i}\\0 \end{bmatrix} \begin{bmatrix} I & -I & 0 \end{bmatrix} - \begin{bmatrix} I\\-I\\0 \end{bmatrix} \begin{bmatrix} N_{i}^{\mathrm{T}} & 0 \end{bmatrix} - \begin{bmatrix} I\\-I\\0 \end{bmatrix} X \begin{bmatrix} I & -I & 0 \end{bmatrix} \\ &+ (\tau_{M} - \rho) \begin{bmatrix} I\\-I\\0 \end{bmatrix} X \begin{bmatrix} A_{i} & BK_{i} & a_{i} + Bk_{i} \end{bmatrix} \\ &+ (\tau_{M} - \rho) \begin{bmatrix} A_{i}^{\mathrm{T}}\\K_{i}^{\mathrm{T}}B^{\mathrm{T}}\\a_{i}^{\mathrm{T}} + k_{i}^{\mathrm{T}}B^{\mathrm{T}} \end{bmatrix} X \begin{bmatrix} I & -I & 0 \end{bmatrix} + \eta \begin{bmatrix} I & 0 & 0\\0 & I & 0\\0 & 0 & 0 \end{bmatrix} \right) \bar{x}_{s} \end{split}$$

$$+ \bar{x}_{s}^{\mathrm{T}} \begin{bmatrix} P \\ 0 \\ 0 \end{bmatrix} Bw + w^{\mathrm{T}}B^{\mathrm{T}} \begin{bmatrix} P & 0 & 0 \end{bmatrix} \bar{x}_{s}$$

$$+ \tau_{M} \bar{x}_{s}^{\mathrm{T}} \begin{bmatrix} A_{i}^{\mathrm{T}} \\ K_{i}^{\mathrm{T}}B^{\mathrm{T}} \\ k_{i}^{\mathrm{T}}B^{\mathrm{T}} + a_{i}^{\mathrm{T}} \end{bmatrix} RBw + \tau_{M} w^{\mathrm{T}}B^{\mathrm{T}}R \begin{bmatrix} A_{i} & BK_{i} & Bk_{i} + a_{i} \end{bmatrix} \bar{x}_{s}$$

$$+ (\tau_{M} - \rho)\bar{x}_{s}^{\mathrm{T}} \begin{bmatrix} I \\ -I \\ 0 \end{bmatrix} XBw + (\tau_{M} - \rho)w^{\mathrm{T}}B^{\mathrm{T}}X \begin{bmatrix} I & -I & 0 \end{bmatrix} \bar{x}_{s}$$

$$+ \tau_{M}w^{\mathrm{T}}B^{\mathrm{T}}RBw - \gamma w^{\mathrm{T}}w + \bar{x}_{s}^{\mathrm{T}} \begin{bmatrix} E_{i}^{\mathrm{T}} & 0 \\ 0 & 0 \\ e_{i}^{\mathrm{T}} & 1 \end{bmatrix} \bar{\Lambda}_{i} \begin{bmatrix} E_{i} & 0 & e_{i} \\ 0 & 0 & 1 \end{bmatrix} \bar{x}_{s} < 0 \quad (6.41)$$

Inequality (6.15) is equivalent to (6.41) for  $\rho = 0$  and using Schur complement, inequality (6.16) is equivalent to (6.41) for  $\rho = \tau_M$ . Since (6.41) is affine in  $\rho$ , inequalities (6.15) and (6.16) imply that (6.41) is satisfied for any  $\rho \in [0, \tau_M]$ . In conclusion, (6.22) is satisfied for  $t_k < t < t_{k+1}$ ,  $k = 0, 1, 2, \ldots$  and any  $x \in \mathcal{R}_i, i = 1, 2, \ldots, M$ .

2. In the second part of the proof, it will be shown that for a given for  $0 < \theta < 1$ ,  $\Omega$  is an attracting invariant set. For any  $x_s \notin \Omega$ , one has

$$V(x_s,\rho) > \sigma_a \mu_\theta^2 + \sigma_b \tag{6.42}$$

It follows from (6.10) that  $||x_s|| > \mu_{\theta}$  and therefore (6.18) leads to

$$\sqrt{\theta\eta} \|x_s\| > \sqrt{\gamma} (\Delta_K \|x_s\| + \Delta_k) \tag{6.43}$$

It follows from (6.17) and (6.43) that

$$\theta \eta x_s^{\mathrm{T}} x_s > \gamma w^{\mathrm{T}} w \tag{6.44}$$

The inequality (6.22) can be written as

$$\dot{V}(x_s,\rho) \le -(1-\theta)\eta x_s^{\mathrm{T}} x_s - \theta \eta x_s^{\mathrm{T}} x_s + \gamma w^{\mathrm{T}} w$$
(6.45)

for  $0 < \theta < 1$ . Therefore it follows from (6.44) that

$$\dot{V}(x_s,\rho) \le -(1-\theta)\eta x_s^{\mathrm{T}} x_s \tag{6.46}$$

and from  $||x_s|| > \mu_{\theta}$ , one has

$$\dot{V}(x_s, \rho) \le -(1-\theta)\eta\mu_{\theta}^2, \text{ for } t_k < t < t_{k+1}$$
(6.47)

Therefore  $V(x_s, \rho)$  decreases between the sampling times for  $||x_s|| > \mu_{\theta}$ . As it was mentioned earlier,  $V(x_s, \rho)$  also decreases at each sampling time. Therefore there is a finite time  $t^{\theta}$  such that  $x_s(t^{\theta}) \in \Phi_{\theta}$  and therefore from (6.18), one has  $V(x_s(t^{\theta}), \rho) \leq \sigma_a \mu_{\theta}^2 + \sigma_b$ , which means  $x_s(t^{\theta}) \in \Omega$ . Therefore,  $\Omega$  is an attracting invariant set.

**Remark 6.1.** The upper bound for ||w|| defined in (6.17) can be obtained as

$$\Delta_{K} = \max_{i,j=1,...,M} \|K_{i} - K_{j}\|$$
  
$$\Delta_{k} = \max_{i,j=1,...,M} \|k_{i} - k_{j}\|$$
 (6.48)

Note that for the case where  $K_i = K_j$  and  $k_i = k_j$ ,  $\Delta_K = \Delta_k = 0$  and (6.20) is automatically satisfied. In this case w = 0 and  $\mu_{\theta} = 0$ .

**Remark 6.2.** For  $\tau_M \to 0$  and

$$\bar{P} = \begin{bmatrix} P & 0 \\ 0 & 0 \end{bmatrix}, \quad N_i = \begin{bmatrix} -PBK_i + I \\ -I \end{bmatrix}, \quad X = (\beta - 2)I \quad (6.49)$$

where  $\beta > \max(\eta, 2)$  and

$$\eta_c = \eta + \frac{\eta\beta}{\beta - \eta},\tag{6.50}$$

the inequalities (6.13), (6.14), (6.15) and (6.16) are reduced to the following inequalities for all  $i \in \mathcal{I}(0)$ 

$$\begin{bmatrix} (A_i + BK_i)^T P + P(A_i + BK_i) + \eta_c I & PB \\ B^T P & -\gamma I \end{bmatrix} < 0$$
(6.51)

and for  $i \notin \mathcal{I}(0)$ 

$$\begin{bmatrix} (\bar{A}_i + \bar{B}\bar{K}_i)^T \bar{P} + \bar{P}(\bar{A}_i + \bar{B}\bar{K}_i) + \bar{E}_i^T \bar{\Lambda}_i \bar{E}_i + \eta_c \begin{bmatrix} I & 0 \\ 0 & 0 \end{bmatrix} & \bar{P}\bar{B} \\ \bar{B}^T \bar{P} & -\gamma I \end{bmatrix} < 0 \quad (6.52)$$

Conditions (6.51) and (6.52) are sufficient conditions for input to state stability of the continuous-time PWA system (6.2) with the following condition satisfied for  $V(x) = x^T P x$ 

$$\dot{V}(x) < -\eta_c x^T x + \gamma w^T w \tag{6.53}$$

This result establishes that the continuous-time PWA controller should satisfy a very important property: non-fragility. In other words, if there exists an error w in the implementation of the continuous-time PWA controller (6.1) as shown in the following

$$u(t) = K_i x(t) + k_i + w(t)$$
(6.54)

and the norm of w is bounded, the norm of the state vector x(t) remains bounded.

#### 6.3 Numerical Example

**Example 6.1.** A state space model was built for an experimental two degrees of freedom helicopter in [38]. In this example, a simplified version of the pitch model of the experimental helicopter (Fig. 6.1) is considered. This model is described by



Figure 6.1: Pitch model for the experimental helicopter [38]

| value  | Unit                                                                      |
|--------|---------------------------------------------------------------------------|
| 0.0283 | $kgm^2$                                                                   |
| 0.9941 | kg                                                                        |
| 0.0134 | m                                                                         |
| 0.0289 | m                                                                         |
| 0.0003 | Nm                                                                        |
| 0.0041 | Nm/rad/s                                                                  |
| 9.81   | $m/s^2$                                                                   |
|        | Value<br>0.0283<br>0.9941<br>0.0134<br>0.0289<br>0.0003<br>0.0041<br>9.81 |

 Table 6.1: Parameters of the helicopter model
 Unit

 Parameter
 Value
 Unit

the following equations

$$\dot{x}_{1} = x_{2}$$
$$\dot{x}_{2} = \frac{1}{I_{yy}} (-m_{heli} l_{cgx} g \cos(x_{1}) - m_{heli} l_{cgz} g \sin(x_{1}) - F_{kM} \operatorname{sgn}(x_{2}) - F_{vM} x_{2} + u)$$
(6.55)

where  $x_1$  and  $x_2$  represent pitch angle and pitch rate, respectively. The values of the parameters are shown in Table 6.1.

The PWA approximation of the following nonlinear function in (6.55)

$$f(x_1) = -m_{heli} l_{cgx} g \cos(x_1) - m_{heli} l_{cgz} g \sin(x_1)$$

$$(6.56)$$

is then computed based on a uniform grid in  $x_1$ . The resulting approximation is shown in Figure 6.1. A PWA model is obtained by replacing  $f(x_1)$  by  $\hat{f}(x_1)$  in (6.55). The PWA model is described by the following equations

$$\begin{split} \dot{x} &= \begin{bmatrix} 0 & 1 \\ 5.3058 & -0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ 22.2968 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_1 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ -8.1786 & -0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -3.1208 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_2 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ -10.5751 & -0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -4.6265 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_3 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.9210 & -0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -12.4780 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_4 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 10.7980 & -0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -29.2108 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_5 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 5.3058 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ 22.2968 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_6 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ -8.1786 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -3.1208 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_7 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ -10.5751 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -4.6265 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_8 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ -10.5751 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -12.4780 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_8 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.9210 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -12.4780 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_9 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.9210 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -12.4780 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_9 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.9210 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -12.4780 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_9 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.0.7980 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -29.2108 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_9 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.0.7980 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -29.2108 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_9 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.0.7980 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -29.2108 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_9 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.0.7980 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -29.2108 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_9 \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.0.7980 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -29.2108 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_{10} \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.0.7980 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ -29.2108 \end{bmatrix} + \begin{bmatrix} 0 \\ 35.3012 \end{bmatrix} u & \text{for } x \in \mathcal{R}_{10} \\ \dot{x} &= \begin{bmatrix} 0 & 1 \\ 1.0.7980 & +0.1447 \end{bmatrix} x + \begin{bmatrix} 0 \\ 0.7980 & -29.2108 \end{bmatrix} + \begin{bmatrix} 0 \\ 0.7980 & -29.2$$

147

$$where \ x = \begin{bmatrix} x_1 \\ x_2 \end{bmatrix} and \\ \mathcal{R}_1 = \{x | -\pi < x_1 < -\frac{3\pi}{5}, x_2 > 0\} \qquad \mathcal{R}_2 = \{x | -\frac{3\pi}{5} < x_1 < -\frac{\pi}{5}, x_2 > 0\} \\ \mathcal{R}_3 = \{x | -\frac{\pi}{5} < x_1 < \frac{\pi}{5}, x_2 > 0\} \qquad \mathcal{R}_4 = \{x | \frac{\pi}{5} < x_1 < \frac{3\pi}{5}, x_2 > 0\} \\ \mathcal{R}_5 = \{x | \frac{3\pi}{5} < x_1 < \pi, x_2 > 0\} \qquad \mathcal{R}_6 = \{x | -\pi < x_1 < -\frac{3\pi}{5}, x_2 < 0\} \\ \mathcal{R}_7 = \{x | -\frac{3\pi}{5} < x_1 < -\frac{\pi}{5}, x_2 < 0\} \qquad \mathcal{R}_8 = \{x | -\frac{\pi}{5} < x_1 < \frac{\pi}{5}, x_2 < 0\} \\ \mathcal{R}_9 = \{x | \frac{\pi}{5} < x_1 < \frac{3\pi}{5}, x_2 < 0\} \qquad \mathcal{R}_{10} = \{x | \frac{3\pi}{5} < x_1 < \pi, x_2 < 0\}$$
(6.57)

The following PWA controller is then designed to stabilize the origin  $(x_1 = x_2 = 0)$ for the PWA system (6.57) using the backstepping method in subsection 5.3.1.

$$u = -0.2919x_1 - 0.1092x_2 - 0.6313, \text{ for } x \in \mathcal{R}_1$$
  

$$u = 0.0900x_1 - 0.1092x_2 + 0.0887, \text{ for } x \in \mathcal{R}_2$$
  

$$u = 0.1579x_1 - 0.1092x_2 + 0.1314, \text{ for } x \in \mathcal{R}_3$$
  

$$u = -0.1961x_1 - 0.1092x_2 + 0.3538, \text{ for } x \in \mathcal{R}_4$$
  

$$u = -0.4475x_1 - 0.1092x_2 + 0.8278, \text{ for } x \in \mathcal{R}_5$$
  

$$u = -0.2919x_1 - 0.1092x_2 - 0.6319, \text{ for } x \in \mathcal{R}_6$$
  

$$u = 0.0900x_1 - 0.1092x_2 + 0.0881, \text{ for } x \in \mathcal{R}_7$$
  

$$u = 0.1579x_1 - 0.1092x_2 + 0.1308, \text{ for } x \in \mathcal{R}_8$$
  

$$u = -0.1961x_1 - 0.1092x_2 + 0.3532, \text{ for } x \in \mathcal{R}_9$$
  

$$u = -0.4475x_1 - 0.1092x_2 + 0.8272, \text{ for } x \in \mathcal{R}_{10}$$

Using Theorem 6.1, a sampling time for discrete time implementation of the proposed PWA controller can be computed so that the closed loop sampled data system converges to a bounded invariant set. In this example, we consider  $\eta$  and  $\gamma$  as optimization parameters. However, to provide a larger upper bound on  $\Delta_K$ , we require that  $\eta > \gamma$  and  $\gamma > 1$ . Now, solving an optimization problem to maximize  $\tau_M$  subject



Figure 6.2: PWA function - Helicopter model

to the constraints of Theorem 6.1 and  $\eta > \gamma > 1$ , one has

$$\tau_M^{\star} = 0.1465, \ \eta = 4.2403, \ \gamma = 4.2403$$
(6.58)

$$P = \begin{bmatrix} 30.4829 & 2.4706\\ 2.4706 & 4.4771 \end{bmatrix}, R = \begin{bmatrix} 44.9622 & 9.0745\\ 9.0745 & 3.1994 \end{bmatrix},$$
(6.59)  
$$X = \begin{bmatrix} 499.9799 & 11.6429\\ 11.6429 & 24.1825 \end{bmatrix}$$
(6.60)

Figure 6.1 shows the trajectories of the nonlinear model (6.55) in feedback connection with the continuous time PWA controller. The trajectories of a sampled data PWA controller with a sampling time of 0.1465 second is shown in Figure 6.1.

#### 6.4 Conclusions

This chapter has presented stability results for closed-loop sampled-data PWA systems under state feedback. PWA sampled-data systems were considered as delay



Figure 6.3: Trajectories of the nonlinear Helicopter model - continuous time PWA controller



Figure 6.4: Trajectories of the nonlinear Helicopter model - sampled data PWA controller

systems with variable delay. The result for PWA systems is equivalent to the nonfragility of the continuous-time PWA controller when the sampling time converges to zero.

# Chapter 7

## Conclusions

The contributions of this thesis are summarized in this chapter. Potential extensions of the proposed methods are then discussed. Finally, open problems for future research are presented. The focus of this thesis has been to develop efficient *computational* controller synthesis methods for PWP/PWA systems. In the following, the fundamental questions raised in Chapter 1 are revisited considering the contributions of this work:

• How can PWA controllers be designed to keep the performance of linear controllers in a neighborhood of the equilibrium point, while guaranteeing a larger region of attraction?

Chapter 3 proposed a two-step synthesis method to achieve both local performance and global stability for nonlinear systems that can be *bounded* by PWA differential inclusions. In this method, a local robust linear controller is first designed for a neighborhood of the desired equilibrium point to satisfy a local performance requirement. The local linear controller is then extended to a PWA controller to globally stabilize the nonlinear system. The proposed method iscast as a set of BMIs and is not a convex problem. An open problem for future research is: Can the PWA extension of linear controllers be formulated as a convex problem?

• Is it possible to formulate the PWA/PWP controller synthesis as a convex optimization problem?

There are two contributions in this work toward answering this question: PWA controller synthesis for PWA slab system in chapter 4 and PWP controller synthesis for PWP systems in strict feedback form in chapter 5.

- In chapter 4, an interesting duality relation was revealed in the LMIs describing sufficient conditions for the stability of PWA slab differential inclusions. This concept was then employed to find a duality relation for the L<sub>2</sub>-gain design. As a result, the definition of the regions of a PWA slab system was extended, the L<sub>2</sub>-gain controller design was formulated as a set of LMIs and this design method was extended to PWA slab systems with an output that is also a PWA function of the state. The new method presented in chapter 4 enables stability and performance analysis, as well as controller synthesis, for PWA slab systems as a solution of convex optimization problems. The new concept of *dual parameter set* was the basis of the development of convex controller synthesis for PWA slab system does not necessarily define a PWA system. The open problem is: What is the dual of a PWA system?
- In chapter 5, the strict feedback form for PWP systems was introduced.
   Backstepping controller synthesis for this large class of PWP systems was formulated as an SOS program, which is a convex optimization problem.
   The synthesis problem was addressed in two cases: SOS Lyapunov functions for PWP systems with discontinuous vector fields and PWP Lyapunov functions for PWP systems with continuous vector fields. One of

the main advantages of the proposed method is that it addresses PWP systems with discontinuous vector fields regardless of possible attractive sliding modes. A very useful extension of the proposed method is backstepping controller synthesis for PWP differential inclusions.

• For a sampled-data implementation of a continuous-time PWA controller, how large can the sampling time be?

Chapter 6 presented stability analysis results for closed-loop sampled-data PWA systems under state feedback. These results were obtained by considering PWA sampled-data systems as delay systems with a variable delay. In chapter 6, it is assumed that a PWA controller is given and then the maximum sampling time is computed as the solution of a convex problem. An interesting extension would be a convex PWA controller synthesis method to guarantee stability of the closed-loop sampled-data system for a given sampling time.

Based on the previous observation, the proposed extensions of the current research are as follows:

- 1. To develop a backstepping PWP controller synthesis method for PWP differential inclusions in strict feedback form
- 2. To develop a convex PWA controller synthesis method to guarantee stability of the closed-loop sampled-data system for a given sampling time

154

Finally, two fundamental open problems to be solved are

- 1. Can general PWP/PWA controller synthesis be converted to a convex problem?
- 2. What is the dual of a PWA system?

Analysis and synthesis of PWS systems thus seems to be a very rich field of study.

# Bibliography

- J. H. Ahlberg, E. N. Nilson, and J. L. Walsh. The Theory of Splines and Their Applications. Academic Press, 1967.
- M. Akar and U. Ozguner. Decentralized sliding mode design for hybrid systems. In Proc. American Control Conference, pages 525 – 529, San Diego, CA, June 1999.
- [3] Joaquin Alvarez, Iouri Orlov, and Leonardo Acho. An invariance principle for discontinuous dynamic systems with application to a coulomb friction oscillator. Journal of Dynamic Systems, Measurement, and Control, 122:687–690, Dec. 2000.
- [4] A. Andronov and C. Chaikin. *Theory of Oscillations*. Princeton University Press, Princeton, New Jersey, 1949.
- [5] Panos J. Antsaklis and Xenofon D. Koutsoukos. Software-Enabled Control: Information Technologies for Dynamical Systems, chapter Hybrid Systems: Review and Recent Progress, pages 272–298. IEEE Press., 2003.
- [6] S. Azuma and J. Imura. Probabilistic controllability analysis of sampleddata/discrete-time piecewise-affine systems. In Proc. American Control Conference, pages 2528–2533, 2004.

- [7] A. Bacciotti and F. M. Ceragioli. L<sub>2</sub>-gain stabilizability with respect to filippov solutions. In Proc. of the 41st IEEE Conference on Decision and Control, Las Vegas, Nevada, December 2002.
- [8] M. Baotic, F.J. Christophersen, and M. Morari. Infinite time optimal control of hybrid systems with a linear performance index. In Proc. of the 42nd IEEE Conference on Decision and Control, Maui, HI, December 2003.
- [9] M. Baotic, F.J. Christophersen, and M. Morari. A new algorithm for constrained finite time optimal control of hybrid systems with a linear performance index. In *Proc. European Control Conference*, Cambridge, UK, September 2003.
- [10] J. R. Basar and G. J. Olsder. Dynamic Non-Cooperative Game Theory. Academic, New York, 1995.
- [11] A. Bemporad. Efficient conversion of mixed logical dynamical systems into an equivalent piecewise affine form. *IEEE Trans. Autom. Control*, 49:832–838, May 2004.
- [12] A. Bemporad, M. Morari, V. Dua, and E. N. Pistikopoulos. The explicit linear quadratic regulator for constrained systems. *Automatica*, 38:3–20, January 2002.
- [13] A. Bensoussan and J. L. Menaldi. Hybrid control and dynamic programming. Dynamics of Continuous, Discrete and Impulsive Systems, 3(4):395–442, 1997.
- [14] C. Bissell. A. A. Andronov and the development of soviet control engineering. *IEEE Control Syst. Mag.*, 18:56–62, February 1998.
- [15] V. D. Blondel and J. N. Tsitsiklis. Complexity of stability and controllability of elementary hybrid systems. *Automatica*, 35:479–489, 1999.

- [16] F. Borrelli, M. Baotic, A. Bemporad, and M. Morari. An efficient algorithm for computing the state feedback optimal control law for discrete time hybrid systems. In *Proc. American Control Conference*, pages 4717–4722, Denver, CO, June 2003.
- [17] S. Boyd, L. El Ghaoui, E. Feron, and V. Balakrishnan. Linear Matrix Inequalities in System and Control Theory (Studies in Applied Mathematics). SIAM, Philadelphia, 1994.
- [18] M. S. Branicky. General hybrid dynamical systems: Modeling, analysis, and control. In R. Alur, T. Henzinger, and E. D. Sontag, editors, *Hybrid Systems III Verification and Control: Lecture Notes in Computer Science*, volume 1066, pages 186–200. Springer-Verlag, Berlin, 1996.
- [19] M. S. Branicky. Multiple Lyapunov functions and other analysis tools for switched and hybrid systems. *IEEE Trans. Autom. Control*, 43:461–474, April 1998.
- [20] M. S. Branicky, V. S. Borkar, and S. K. Mitter. A unified framework for hybrid control: Model and optimal control theory. *IEEE Trans. Autom. Control*, 43:31–45, January 1998.
- [21] P. E. Caines and Y. J. Wei. Hierarchical hybrid control systems: a lattice theoretic formulation. *IEEE Trans. Autom. Control*, 43:501–508, April 1998.
- [22] Victoriano Carmona, Emilio Freire, Enrique Ponce, and Francisco Torres. On simplifying and classifying piecewise-linear systems. *IEEE Trans. Circuits* Syst. I, 49:609–620, May 2002.
- [23] C. G. Cassandras, D. L. Pepyne, and Y. Wardi. Optimal control of a class of hybrid systems. *IEEE Trans. Autom. Control*, 46:398–415, March 2001.

- [24] F. M. Ceragioli. Discontinuous ordinary differential equations and stabilization. PhD thesis, Università di Firenze, 1999.
- [25] D. Chatterjee and D. Liberzon. Stability analysis of deterministic and stochastic switched systems via a comparison principle and multiple Lyapunov functions. SIAM J. CONTROL OPTIM., 45:174–206, 2006.
- [26] M. Chen, C. R. Zhu, and G. Feng. Linear-matrix-inequality based approach to H<sub>∞</sub> controller synthesis of uncertain continuous-time piecewise linear systems. *IEE Proceedings - Control Theory and Applications*, 151:295–301, May 2004.
- [27] T. Chen and B. Francis. Optimal Sampled Data Control Systems. Springer-Verlag, 1995.
- [28] Y. C. Cho, C. G. Cassandras, and D. L. Pepyne. Forward decomposition algorithms for optimal control of a class of hybrid systems. *International Journal of Robust and Nonlinear Control*, 11:497–513, 2001.
- [29] L. O. Chua. Analysis and synthesis of multivalued memoryless nonlinear networks. *IEEE Trans. Circuit Theory*, 14:192–209, June 1967.
- [30] L. O. Chua. Efficient computer algorithms for piecewise linear analysis of resistive nonlinear networks. *IEEE Trans. Circuit Theory*, 18:73–85, January 1971.
- [31] F. H. Clarke, Y. S. Ledyaev, R. J. Stern, and P. R. Wolenski. Nonsmooth Analysis and Control Theory. Springer Verlag, New York, 1998.
- [32] Dan Dai, Tingshu Hu, Andrew R. Teel, and Luca Zaccarian. Analysis of systems with saturation/deadzone via piecewise-quadratic lyapunov functions. In Proceedings of the 2007 American Control Conference, pages 5822–5827, New York City, USA, 2007.

- [33] G. Davrazos and N. T. Koussoulas. A review of stability results for switched and hybrid systems. In *Mediterranean Conference on Control and Automation*, 2001.
- [34] C. de Boor. A Practical Guide to Splines. Springer-Verlag, 1978.
- [35] R. A. Decarlo, M. S. Branicky, S. Pettersson, and B. Lennartson. Perspectives and results on the stability and stabilizability of hybrid systems. *Proc. IEEE*, 88:1069–1082, July 2000.
- [36] John C. Doyle, Bruce A. Francis, and Allen R. Tannenbaum. Feedback Control Theory. Macmillan, 1990.
- [37] C. Ebenbauer. Polynomial Control Systems: Analysis and Design via Dissipation Inequalities and Sum of Squares. PhD thesis, University of Stuttgart, 2005.
- [38] R. Endreß and L. Rodrigues. Modeling and control of two degrees of freedom quanser helicopter. Technical report, Concordia University, Jan. 2008.
- [39] M. Fazel, H. Hindi, and S. Boyd. Rank minimization and application in system theory. In Proc. American Control Conference, pages 3273–3278, Boston, MA, June 2004.
- [40] G. Feng. Controller design and analysis of uncertain piecewise-linear systems. *IEEE Trans. Circuits Syst. I*, 49:224–232, February 2002.
- [41] G. Feng. Linear-matrix-inequality based approach to  $H_{\infty}$  controller synthesis of uncertain continuous-time piecewise linear systems. *IEE Proc. Circuit Theory Appl.*, 151(3):295–301, May 2004.

- [42] G. Feng, G.P. Lu, and S.S. Zhou. An approach to H∞ controller synthesis of piecewise linear systems. Communications in Information and Systems, 2:245–254, December 2002.
- [43] G. Ferrari-Trecate, F.A. Cuzzola, D. Mignone, and M. Morari. Analysis and control with performance of piecewise affine and hybrid systems. In Proc. American Control Conference, pages 200–205, Arlington, VA, June 2001.
- [44] A. F. Filippov. Differential equations with discontinuous right-hand sides. *Mtematischeskii*, 51:199–231, 1960, in Russian, Translated in English, (1964) *American Mathematical Society Transactions*, Series 2, Vol. 42.
- [45] I.A. Fotiou, A.G. Beccuti, G. Papafotiou, and M. Morari. Optimal control of piece-wise polynomial hybrid systems using cylindrical algebraic decomposition. In *HSCC (Hybrid Systems: Computation and Control )*, pages 227–241, Santa Barbara, CA, March 2006.
- [46] G. K. Fourlas, K. J. Kyriakopoulos, and C. D. Vournas. Hybrid systems modeling for power systems. *IEEE Trans. Circuits Syst.*, 4(3):16–23, 2004.
- [47] Z. Galias. Rigorous investigations of piecewise linear circuits. In Proc. Int. Conference on Signals and Electronic Systems, ICSES'02, pages 141–146, Świeradów Zdrój, 2002.
- [48] K. Gokbayrak and C. G. Cassandras. A hierarchical decomposition method for optimal control of hybrid systems. In Proc. of the 39th IEEE Conference on Decision and Control, pages 1816–1821, Sydney, Australia, December 2000.
- [49] J. M. Gonçalves, A. Megretski, and M.A.Dahleh. Global analysis of piecewise linear systems using impact maps and surface Lyapunov functions. *IEEE Trans. Autom. Control*, 48:2089–2106, December 2003.

- [50] G. Grammel. Maximum principle for a hybrid system via singular pertrubations. SIAM J. Contr. Optim., 37(4):1162–1175, 1998.
- [51] P. Grieder, F. Borrelli, F. D. Torrisi, and M. Morari. Computation of the constrained infinite time linear quadratic regulator. In Proc. American Control Conference, Denver, CO, June 2003.
- [52] P. Grieder, M. Kvasnica, M. Baotic, and M. Morari. Low complexity control of piecewise affine systems with stability guarantee. In *Proc. American Control Conference*, Boston, MA, June 2004.
- [53] P. Grieder and M. Morari. Complexity reduction of receding horizon control. In Proc. of the 42nd IEEE Conference on Decision and Control, Maui, HI, December 2003.
- [54] P. Grieder, P. Parillo, and M. Morari. Robust receding horizon control analysis and synthesis. In Proc. of the 42nd IEEE Conference on Decision and Control, Maui, HI, December 2003.
- [55] L.C.G.J.M. Habets and J.H. van Schuppen. Control of piecewise-linear hybrid systems on simplices and rectangles. In *Proceedings of Workshop Hybrid Sys*tems, Computation and Control. Springer-Verlag, Rome, Italy, March 28-30 2001.
- [56] A. Hassibi and S. Boyd. Quadratic stabilization and control of piecewise-linear systems. In Proc. American Control Conference, pages 3659–3664, Philadelphia, PA, June 1998.
- [57] S. Hedlund and M. Johansson. PWLTOOL: A MATLAB toolbox for analysis of piecewise linear systems. Technical report, Department of Automatic Control, Lund Institute of Technology, March 1999.

- [58] W. P. M. H. Heemels, B. De Schutter, and A. Bemporad. Equivalence of hybrid dynamical models. *Automatica*, 37:1085–1091, July 2001.
- [59] W.P.M.H. Heemels, S. Weiland, and A.L. Juloski. Hybrid Systems: Computation and Control; 10th International Workshop, HSCC 2007, chapter Inputto-state stability of discontinuous dynamical systems with an observer-based control application, pages 259–272. Springer, 2007.
- [60] J. P. Hespanha. Stabilization through hybrid control. UNESCO Encyclopedia of Life Support Systems, 2004.
- [61] J. P. Hespanha. Uniform stability of switched linear systems: extensions of lasalle's invariance principle. *IEEE Trans. Autom. Control*, 49:470–482, April 2004.
- [62] J. Imura. Optimal continuous-time control of sampled-data piecewise affine systems. In Proc. American Control Conference, pages 5317–5322, 2003.
- [63] J. Imura. Optimal control of sampled-data piecewise affine systems and its application to CPU processing control. In Proc. of the 42nd IEEE Conf. on Decision and Control, pages 161–166, 2003.
- [64] A. Jadbabaie. Robust, Non-Fragile Controller Synthesis Using Model Based Fuzzy System: A Linear Matrix Inequality Approach. M.Sc. thesis, The University of New Mexico, Albuquerque, New Mexico, November 1997.
- [65] K.H. Johansson. Theme 6.43: Control systems, robotics and automation.
   In H. Unbehauen, editor, *Encyclopedia of Life Support Systems (EOLSS)*.
   Developed under the auspices of UNESCO, 2004.
- [66] M. Johansson. Piecewise Linear Control Systems. Springer, Berlin, 2003.

- [67] M. Johansson and A. Rantzer. Computation of piecewise quadratic Lyapunov functions for hybrid systems. *IEEE Trans. Autom. Control*, 43:555–559, April 1998.
- [68] T. Kailath. *Linear Systems*. Prentice Hall, 1989.
- [69] E. C. Kerrigan and D. Q. Mayne. Optimal control of constrained, piecewise affine systems with bounded disturbances. In Proc. of the 41st IEEE Conference on Decision and Control, Las Vegas, NV, December 2002.
- [70] H. Khalil. Nonlinear Systems. Prentice-Hal. Inc., 3rd ed., 2002.
- [71] M. Kocvara, F. Leibfritz, M. Stingl, and D. Henrion. A nonlinear SDP algorithm for static output feedback problems in COMPl<sub>e</sub>ib. Technical report, University of Trier, Germany, 2004.
- [72] M. Krstic, I. Kanellakopoulos, and P. Kokotovic. Nonlinear and Adaptive Control Design. John Wiley and Sons, New York, 1995.
- [73] Adam Lagerberg and Bo Egardt. Model predictive control of automotive powertrains with backlash. In 16th IFAC World Congress, July 2005.
- [74] D. Liberzon. Switching in systems and Control. Birkhauser, Boston, 2003.
- [75] H. Lin and P. J. Antsaklis. Stability and stabilizability of switched linear systems: a short survey of recent results. In Proc. International Symposium on Intelligent Control, pages 24–29, Limassol, Cyprus, June 2005.
- [76] J. Löfberg. YALMIP : A toolbox for modeling and optimization in MATLAB.In Proceedings of the CACSD Conference, Taipei, Taiwan, 2004.
- [77] Albert C. J. Luo. A periodically forced, piecewise linear system, part ii: The fragmentation mechanism of strange attractors and grazing. *Communications* in Nonlinear Science and Numerical Simulation, 12:986–1004, 2007.
- [78] D. Q. Mayne, J. B. Rawlings, C. V. Rao, and P. O. M. Scokaert. Constrained model predictive control: Stability and optimality. *Automatica*, 36:789–814, June 2000.
- [79] A. N. Michel and B. Hu. Towards a stability theory of general hybrid dynamical systems. *Automatica*, 35:371–384, 1999.
- [80] Richard M. Murray. Top ten research problems in nonlinear control. http://www.cds.caltech.edu/ murray/archive/topten/, 1996.
- [81] K. G. Murty and S. N. Kabadi. Some NP-complete problems in quadratic and nonlinear programming. *Mathematical Programming*, 39:117–129, 1987.
- [82] P. Naghshtabrizi, J. P. Hespanha, and A. R. Teel. On the robust stability and stabilization of sampled-data systems: A hybrid system approach. In 45th IEEE Conference on Decision and Control, pages 4873–4878, San Diego, CA, USA, December 2006.
- [83] Yu. Nesterov and A. Nemirovsky. Interior-point polynomial methods in convex programming, volume 13 of Studies in Applied Mathematics. SIAM, June 1994.
- [84] R. Palm and T. Runkler. Decentralized control of hybrid systems autonomous decentralized systems. In *Proceedings. 5th International Symposium on*, pages 287 – 294, Dallas, TX, March 2001.
- [85] A. Papachristodoulou and S. Prajna. A tutorial on sum of squares techniques for systems analysis. In Proc. American Control Conference, pages 2686–2700, Pasadena, CA, June 2005.
- [86] G. J. Pappas, G. Lafferriere, and S. Sastry. Hierarchically consistent control systems. In Proc. of the 40th IEEE Conference on Decision and Control, pages 4336–4341, Tampa, FL, December 1998.

- [87] Peter Paul. Piecewise Polynomial System Approximation for Nonlinear Control. PhD thesis, Case Western Reserve University, 1994.
- [88] A. Pavlov, N. Wouw, and H. Nijmeijer. Convergent piecewise affine systems: analysis and design part I: continuous case. In Proc. of the 44th IEEE Conference on Decision and Control and European Control Conference, Seville, Spain, December 2005.
- [89] A. Pavlov, N. Wouw, and H. Nijmeijer. Convergent piecewise affine systems: analysis and design part II: discontinuous case. In Proc. of the 44th IEEE Conference on Decision and Control and European Control Conference, Seville, Spain, December 2005.
- [90] P. Peleties and R. A. DeCarlo. Asymptotic stability of m-switched systems using Lyapunov-like functions. In *Proceedings of the American Control Conference*, pages 1679–1684, 1991.
- [91] S. Pettersson. Analysis and design of hybrid systems. PhD thesis, Chalmers University of Technology, Gteborg, June 1999.
- [92] N. B. O. L. Pettit. Analysis of Piecewise Linear Dynamical Systems. Research Studies Press Ltd./ John Wiley & Sons Limited, 1995.
- [93] S. Prajna and A. Papachristodoulou. Analysis of switched and hybrid systems
  beyond piecewise quadratic methods. In *Proc. American Control Conference*, pages 2779–2784, Denver, CO, June 2003.
- [94] S. Prajna, A. Papachristodoulou, P. Seiler, and P. A. Parrilo. SOSTOOLS and Its Control Applications, pages 273–292. Springer-Verlag, 2005.
- [95] S. Prajna, A. Papachritodoulou, P. Seiler, and P. A. Parrilo. SOSTOOLS sum of squares optimization toolbox, user's guide, 2004.

- [96] A. Rantzer and M. Johansson. Piecewise linear quadratic optimal control. *IEEE Trans. Autom. Control*, 45:629–637, April 2000.
- [97] L. Rodrigues. Dynamic output feedback controller synthesis for piecewise-affine systems. PhD thesis, Stanford University, Jun. 2002.
- [98] L. Rodrigues. Stability analysis of piecewise-affine systems using controlled invariant sets. Systems and Control Letters, 53:157–169, 2004.
- [99] L. Rodrigues. Stability of sampled-data piecewise-affine systems under state feedback. *Automatica*, 43:1249–1256, 2007.
- [100] L. Rodrigues and E. K. Boukas. Piecewise-linear  $H_{\infty}$  controller synthesis with applications to inventory control of switched production systems. *Automatica*, 42:1245–1254, 2006.
- [101] L. Rodrigues and S. Boyd. Piecewise-affine state feedback for piecewise-affine slab systems using convex optimization. Systems and Control Letters, 54:835– 853, 2005.
- [102] L. Rodrigues, A. Hassibi, and J.P. How. Output feedback controller synthesis for piecewise-affine systems with multiple equilibria. In Proc. American Control Conference, pages 1784–1789, Chicago, IL, June 2000.
- [103] L. Rodrigues and J. P. How. Observer-based control of piecewise-affine systems. International Journal of Control, 76:459–477, March 2003.
- [104] L. Rodrigues and J.P. How. Synthesis of piecewise affine controllers for stabilization of nonlinear systems. In Proc. of the 42nd IEEE Conference on Decision and Control, pages 2071–2076, Maui, Hawaii, December 2003.

- [105] L. Rodrigues and J.P. How. Towards unified controller synthesis for hybrid systems: State feedback of sequential systems. In Proc. International Conference on Advances in Dynamics, Instrumentation and Control, pages 298–311, 2004.
- [106] N. Rouche, P. Habets, and M. Laloy. Stability Theory by Liapunov's Direct Method. Springer-Verlag, 1977.
- [107] H. L. Royden. *Real Analysis.* Prentice Hall, 1988.
- [108] W. Rugh and J. S. Shamma. Research on gain scheduling. Automatica, 36(9):1401–1425, November 2000.
- [109] B. Samadi and L. Rodrigues. Controller synthesis for piecewise affine slab differential inclusions: A duality-based convex optimization approach. In Proc. of the 46th IEEE Conference on Decision and Control, pages 4999 – 5004, New Orleans, LA, USA, December 2007.
- [110] C. Scherer and S. Weiland. LMIs in Controller Analysis and Synthesis.
- [111] T. Schlegl, M. Buss, and G. Schmidt. A hybrid systems approach toward modeling and dynamical simulation of dextrous manipulation. *IEEE/ASME Trans. Mechatronics*, 8(3):352–361, September 2003.
- [112] R. Sengupta and S. Lafortune. An optimal control theory for discrete event systems. SIAM J. Contr. Optim., 36(2):488–541, 1998.
- [113] M. S. Shaikh and P. E. Caines. On the optimal control of hybrid systems: Analysis and zonal algorithms for trajectory and schedule optimization. In Proc. of the 42nd IEEE Conference on Decision and Control, pages 2144– 2149, Maui, HI, December 2003.

- [114] M. S. Shaikh and P. E. Caines. On the optimal control of hybrid systems: optimization of switching times and combinatoric location schedules. In Proc. American Control Conference, pages 2773–2778, Denver, CO, June 2003.
- [115] J.J. E. Slotine and W. Li. Applied Nonlinear Control. Prentice-Hall, Englewood Cliffs, NJ, 1991.
- [116] O. Sluphaug. On Robust Constrained Nonlinear and Hybrid Control. PhD thesis, Norwegian University of Science and Technology, October 1998.
- [117] E. D. Sontag. Nonlinear regulation: The piecewise linear approach. IEEE Trans. Autom. Control, 26(2):346–358, April 1981.
- [118] E. D. Sontag. A Lyapunov-like characterization of asymptotic controllability. SIAM J. Control Optim., 21:462471, 1983.
- [119] J. Soo and M. Tomizuka. Limit cycles due to friction forces in flexible joint mechanisms. In Proceedings 2005 IEEE/ASME International Conference on Advanced Intelligent Mechatronics, pages 723 – 728, Monterey, California, USA, Jul. 2005.
- [120] T. E. Stern. Piesewise linear network analysis and synthesis. In Proc. of Symp. on Circuit Analysis, pages 315 – 346, Polytechnic Institute of Brooklyn, NY, 1956.
- [121] J. F. Strum. Using SeDuMi 1.05, a MATLAB toolbox for optimization over symmetric cones. Technical report, Tilburg University, 2001.
- [122] Z. Sun. Canonical froms of switched linear control systems. In Proc. American Control Conference, pages 5182–5187, 2004.
- [123] Z. Sun and S. S. Ge. Sampling and control of switched linear systems. In Proc. of the 41st IEEE Conference on Decision and Control, pages 4413–4418, 2002.

- [124] Onur Toker and Hitay Ozbay. On the NP-hardness of solving bilinear matrix inequalities and simultaneous stabilization with static output feedback. In *Proc. American Control Conference*, pages 2525–2526, Seattle, Washington, June 1995.
- [125] C. J. Tomlin, J. Lygeros, and S. S. Sastry. A game theoretic approach to controller design for hybrid systems. *Proc. IEEE*, 88:949–970, July 2000.
- [126] F. D. Tossisi and A. Bemporad. Hysdel a tool for generating computational hybrid models for analysis and design problems. *IEEE Trans. Control Syst. Technol.*, 12:235–249, March 2004.
- [127] W. M. Wonham. Supervisory control of discrete-event systems. Technical report, Systems Control Group, Dept. of ECE, University of Toronto, 1997-2007. ECE 1636F/1637S.
- [128] H. Ye, A. N. Michel, and L. Hou. Stability theory for hybrid dynamical systems. *IEEE Trans. Autom. Control*, 43:461–474, April 1998.
- [129] A. M. G. Zhai, H. Lin, and K. Yasuda. Stability analysis for switched systems with continuous-time and discrete-time subsystems. In Proc. American Control Conference, pages 4555–4560, 2004.

## Index

Piecewise affine system, 3Piecewise linear system, 3Piecewise polynomial system, 3Piecewise smooth system, 1, 19PWA slab system, 44